Andreea G. Andrei

FUTURE OF THE EUROPEAN DATA CENTER INDUSTRY

Analysis of the industry and the market opportunities for future expansion

Master's Thesis CENTRIA UNIVERSITY OF APPLIED SCIENCES M.B.A. International Business Management May 2024

ABSTRACT

Centria University	Date	Author				
of Applied Sciences	May 2024	Andreea Georgiana Andrei				
Degree programme						
International Business Management						
Name of thesis						
FUTURE OF THE EUROPEAN DATA CENTER INDUSTRY						
Analysis of the industry and the market opportunities for future expansion						
Centria supervisor		Pages				
Dr. Johanna Hallbäck 90 + 6						
Instructor representing commissioning institution or company						
No commissioner						

In this thesis, a market opportunity analysis and market selection for future European data center industry expansion were conducted. The purpose was to understand the data center industry in Europe and to explore the opportunities to expand it. Based on the research findings, the main objectives were to determine which factors had the biggest impact in local market expansion by analysing the current state of the industry in Europe, as well as to research the factors that most influence the quality and functionality of a data center, conduct a systematic market selection process, determine which countries would be the most viable location candidates to develop the industry within and provide market entry recommendations for the selected countries.

The theoretical framework consisted of research on the topics of market opportunity analysis, market research, market selection and market entry strategies followed by an examination of the European data center industry based on qualitative and quantitative secondary data from sources commonly considered reliable, upon which an assessment and comparison were carried out to determine the factors contributing to the 2024 international environment.

Consequently, the data center industry was investigated to identify the factors prioritized in the decision-making process of industry development. Based on these findings, the PESTEL tool was utilized to conduct a preliminary screening of all European countries, with relevant statistical data collected from ethical sources. Ranked tables were then created for all analysed factors, with general conclusions drawn from political, economic, social, technological, ecological and legal perspectives.

In a final step, the resulting information was compiled into two tables summarizing each European country's performance across the six categories of factors, including those with the most favourable statistical data for local data center industry expansion and those with the least favourable data. A market opportunity analysis was conducted and the final best candidates for local expansion were determined by excluding countries where the industry was already well developed, aiming to achieve equitable, uniform international expansion. The thesis concluded with considerations on potential market entry strategies for the selected countries: Croatia, Czechia, Denmark, Finland, Ireland, Latvia, Lithuania, Portugal, and Slovenia.

Key words

Data center, Europe, market entry strategy, market research, market selection process, market opportunity analysis

CONCEPT DEFINITIONS

DC

Data Center

FDI

Foreign Direct Investment

GDP

Gross Domestic Product

GNP

Gross National Product

HVAC

Heating, Ventilation, Air Conditioning

MW

Megawatt

PBDC

Purpose-Built Data Centers

PESTEL

Political, Economic, Sociocultural, Technological, Environmental, Legal

PPP

Purchasing Power Parity

ABSTRACT CONCEPT DEFINITIONS CONTENTS

1 INTRODUCTION	1
2 RESEARCH ON MARKET OPPORTUNITY ANALYSIS	3
2.1 Market segmentation	3
2.2 PESTEL analysis	5
2.3 Use of secondary data	8
3 RESEARCH ON MARKET ENTRY STRATEGY	11
3.1 Market entry strategy	11
3.2 Market entry considerations	12
3.3 Marketing mix	14
4 RESEARCH METHODOLOGY	16
5 DATA CENTER INDUSTRY OVERVIEW	19
5.1 History and development	20
5.2 Expansion requirements assessment	22
5.2.1 Geographical surface	22
5.2.2 Population	23
5.2.3 Access to electricity	24
5.2.4 Internet usage	24
5.2.5 Local factors conclusions	25
5.2.6 Average temperature	25
5.2.7 Electric power stability	26
5.2.8 Risk of catastrophic events	27
5.2.9 GDP	28
5.2.10 Electricity rate	29
5.2.11 Profit tax rate	30
5.2.12 PPP	32
5.2.13 Conclusions	33
5.3 Selecting the right location for a data center	33
5.3.1 Energy efficiency	34
5.3.2 Data center construction cost	35
5.3.3 Risk of natural disasters and weather events	57
5.3.4 Access to infrastructure and supplies	
5.3.5 Conclusions	38
6 PRELIMINARY SCREENING AND PESTEL ANALYSIS FOR DATA CENTERS	IN
0.1 FOILUCAL PREHIMINARY SCREENING	39
0.1.1 Political stability index	59
0.1.2 Kestrictions on ioreign investments	41
0.1.5 Difficulty index level for foreign controlled enterprises to invest locally	41
0.1.4 COFFUPTION INDEX	42
0.1.5 Security inreat maex	43

(1) Delitical multiminant among a conclusion	11
6.1.6 Political preliminary screening conclusion	
6.2 Legal preliminary screening	
6.2.1 Trade freedom index	
6.2.2 Taxes on the income or profits of corporations	
6.2.3 Investment ireedom index	
6.2.4 Fiscal freedom index	
6.2.5 Legal preliminary screening conclusion	
6.3 Ecological preliminary screening	50
6.3.1 Share of renewable energy	50
6.3.2 Emissions index	51
6.3.3 Ecological preliminary screening conclusion	53
6.4 Technological preliminary screening	53
6.4.1 Internet speed	53
6.4.2 Internet bandwidth	54
6.4.3 Electricity production capacity	55
6.4.4 Renewable power capacity	56
6.4.5 Road quality index	57
6.4.6 Railroad quality index	59
6.4.7 Port infrastructure quality index	60
6.4.8 Air infrastructure quality index	61
6.4.9 Technological preliminary screening conclusion	63
6.5 Socio-cultural preliminary screening	63
6.5.1 Rule of law	63
6.5.2 Globalization index	64
6.5.3 Positive attitude towards immigrants index	65
6.5.4 Socio-cultural preliminary screening conclusion	67
6.6 Economic preliminary screening	67
6.6.1 Labour cost index	67
6.6.2 Transport prices	68
6.6.3 Business freedom index	69
6.6.4 Communication prices	70
6.6.5 Machinery and equipment prices	71
6.6.6 Economic preliminary screening conclusion	72
6.7 Preliminary screening conclusion	73
7 MARKET ENTRY CONSIDERATIONS FOR DATA CENTER EXPANSION	77
7.1 Croatia	77
7.2 Czechia	78
7.3 Denmark	79
7.4 Finland	79
7.5 Ireland	81
7.6 Latvia	
7.7 Lithuania	
7.8 Portugal	
7.9 Slovenia	
8 CONCLUSIONS	
8.1 Discussion and summarizing conclusion	
8.2 Ethics and reliability of the thesis	
	0

REFERENCES	91
APPENDICES	

FIGURES

FIGURE 1. Number of DCs in Europe by country 2021	22
FIGURE 2. Number of DCs in Europe by country 2021 compared to country surface	24
FIGURE 3. Number of DCs in Europe by country 2021 compared to average yearly temperature	27
FIGURE 4. Number of DCs in Europe by country 2021 compared to electric power transmission a	nd
distribution losses	28
FIGURE 5. Number of DCs in Europe by country 2021 compared to risk of droughts, floods, extre	me
temperatures	29
FIGURE 6. Number of DCs in Europe by country 2021 compared to GDP per capita in 2021	30
FIGURE 7. Number of DCs in Europe by country 2021 compared to electricity prices for non-hous	se-
hold customers including taxes	31
FIGURE 8. Number of DCs in Europe by country 2021 compared to commercial profit tax	32
FIGURE 9. Number of DCs in Europe by country 2021 compared to purchasing power parity	33
FIGURE 10. Preliminary screening conclusion – number and type of identified risks for European	
countries	75
FIGURE 11. Preliminary screening conclusion - ranking of leading European countries	76
FIGURE 12. The most viable markets after preliminary screening	77

PICTURES

TABLES

TABLE 1. Ranking of European countries by political stability index in 2021	41
TABLE 2. Ranking of European countries by corruption index in 2022	44
TABLE 3. Ranking of European countries by security threats index in 2022	45
TABLE 4. Ranking of European countries by trade freedom in 2022	47
TABLE 5. Ranking of European countries by taxes on the income or profits of corporations in 20	19 as
percentage of GDP	48
TABLE 6. Ranking of European countries by investment freedom in 2022	49
TABLE 7. Ranking of European countries by fiscal freedom index in 2022	50
TABLE 8. Ranking of European countries by share of renewable energy in 2020	52
TABLE 9. Ranking of European countries by emissions index in 2020	53
TABLE 10. Ranking of European countries by internet speed in 2023	55
TABLE 11. Ranking of European countries by internet bandwidth in 2016	56
TABLE 12. Ranking of European countries by electricity production capacity in 2020	57
TABLE 13. Ranking of European countries by renewable power capacity in 2020	58
TABLE 14. Ranking of European countries by roads quality in 2019	59
TABLE 15. Ranking of European countries by railroads quality in 2019	60
TABLE 16. Ranking of European countries by ports infrastructure quality in 2019	62
TABLE 17. Ranking of European countries by air infrastructure quality in 2019	63
TABLE 18. Ranking of European countries by rule of law in 2021	65
TABLE 19. Ranking of European countries by social globalization in 2020	66
TABLE 20. Ranking of European countries by positive general attitude towards immigrants in 20	022 67
TABLE 21. Ranking of European countries by labour cost in 2022	69
TABLE 22. Ranking of European countries by transport prices in 2017	70
TABLE 23. Ranking of European countries by business freedom index in 2022	71
TABLE 24. Ranking of European countries by communication prices in 2017	72

TABLE 25. Ranking of European countries by machinery and equipment prices in 2017......73

1 INTRODUCTION

This thesis focuses on analysing the European data center (DC) industry and its expansion. Currently, most European DCs are being built in a few Western countries, mainly Germany (487 DCs), UK (456 DCs) and Netherlands (281 DCs), while other European countries have as few as one DC (Figure 1). In order to continue expanding the DC industry efficiently, a thorough analysis of the industry growth, market potential and opportunity should be conducted.

The main purpose of this thesis is to understand the DC industry in Europe and to explore the opportunities to expand it. In order to do so, an overview of the industry and how it came to function as it does today will be developed by investigating the reasons why the DC industry developed unevenly across Europe and uncovering what were the most important expansion deciding factors, which will help provide an industry guideline on the priorities that need to be considered in order for the best possible outcome to be reached. Learning the driving factors of the expansion of this industry by investigating factors that ensure suitability for local DC development will uncover the most significant strengths and weaknesses that need to be considered in the decision-making process.

The main objectives of this thesis will be to first determine which factors had the biggest impact in local market expansion by analysing the current state of the industry in Europe, as well as research the factors that influence the quality and functionality of a DC the most, conduct a systematic market selection process, determine which countries would be the most viable location candidates to develop the industry within and provide market entry recommendations for the selected countries.

This thesis will start by learning the current state of the DC industry in Europe and the factors that drove its development. Afterwards it will continue with learning which local factors make a certain country the best option on the international market and how to prioritize them. Lastly, it will rank the European countries for future development based on the learned factors.

Another aspect this thesis is going to consider is the construction of the DC building itself. Construction of DCs is a niched section of the construction field as this particular type of building has a multitude of specific design and technical requirements that must be considered to achieve quality operations. DC construction projects are complicated but precise and despite the fairly new entrance on the market, they are being built across the world at a fast rate in order to cover humanity's increasing need for cloud storage. Looking at construction specifics is an important aspect of selecting new potential DC locations as it impacts its functionality, cost, maintenance and operation.

Most DC construction projects are bid by international clients and built in a remote location, using teams and resources that are both local and imported, depending on specific needs. Based on the author's personal experience of working for 5 years in the DC construction industry representing both investing clients and construction crews, the research will begin starting from a personal hypothesis that the most important factors that need to be met by a good DC location are energy efficiency or access to renewable energy – as DCs consume a very high amount of power to function, it has a strong impact on its operating cost and its environmental effect; data security – as the building should be safe enough to store users' sensitive information; and high speed telecommunication network – as it is vital to be interconnected with the network.

The research methodology of this thesis begins with a theoretical exploration of market opportunity analysis, market research, and market entry strategies. Utilizing secondary data, including both raw and compiled sources, the author conducts a structured analysis of the European DC industry. This includes a literature review, secondary data analysis, and utilization of tools such as PESTEL analysis to assess political, economic, social, technological, ecological, and legal factors across European countries. Through quantitative and qualitative analysis, the methodology aims to identify potential expansion locations, focusing on equitable international expansion and considering regulatory environments, market demand, and local competitive advantages.

This thesis is divided into two main themes: a theoretical framework included in Chapters 2 and 3, as well as a practical application included in Chapters 5 to 7. Chapter 2 and 3 contain the theoretical framework and research conducted on existing literature on the main research topics: market opportunity analysis, market research and market entry strategy. Chapter 4 describes the research methodology. Chapter 5 explores research of the DC industry, including its history, development, expansion requirements and location selection. Chapter 6 contains a practical application of the preliminary screening and PESTEL analysis. Chapter 7 is a practical application of the market entry considerations applied for the expansion of the European DC industry. Lastly, Chapter 8 includes overall conclusions on the research, the practical application results, reflections on the initial hypothesis and information on the thesis's ethics and reliability.

2 RESEARCH ON MARKET OPPORTUNITY ANALYSIS

In today's globalized market, identifying profitable market opportunities is essential for businesses looking for sustainable growth and competitive advantage. Market opportunity analysis aims to guide organizations through the complex process of evaluating potential markets and making informed decisions about market entry and expansion strategies. This chapter provides an overview on market opportunity analysis, drawing insights from Doole, Lowe, and Kenyon's "International Marketing Strategy" (2017), Stevens, Sherwood, Dunn and Loudon's "Market Opportunity Analysis" (2002), Rana and Shrivastava's "Doing Business in Emerging Markets" (2007) and Hollensen's "Global Marketing" (2014).

2.1 Market segmentation

Market segmentation is a process that helps divide a large group of target markets into subgroups by using various relevant characteristics that could separate them into different segments with shared properties (Doole, Lowe & Kenyon 2017, 122-146). These characteristics could be based on geographic, demographic, product usage and product benefit factors (Stevens, Sherwood, Dunn & Loudon 2002, 32-38).

Market segmentation based on geographic factors divides markets by using geographical borders such as cities, regions or countries. Demographic segmentation is based on dividing factors such as age, gender, income or education. Segmentation by product usage patterns divides consumers between users and nonusers in order to determine usage rates. Segmentation by perceived benefits classifies consumers based on the sought benefits for a certain product or service. (Stevens, Sherwood, Dunn & Loudon 2002, 32-38.)

Market segmentation helps businesses assess a wide international market area in a time and cost efficient manner by indicating how to prioritize the targeted markets, as well as indicating the unique preferences of different customer segments (Doole, Lowe & Kenyon 2017, 122-146). Traditional segmentation criteria (measurability, accessibility, profitability and actionability) can also be effective (Stevens, Sherwood, Dunn & Loudon 2002, 32-38), however, contemporary segmentation strategies are proven to be more accurate thanks to using psychographic, behavioural and technographic criteria which refines the segmentation significantly (Doole, Lowe & Kenyon 2017, 122-146).

Psychographic segmentation is based on lifestyle dividing factors such as customer activities, interests and opinions (Doole, Lowe & Kenyon 2017, 122-146). Behavioural segmentation is based on patterns of consumption, brand loyalty, sought benefits and buying motivation (Hollensen 2014, 273-286). Technographic segmentation is based on patterns of consumers' technological adoption and usage, such as digital behaviours and preferences (Doole, Lowe & Kenyon 2017, 122-146).

The main type of market segmentation that businesses tend to use is based on geographical criteria. Once certain countries are selected, some businesses may choose to refine their market strategy by segmenting the market further based on additional national segmentation criteria such as demographic, economic and psychographic (Doole, Lowe & Kenyon 2017, 122-146). For a global market segmentation that considers country specific factors, Hassan and Stephen (2005) propose a hierarchical approach which recognizes both similarities and differences across markets, enabling business to develop cross-national market segments. This strategy focuses on consumer characteristics rather than countries, avoiding national stereotyping and acknowledging variations within nations (Doole, Lowe & Kenyon 2017, 122-146).

The implementation of a hierarchical segmentation strategy involves several stages starting with the assessment of the infrastructure availability and accessibility of the initial targeted global market. The viable countries are then compared based on favourable criteria related to market size and demand in order to determine mini-segments considering factors such as information search, behaviour and product preferences. Similarities across segments are then analysed from the perspective of various relevant factors and finally, cross-national segments are identified based on patterns of similar responses. (Doole, Lowe & Kenyon 2017, 122-146.)

Once market segments are determined and customer characteristics identified, market potential can be estimated in order to understand the expected sales volume on each segment. This process provides a quantitative measure that helps evaluate profitability and determine the viability of a specific market segment (Stevens, Sherwood, Dunn & Loudon 2002, 32-38).

Macro-oriented market screening tools, such as PESTEL analysis, help businesses assess external factors, including political, economic, social, technological, environmental and legal as it helps provide a list of issues that can influence the business strategy (Hollensen 2014, 273-286).

2.2 PESTEL analysis

One of the best tools that can be used for understanding the macro-environment and its trends is the PESTEL analysis, also known as STEEP, as it helps provide a list of issues that can influence the business strategy as well as identify factors that can potentially lead to changes based on the main macro-environmental factors, listed by Wheelen (2018, 129) as: political (indicators include the form of gov-ernment, political ideology, stability of government, attitude towards foreign companies, terrorist activity), economic (indicators include development, income, GDP trends, monetary and fiscal policies, currency, wages), social (indicators include customs, language, demographics, lifestyle, religion, attitude towards foreigners), technological (indicators include regulations, energy availability and cost, natural resources, transportation network), ecological (indicators include tax laws, regulations on foreign ownership of assets, trade regulations, foreign policies).

Political factors can influence a business's local and international operation through political pressures that impact the home country's international business expansion decisions due to political threats in the host markets, need of adaptation to new regulatory environments, taxation guidelines, political stability, trade regulations, labour laws and safety regulations. Governmental organizations' level of foreign business support also has a great impact on international expansion particularly through regulatory backing and financial assistance policies. Access to accurate information about the host markets' political environment is essential for an effective political preliminary screening. (Wheelen 2018, 127-156.)

Economic factors tend to impact a business's profitability due to the direct impact of economic issues in the host country with indicators such as gross domestic product (GDP), inflation, interest rates and credit availability. While developing countries tend to be overlooked in market opportunity assessments, according to the International Monetary Fund, even though they account for less than a third of the world's GDP, they also make up more than half of the global GDP growth. The economic development of Brazil, Russia, India and China (BRIC) has seen an important impact on the global economy, with China becoming the world's second-largest economy in 2007, India the primary location for outsourcing services, computer software and telecommunications, and Eastern Europe one of the main manufacturing suppliers for the European Union. (Wheelen 2018, 127-156.)

Socio-cultural factors help businesses understand how the needs of consumers are formed and what motivates them to purchase a service or product. Some of the main factors influencing customer preferences, purchasing decisions as well as package design and marketing are local culture, language, customs, values and norms (Doole, Lowe & Kenyon 2017, 78-104). Hollensen (2014, 244-263) defines culture as the learned ways in which a society understands, decides and communicates. Languages are an important component of culture and communication therefore understanding both spoken and silent languages is essential.

Communication techniques vary across different nations, with some languages being based strictly on words, while others taking into account surroundings, the social status of the message giver or other non-verbal elements. High-context cultures such as Japan and Arab countries rely more on indirect communication and prioritize relationship-building, while low-context cultures like those in Northern Europe prefer direct communication and individualism. (Hollensen 2014, 244-263.)

The sociocultural environment is also influenced by demographic changes, lifestyle trends and social dynamics, for instance, generational groups such as Millennials and Generation Z show different preferences when it comes to consumption patterns and brand perception (Doole, Lowe & Kenyon 2017, 78-104). Social changes, such as increasing urbanization and environmental awareness, shape the consumer's attitude towards sustainability, ethical sourcing and corporate social responsibility (Hollensen 2014, 244-263). Other cultural indicators such as religion, education, family and socio-economic factors tend to influence individual behaviour and decision-making in business settings as well.

Societal environments also have impact how multinational corporations (MNCs) conduct their activities, such as marketing, manufacturing and finance. For example, MNCs operating in Europe have lower labour productivity due to a shorter workweek and restrictions on laying off unproductive workers compared to other continents. This situation influences European based MNCs to expand their operations to other foreign countries where labour is cheaper and labour productivity is higher. Firms need to be flexible in order to accommodate the many differences in societal environments among countries. (Wheelen 2018, 129-150.) Ecological factors such as carbon footprint, climate change, extreme weather events or access to natural resources, have a great impact on potential markets as they can pose a wide variety of risks as well as opportunities. The ecological aspect of the environment has been changing, making it challenging for businesses to keep up with updating their policies to minimize their impact on the natural environment. Climate change has a significant impact on businesses as well, creating multiple types of risks which could be mainly categorized as: regulatory, supply chain, product and technology, litigation, reputational and physical risks. (Wheelen 2018, 129-150.)

Regulatory risks apply to compliance with regulations related to environmental issues. The Kyoto Protocol's first commitment period required 37 industrialized countries and the European Community to reduce greenhouse gas (GHG) emissions to about 5% against 1990 levels, while the second commitment period required the members to commit to reduce emissions by at least 18% below 1990 levels from 2013 to 2020. The European Union has an emissions trading program that allows companies that surpass a specific GHG emission level to buy additional allowances from companies whose emissions are lower than the allowed level. Businesses can also earn credits for their emissions. The risk of litigation rises for companies that produce large quantities of carbon emissions as they face the possibility of lawsuits. (Wheelen 2018, 129-150.)

Supply chain risk is another category of risk that companies face due to suppliers passing carbon related expenses enforced by governmental regulations onto their clients. Global supply chains are also at risk due to the increasing frequency of major storms and flooding, which continues to cause problems for ports due to rising sea levels. Additionally, transportation costs have also been increasing due to the scarcity of fossil-based fuel. (Wheelen 2018, 129-150.)

The risk associated with product and technology is that environmental sustainability is starting to enforce multiple regulatory conditions on businesses. A study conducted by Environics revealed that 60% of U.S. participants are more likely to purchase products and services from companies that are conscious of their impact on the environment. (Wheelen 2018, 129-150.)

The risk of litigation is mainly faced by companies that produce substantial carbon emissions. An example of this risk is the lawsuit filed against oil and gas companies in the federal district court of Mississippi, claiming that businesses such as tabaco, oil and gas companies were responsible for contributing to the severity of hurricane Katrina through their high GHG emissions. (Wheelen 2018, 129-150.)

Reputational risk is another concern for companies as their impact on the environment can have a significant effect on the overall image that customers perceive them through. Negative perceptions related to climate change can put the value of a company's brand at risk. On the other hand, companies with a good record of environmental sustainability may have a competitive advantage in attracting and retaining loyal consumers, employees and investors. (Wheelen 2018, 129-150.)

Climate change is a direct physical risk through droughts, floods, storms and rising sea levels, which have resulted in the melting of glaciers and an increase in sea levels by 2.5cm per decade due to a two to three degrees Celsius increase in average Arctic temperatures over the past 50 years. (Wheelen 2018, 129-150.)

Legal factors such as industry regulation policies, permits required to operate or employment laws tend to impact both industry competition levels and the effectiveness of business strategies. High taxation rates and labour laws in Western Europe influence companies to change their competitive strategies or relocate to other countries. For example, due to Germany's high labour and tax costs, German companies need to compete in the high quality products market or move their manufacturing operations to lower cost countries. Governmental bureaucracy can also create regulations with high impact on business operation profitability, making some countries more viable than others. (Wheelen 2018, 129-150.)

A company that operates in different geographic regions deals with a wide variety of unique societal environments that include a diverse range of sociocultural, technological, economic, ecological, political and legal variables. The societal environment can vary so much from one region to another that the business must be highly adaptable in its internal environment and strategic management process in order to thrive. Successful international expansion depends on adapting to local market conditions, building relationships with key stakeholders and maintaining adaptability in response to the evolving market dynamics.

2.3 Use of secondary data

Various sources including public libraries, electronic databases and internet resources provide valuable information for preliminary market screening. However, technological progress revolutionized international market research by offering easy access to online databases of secondary information. These da-

tabases have a significant advantage over printed data sources as they facilitate access to updated information which can easily be accessed from anywhere in the world. Some of the main advantages of online databases of secondary data include real time updates, cost effectiveness, 24/7 accessibility and targeted information retrieval. The consolidation of global market information into these databases allows researchers to conduct comprehensive studies and access critical information for international marketing strategies with more ease than ever before (Doole, Lowe & Kenyon 2017, 132-133).

Oftentimes businesses need to determine whether an opportunity is worth chasing without investing a large quantity of resources upfront. Employing a "quick and dirty assessment" (Stevens, Sherwood, Dunn & Loudon 2002, 38-40) approach can provide valuable insights early in the evaluation process by combining the census of business, the census of selected services, of wholesale trade, of manufacturing, of population and other censuses. These databases can be used to estimate the number of customers, the market size, sales potential and competitiveness, helping businesses reach a preliminary understanding of the market. Using available secondary data can help businesses estimate market potential and make sales projections, however, it is important to cross-validate estimates using alternative calculations and considering factors such as costs per capita to achieve a more accurate assessment (Stevens, Sherwood, Dunn & Loudon 2002, 38-40).

While online databases offer a wide range of advantages, ensuring the credibility and validity of the data remains the researcher's responsibility. Despite the multitude of benefits, researchers need to evaluate the reliability of the online sources that they are using in order to mitigate the risk of misinformation. While secondary data is valuable in international marketing research, several challenges must be considered due to variations in data quality and consistency across countries. One of the main issues may be the availability of quality secondary information, which can vary significantly between countries. Differences in units of measurement and data collection methodologies as well as political considerations can impact the accuracy of the data leading to discrepancies in the reported values. (Doole, Lowe & Kenyon 2017, 132-133.)

Another disadvantage of using secondary data is potentially working with outdated information which may mislead researchers and affect the relevance of their findings. Developing countries are known to be the most likely to face inconsistencies in data classification and shortage of reliable data. Despite these limitations, secondary data remains a valuable source, especially for smaller businesses that lack resources for primary research. Online databases offer access to real time information for international

10

market research, increasing the efficiency of the decision-making processes. By understanding and addressing the limitations of secondary data, businesses can use it to assess global markets effectively. (Doole, Lowe & Kenyon 2017, 132-133).

3 RESEARCH ON MARKET ENTRY STRATEGY

Expanding a business internationally can be a great opportunity for any organization, but it requires careful planning and analysis. Choosing the right market and understanding the opportunities it presents can make the difference between success and failure. This chapter will discuss the importance of market entry strategy, mainly based on Svend Hollernse's "Global Marketing" (2014), Thomas Wheelen's "Strategic Management and Business Policy" (2018), Doole, Lowe, and Kenyon's "International Marketing Strategy" (2017), Stevens, Sherwood, Dunn and Loudon's "Market Opportunity Analysis" (2002) and Rana and Shrivastava's "Doing Business in Emerging Markets" (2007).

3.1 Market entry strategy

Market entry strategies include a range of approaches for entering new markets, each with its own advantages, risks and considerations. Once a certain foreign country has been selected for expansion, the business has to decide which is the best way to enter its market by assessing each entry modes' advantages and disadvantages and deciding which are best suited for its type of activity, as well as which are best suited for the internal environment of said foreign country. The main types of entry modes are export modes, intermediate modes (contractual) and hierarchical modes (investment). (Hollensen 2014, 336.)

Export modes are the most focused on externalizing the business which gives the least amount of control over the new investment. Business can be practiced indirectly by using a middle man, or directly by remotely controlling the sales, most often with the use of the internet. This mode's advantage is a high level of flexibility and low risk, while its disadvantage is a low level of control. The intermediate modes are mainly contractual such as setting up licenses, franchises, joint ventures or strategic alliances. This particular split ownership setup means that the control and risk are shared but the flexibility is low which is the main disadvantage. The hierarchical modes are used when the company is willing to make a substantial investment in the new market by making acquisitions or establishing new wholly owned subsidiaries. This mode's advantages come from the full internalization which means it has the highest level of control compared to the other entry modes. The disadvantages are high risk and low flexibility. (Hollensen 2014, 332-339.) The simplest form of market entry is export, which involves selling products or services to customers in foreign markets (Doole, Lowe & Kenyon 2017, 167). Exporting offers low entry barriers and allows companies to test international waters with minimal investment. However, it may limit control over distribution and expose firms to currency fluctuations and trade barriers.

Licensing and franchising involve granting foreign entities the right to use intellectual property, trademarks, or business models in exchange for royalties or fees (Hollensen 2014, 332-339). This approach enables rapid market entry while minimizing investment and operational risks. Joint ventures and strategic alliances create partnerships with local firms to leverage their market knowledge, resources and local distribution networks (Stevens, Sherwood, Dunn & Loudon 2002, 3-7). Joint ventures allow companies to share risks and resources while gaining access to local expertise and infrastructure. However, they require careful negotiation and management of relationships to ensure the alignment of both parties' objectives and strategies.

Direct investment, such as greenfield investments or acquisitions, involves establishing a physical presence in foreign markets through the construction of new facilities or the acquisition of existing businesses (Rana & Shrivastava 2007, 97-115). Direct investment offers full control over operations and enables firms to gain a larger share of the value chain, however, it requires substantial investment, assumes higher risks and involves facing foreign regulation policies and cultural differences.

3.2 Market entry considerations

According to Root [2020, 332], there are three rules a business can use in order to determine the best suited entry mode: the naïve rule, pragmatic rule and strategy rule. The Naïve rule is a standard procedure a company can have for all market entries that uses the same business model every time and doesn't adapt to the local environment. This requires the lowest amount of research but also gives the lowest success rate. The Pragmatic rule is that the company adapts entry mode based on each foreign market entry with the caveat that not all possible alternatives are being investigated in order to lower the amount of research needed. The Strategy rule has the highest success rate as it evaluates all possible options which reduces possible risks significantly and gives the highest level of understanding on the local market. (Hollensen 2014, 332-339.)

Hollensen categorizes other factors influencing the choice of market entry mode between internal factors (firm size, international experience), desired mode characteristics (risk averse, control, flexibility), transaction specific factors (tacit nature of know-how, opportunistic behaviours) and external factors (sociocultural distance between home country and host country, market size and growth, intensity of competition etc). (Hollensen 2014, 332-339.)

Internal factors play a crucial role in determining the mode of entry for firms when expanding into international markets. One key factor is the size of the firm, as larger firms have greater resource availability and are better able to make high resource commitments to foreign markets. Contrarily, SMEs are more likely to use export modes as they do not have the resources to achieve a high degree of control over their international operations. As the firm grows, it will increasingly use the hierarchical model.

The physical characteristics of the product or service are also important in order to determine where production is located and what entry mode is most suitable. For example, products with high value or weight ratios are typically used for direct exporting, while soft drinks and beer companies typically establish licensing agreements or invest in local bottling or production facilities due to prohibitive shipment costs. The nature of the product affects entry mode selection because products widely vary in their use and characteristics. (Hollensen 2014, 332-339.)

When a firm is considering entering a foreign market, it must take into account various external factors that can influence its entry. One such factor is the sociocultural distance between the home country and the host country. Sociocultural similarities between the two countries can lead to similar business and industrial practices, a similar language, educational level and cultural characteristics. Sociocultural differences can create internal uncertainty for the firm and influence the mode of entry desired by that firm. The greater the perceived distance between the home and host country in terms of culture, economic systems and business practices, the more likely it is that the firm will shy away from direct investment in favour of low risk entry modes like agents or importers.

Another factor to consider is country risk or demand uncertainty. Foreign markets are generally riskier than the domestic market, and a firm's exposure to risk is a function of the market itself and its method of involvement. When country risk is high, a firm should limit its exposure by lowering its resource commitments in that particular market. This means that firms will favour entry modes that involve relatively low resource commitments, such as export modes.

The unpredictability of the political and economic environment of the host market can increase the perceived risk and demand uncertainty experienced by the firm, which in turn makes firms less inclined to enter the market with entry strategies that require high resource commitments. The size and growth rate of the country are also key factors to consider when determining the mode of entry. The larger the country and the size of its market and the higher the growth rate, the more likely management will be to commit resources to its development and to consider establishing a wholly owned sales subsidiary or participating in a majority-owned joint venture. (Hollensen 2014, 332-339.)

The business also needs to consider the flexibility associated with a given mode of entry. Hierarchical modes are typically the least flexible and most difficult to change in the short run. Contractual agreements and joint ventures limit the firm's ability to adapt or change strategy when market conditions are changing rapidly.

Another important factor to consider is the tacit nature of the firm-specific know-how. When the knowledge transferred is tacit, it is challenging to articulate, making the drafting of a contract problematic. The difficulties and costs involved in transferring know-how motivates firms to use hierarchical modes, which are better able to facilitate the intraorganizational transfer of knowledge. Therefore, the greater the tacit component of firm specific know-how, the more a firm will favour hierarchical modes. (Hollensen 2014, 332-339.)

3.3 Marketing mix

The concept of the marketing mix originates from the microeconomic theory of the single P (price) (Chong, 2003) that was later introduced as the "marketing mix" by McCarthy (1964, 60) commonly known today as the "4Ps": product, price, place and promotion. While the "4Ps" are not a scientific theory, the marketing mix is a conceptual framework for the main decisions that managers need to make in order to align their product or service with customer needs.

These tool helps develop both long-term and short-term strategies (Palmer 2004) since its introduction in the 1940s, the marketing mix has been significantly influencing marketing and research (Grönroos, 1994, 4-20). Kent (1986, 145-154) refers to the "4Ps" as "the holy quadruple…of the marketing faith…written in tablets of stone".

The product mix component focuses on developing a product or service offering that meets the needs of the targeted market through its physical attributes, features, quality, packaging and branding. Businesses can create competitive advantage by identifying the main selling point of the product or service that differentiates it from similar products and highlighting the features that bring innovation for the potential customer. (Kent 1986, 145-154.)

The pricing mix component refers to the monetary value of the product or service. Setting the right price is essential for achieving profitability and attracting customers. Factors such as production costs, competition, the target market's price sensitivity and perceived value must be considered in pricing strategies in order to include cost-based pricing, value-based pricing, skimming, penetration pricing, or promotional pricing. (Grönroos 1994, 4-20.)

The place mix component refers to the distribution channels and methods used to make the product available to the target market. This involves determining the right locations, outlets and intermediaries to reach customers effectively. Aspects such as logistics, inventory management and must be considered in the distribution strategies in order to include direct selling, wholesaling, retailing, or e-commerce. (Constantinides 2006, 57-76.)

The promotion mix component includes all the activities that are meant to promote the service or product to the target market. This can be done through advertising, public relations, sales promotion, personal selling and direct marketing efforts. An integrated promotional mix should be developed in order to create awareness, generate interest, stimulate desire and encourage action from potential customers. (Kent 1986, 145-154.)

While the marketing mix is a valuable concept for the marketing decision making-process, Constantinides (2006, 57-76) notes some limitations, pointing out that the traditional pillars of marketing management have faced substantial criticism including the lack of consideration for customer behaviour, customer interaction, theoretical content and personification of marketing activities.

However, by carefully analysing and managing each component of the "4Ps", businesses can create a comprehensive marketing strategy that aligns with customer needs while helping their organizations make strategic decisions when introducing new products or services on the market.

4 RESEARCH METHODOLOGY

This chapter provides an overview on the research methodology that was chosen for accomplishing the thesis goals and it describes the research design, approach, data collection and analysis methods. The detailing aims to offer transparency on how the research was conducted. This thesis is built on the hypothesis that, since the DC industry appeared in Europe, it has been developing unevenly across its countries and in order to continue expanding in an efficient, equitable way, one must first understand which factors have influenced the decision making processes that led to its current state. To achieve this goal, the research questions that act as a structuring component are: 1. What factors explain the development and current state of the European data center industry today?; 2. What are the most influential deciding factors that make a market viable for a DC and how should they be ranked?; 3. Which European countries are viable for continuing the DC industry expansion according to the identified factors?; 4. What entry strategies should be employed for the countries identified as the most viable markets for industry expansion?

In order to find the answers to these questions, the thesis started with a theoretical research on market opportunity analysis, market research and entry strategies where the thesis author learned about market trends and how to understand their regulatory environment, how to recognize the cultural, social, geographic, political, technological and ecological aspects of markets, as well as how to identify gaps on the international market and how to evaluate the feasibility and viability of entering a chosen market.

The author's research methodology is a comprehensive approach that uses secondary data to analyse market opportunities and select potential expansion locations for the European DC industry. It begins with a thorough literature review and analysis of secondary data aiming to understand the current state of the industry and identify relevant factors influencing expansion by creating illustrative comparisons of various statistical numbers versus the current numbers of built DCs in each European country and drawing dedicated conclusions. Utilizing the PESTEL tool, the researcher conducts a detailed assessment of political, economic, social, technological, ecological, and legal factors across the European countries. Drawing on statistical data from reputable sources (Eurostat, IEA, Global Economy, OECD, Statista, Statistics Finland, The World Bank, World Data, World Population Review), ranked tables are created summarizing each country's performance across these factors, facilitating comparison and identification of favourable conditions for industry growth. Through a market opportunity analysis,

countries with the highest potential for local expansion are selected, with a focus on achieving equitable international expansion by excluding the already well developed DC markets. Finally, the researcher provides considerations on potential market entry strategies for the selected countries, taking into account regulatory environments, market demand and local competitive advantages. Overall, this methodology combines quantitative and qualitative analysis to guide strategic decision-making in the future expansion of the European DC industry.

The main type of data that this thesis is based on is secondary data including both raw and compiled data (Saunders 2006). The benefits of conducting secondary research for foreign markets include being less costly and timely than conducting research abroad, as well as not requiring contacts outside of the home country, and providing objectivity since the author is not constrained by overseas customs. Additionally, secondary research was used as a preliminary stage in the market screening process to quickly generate background information and eliminate countries from consideration (Hollensen 2014). The use of secondary data allowed comparison of various sets of data and facilitated the opportunity to compare findings within existing datasets (Saunders 2006). By reanalysing and comparing secondary data, the author was led to unexpected discoveries of patterns thanks to which she was able to link sets of statistical data in order to reach conclusions on the reasons why the DC industry has developed more in some countries than others.

Ensuring the credibility of research findings requires focusing on two main elements: validity and reliability. According to Yin (2009), validity refers to the soundness of a concept, conclusion, or measurement, ensuring accurate alignment with the real world. On the other hand, reliability, as defined by Yin (2009), relates to the consistency or stability of a measurement or instrument. Therefore, a study achieves validity when it effectively measures its intended target and reliability when it consistently produces similar outcomes over time. The assurance of these qualities is essential for establishing the credibility and applicability of research outcomes as researchers are tasked with designing and executing their studies to guarantee both the accuracy and trustworthiness of their results.

Other field authorities agree with the significance of validity and reliability. According to Cohen, Manion, and Morrison (2018), "Validity and reliability are critical to the scientific foundation of educational research". They explain that sustaining validity involves pinpointing the constructs to be measured and selecting suitable methods and measures, while maintaining reliability requires the use of uniform methods and procedures to achieve consistent results. Similarly, Saunders, Lewis & Thornhill (2016, 202) state that "Validity and reliability are essential to any research design". They stress that ensuring validity means formulating research questions that precisely reflect the study's objectives, while guaranteeing reliability involves maintaining consistency in data collection and analysis.

While the author ensured the validity and reliability of the data sources, the main disadvantages of using secondary data are the lack of guarantee of the dataset quality and the possibility of the information being outdated since it is often collected in previous years. The use of both qualitative and quantitative data supports the recognition that secondary data drawn from various documents and statistical data sets offers a comprehensive understanding of the complexity of the DC industry. The author acknowledges the significance of multiple-source secondary data, understanding that it involves the combination of distinct data sets prior to access. The thesis searches various documentary sources including compilations of statistical source adaptations of Eurostat, Statista and The World Bank.

To conclude, the author acknowledges that assurance of research validity and reliability is essential for credible and applicable results. Researchers must conscientiously execute their studies to ensure the dependability of their findings. Through this approach, they contribute to advancing knowledge in their respective fields and making meaningful societal contributions.

5 DATA CENTER INDUSTRY OVERVIEW

A data center (DC) is a physical room, building or facility that houses IT infrastructure used for building, operating and delivering applications as well as for storing and managing the data associated with those applications (IBM 2024). The DC industry is a rapidly growing sector (Tech News And Updates 2023) that encompasses the design, construction, operation and maintenance of DCs used by a wide range of organizations, including businesses, governments and institutions, in order to support a variety of applications and services, such as websites, online applications, cloud computing, big data analytics and the Internet of Things (IoT) (De Sousa 2018). DCs are essential for the growing demand for cloud storage services and the increasing reliance on digital technologies.

The DC industry is characterized by various trends including cloud computing, edge computing, sustainability and security (Stackscale 2024). The adoption of cloud computing has increased the demand for DC services, as more organizations are outsourcing their computing and storage needs to cloud service providers (Howard 2024). The growth of the IoT and the increasing need for real-time data processing has led to the emergence of edge computing, which involves placing DCs closer to the sources of data, such as factories, retail stores and other locations (Gillis 2023).

The DC industry is one of the main drivers of today's technological innovation and economic growth and it is expected to continue to expand in the upcoming years as the demand for services that rely on data storage continue to grow. The European industry has been steadily growing due to an increasing reliance on digital technologies, adoption of cloud computing, sustainability trends, data security, machine learning and artificial intelligence. According to estimates, the European DC market is expected to reach over \$50 billion in value by 2025 (Gooding 2024).

However, there is a growing concern about sustainability within the DC industry as the buildings require a large amount of electricity in order to operate which leads to a high GHG emission level (Device42 2023). Sustainable practices focus on reducing energy consumption and carbon emissions by adopting renewable energy sources, such as wind and solar power, to cover their energy consumption (Directorate-General for Energy 2024). Another concern is data security as it is a critical factor considering the potential sensibility of the information that a DC may be storing. The demand for measures such as encryption and firewall protection continue to grow in order to prevent potential cyber threats against the DC (Fortinet 2023). Demands for implementing regulations that ensure data protection such as The General Data Protection Regulation (GDPR) are also on the rise as they enforces strict rules for the protection of personal data in the EU (ManageEngine 2023).

5.1 History and development

The term 'data center' was first used in 1990 when IT operations began to expand fast enough that the structures and materials needed to build one became affordable enough for investors to agree that building a special computer room was a good business decision. In spite of that, the first DC-type building ever created was in the 1940s at the University of Pennsylvania, created for military purposes and later on in the 1960s to support large mainframe computers, which were used by governments, banks, and other organizations to process large amounts of data. These DCs were typically large, specialized facilities that were expensive to build and operate. Since then, they have evolved as technology has advanced and the demand for data storage and processing has grown (Team Evoque 2021).

As personal computers and other devices became more powerful and widely available, the need for DCs increased, and they began to be used to host servers and other equipment that supported business operations and provided access to information and services over the internet (Team Evoque 2021). Today, DCs come in a wide range of sizes and configurations, from large facilities that occupy entire buildings, to smaller, more specialized centers that are designed to support specific types of applications or services. With the rise of cloud computing, DCs have also become more distributed, with multiple centers located in different regions around the world to provide better performance and availability for users.

DCs are important because they provide the infrastructure needed to store, process, and manage large amounts of data. They are the backbone of the internet, as they host the servers and other hardware that power websites, online applications, and other services that are accessed over the internet (Faist Group 2023). In addition to hosting servers and other hardware, DCs also provide networking, storage, and security infrastructure, as well as power and cooling systems to ensure that the equipment can operate

reliably and efficiently. They are essential for supporting the growing demand for data services, such as cloud computing, big data analytics, and the IoT.



FIGURE 1. Number of data centers in Europe by country 2021 (adapted from Statista 2021)

Thanks to today's technology and international investment demand, a DC can be built just about anywhere and provide services that benefit not only local users but also international. This sort of flexibility is bound to facilitate industrial expansion but looking at the current situation in Europe, represented in Figure 1, it can be observed that there is a large difference in the development level from country to country, with the number of existing DCs varying between 487 and 1. The 3 leading countries as of 2021, Germany (487 DCs), UK (456) and Netherlands (281) house 42.4% of the entire European DC industry.

5.2 Expansion requirements assessment

It is relevant to understand why the current expansion of the industry is outspread unevenly across Europe. According to Figure 1, no striking pattern can be observed at first glance – no relation to country surface, geographical location or population density. For a more in depth analysis, secondary data will be researched using external regional and country sources for a set of relevant factors, as well as proxy indicators and compare them to the existing number of DCs for each European country by creating comparative illustrations. In order to proportionally categorise each value, a scale from 1 to 10 will be used, dividing the range of values into 10 equal intervals and assigning a scale value of 1 to the interval that includes the smallest assessed values, and a scale value of 10 for the interval that includes the largest assessed values.

5.2.1 Geographical surface

The first factor considered relevant to compare with the number of existing DCs today, is the geographical surface of each country. According to the research and as illustrated in Figure 2, no direct corelation can be observed between the number of existing DCs and the country surface, as the country holding the highest numbers of DCs Germany (487 DCs across 357, 386 km2) has a value of 10/10 on the DC scale and 6/10 on the surface scale; and the largest country, Russia, (172 DCs across 3,968,200 km2) has a value of 4/10 on the DC scale and 6x10/10 on the surface scale. Another relevant consideration is that Netherlands (281 DC across 21,198km2) has the lowest value point on the surface scale (1/10) while being the third most developed industry with a 6/10 on the DC scale and holding 9,74% of the entire European industry. The rest of the countries show no regular pattern but the fact that climate and economic factors may be relevant are observed therefore, they will be analysed as well.

(456 DC) UK (242,495 km2) (281 DC) Netherlands (41,198 km2) (264 DC) France (551,695 km2) (172 DC) Russia* (3,968,200 km2) (136 DC) Poland (312,685 km2) (131 DC) Italy (301,318 km2)	
(281 DC) Netherlands (41,198 km2) (264 DC) France (551,695 km2) (172 DC) Russia* (3,968,200 km2) (136 DC) Poland (312,685 km2) (131 DC) Italy (301,318 km2)	
(264 DC) France (551,695 km2) (172 DC) Russia* (3,968,200 km2) (136 DC) Poland (312,685 km2) (131 DC) Italy (301,318 km2) (125 DC) Service (406,511 km2)	
(172 DC) Russia* (3,968,200 km2) (136 DC) Poland (312,685 km2) (131 DC) Italy (301,318 km2)	
(136 DC) Poland (312,685 km2) (131 DC) Italy (301,318 km2) (125 DC) Service (408 511 km2)	
(131 DC) Italy (301,318 km2)	
(125 DC) Secie (408 511 Jan 2)	
(125 DC) Spain (498,511 km2)	
(104 DC) Switzerland (41,290 km2)	
(81 DC) Sweden (450,295 km2)	
(74 DC) Belgium (30,510 km2)	
(66 DC) Austria (83,858 km2)	
(57 DC) Ukraine (603,628 km2)	
(49 DC) Ireland (70,273 km2)	
(38 DC) Denmark (44,493 km2)	
(37 DC) Norway (385,178 km2)	
(34 DC) Finland (338,145 km2)	
(33 DC) Czech Republic (78,866 km2)	
(33 DC) Turkey (23,764 km2)	
(24 DC) Bulgaria (110,994 km2)	
(24 DC) Latvia (64,589 km2)	
(24 DC) Slovenia (20,273 km2)	
(26 DC) Romania (238,397 km2)	
(17 DC) Lithuania (65,300 km2)	
(17 DC) Portugal (88,416 km2)	
(13 DC) Hungary (93,030 km2)	
(13 DC) Slovakia (49,036 km2)	
(13 DC) Greece (131,940 km2)	
(9 DC) Belarus (207,600 km2)	
(8 DC) Croatia (56,594 km2)	
(8 DC) Serbia (88,361 km2)	
(6 DC) Iceland (102,775 km2)	
(6 DC) Macedonia (25,713 km2)	
(6 DC) Moldova (33,846 km2)	
(5 DC) Cyprus (9,251 km2)	
(5 DC) Georgia (2,642 km2)	
(4 DC) Malta (316 km2)	
(2 DC) Bosnia and Herzegovina (51,129 km2)	
(1 DC) Albania (28,748 km2)	_
(1 DC) Azerbaijan (6,960 km2)	

*Russia's surface would be significantly larger than the maximum 10 value used for this graphic (more than 6 times larger than the upper bound of the final interval) and the illustration does not reflect the right proportion.

FIGURE 2. Number of data centers in Europe by country 2021 compared to country surface (adapted from Statista 2021 & The World Bank 2020)

5.2.2 Population

The next factor compared with the numbers of DCs per country is the total population per country. The illustration resulted from the proportional value comparison (APPENDIX 1) does not show a relevant pattern as the country holding the highest numbers of DCs (Germany, 487 DCs to 83,196 thousands of people) has a value of 10/10 on the DC scale and 6/10 on the population scale, while the country with the largest population (Russia, 172 DCs to 143,449 thousands of people) has a value of 4/10 on the DC scale and 10/10 on the population scale. Again Netherlands (281 DC across 17,533 thousands of people) has a low value point on the population comparison scale (2/10) while being the third most devel-

oped industry in European. The rest of the countries show no regular pattern either but population density may be relevant as it could impact real estate local pricing and operational security aspects so these factors are going to be compared as well.

The values of population density per country compared to the values of existing data centers per country (APPENDIX 2), do not show a relevant pattern either, as the country holding the highest numbers of DCs (Germany, 487 DCs and 238 people/km2) has a value of 10/10 on the DC scale and 2/10 on the density scale, while the country with the largest population density (Malta, 4 DCs and 1610 people/km2) has a value of 1/10 on the DC scale and 10/10 on the population scale. The rest of the countries show no regular pattern either but other population related factors may be relevant to compare such as access to electricity and internet use.

5.2.3 Access to electricity

The population percentage with access to electricity illustration (APPENDIX 3) shows no relevant pattern either, as all European countries with existing data centers have electrical access in a proportion of 100%. In this case we are going to follow up on the statistics of internet use as it could be a better indicator of local need for cloud storage, as well as on the quality and pricing of electricity as they play a major role in operation cost and functionality of DCs.

5.2.4 Internet usage

The illustration comparing the percentage of population using internet with DC numbers (APPENDIX 4), does not show a relevant pattern as the comparison reveals both very high and very low values on the internet use scale for countries with either many or few DCs. The country holding the highest number of DCs (Germany, 487 DCs and 91% of population using internet) has a value of 10/10 on the DC scale and 10/10 on the internet use scale; while the country with the largest use of internet (Iceland, 6 DCs and 100% of population using internet) has a value of 1/10 on the internet use scale. Italy (131 DC, 75% internet use), holding the lowest value on the internet use scale (1/10) is the 7th most developed European industry which further indicates that local internet use is not a strong influence on local DC development.

5.2.5 Local factors conclusions

So far we have learned that local factors such as surface, population, population density and population access to electricity and internet did not play a major role in deciding where the DC industry was going to expand, which leads us to believe that international market factors are more relevant than local factors within this industry. In order to test this assumption, a comparison of the yearly average temperature of each country will pe performed as literature suggests cool temperatures are one of the biggest impacting factors for long term savings on energy use (Lettiere 2021).

5.2.6 Average temperature

Assessing Figure 3, we may observe that the countries holding the highest numbers of DCs do have a low yearly average comparing to the ones with higher temperatures and the extreme temperatures (both high and low) are not preferred but rather the moderately low ones. The country holding the highest numbers of DCs (Germany, 487 DCs and 8.4C) has a value of 10/10 on the DC scale and 6/10 on the average temperature scale, while the country with the lowest temperature average (Russia, 172 DCs and -5.35C) has a value of 4/10 on the DC scale and 1/10 on the temperature scale and the country with the highest temperature average (Malta, 4 DCs and 19.2C) has a value of 1/10 on the DC scale and 10/10 on the temperature scale. To conclude, the countries with high numbers of existing DCs show a pattern of moderately low temperatures, below 13C yearly average and about 6/10 value on the comparison scale.

(487 DC)	Germany	(8.4 C)	
(456 DC)	UK	(8.45 C)	
(281 DC)	Netherlands	(9.25 C)	
(264 DC)	France	(10.7 C)	
(172 DC)	Russia	(-5.35 C)	
(136 DC)	Poland	(7.85 C)	
(131 DC)	Italy	(13.45 C)	
(125 DC)	Spain	(13.30 C)	
(104 DC)	Switzerland	(5.10 C)	
(81 DC)	Sweden	(1.75 C)	
(74 DC)	Belgium	(9.55 C)	
(66 DC)	Austria	(6.35 C)	
(57 DC)	Ukraine	(8.3 C)	
(49 DC)	Ireland	(9.3 C)	
(38 DC)	Denmark	(-3.7 C)	
(37 DC)	Norway	(1.5 C)	
(34 DC)	Finland	(1.55 C)	
(33 DC)	Czech Republic	(7.55 C)	
(33 DC)	Turkey	(9.9 C)	
(24 DC)	Bulgaria	(10.55 C)	
(24 DC)	Latvia	(5.6 C)	
(24 DC)	Slovenia	(8.9 C)	
(26 DC)	Romania	(8.8 C)	
(17 DC)	Lithuania	(6.2 C)	
(17 DC)	Portugal	(15.15 C)	
(13 DC)	Hungary	(9.75 C)	
(13 DC)	Slovakia	(6.8 C)	
(13 DC)	Greece	(15.4 C)	
(9 DC)	Belarus	(6.15 C)	
(8 DC)	Croatia	(10.9 C)	
(8 DC)	Serbia	(10.55 C)	
(6 DC)	Iceland	(-0.7 C)	
(6 DC)	Macedonia	(9.8 C)	
(6 DC)	Moldova	(9.45 C)	
(5 DC)	Cyprus	(18.45 C)	
(5 DC)	Georgia	(5.65 C)	
(4 DC)	Malta	(19.2 C)	
(2 DC)	Bosnia and Herzegovina	(9.85 C)	
(1 DC)	Albania	(11.4 C)	
(1 DC)	Azerbaijan	(11.95 C)	

FIGURE 3. Number of data centers in Europe by country 2021 compared to average yearly temperature (1961-1990) (adapted from Statista 2021 & Mitchell, Carter, Jones, Hulme & New 2003)

5.2.7 Electric power stability

Another highly relevant aspect that may reveal a pattern of importance of international factor comparison with high impact on DC functionality is the likelihood of electric power transmission and distribution losses as energy loss is considered to be one of the biggest threats to DC operation. As illustrated in Figure 4, there is indeed a tendency of low risk of energy loss (under 10% in the leading DC number countries) and high risk (up to 24%) in the countries with the lowest industry presence. The country holding the highest numbers of DCs (Germany, 487 DCs shows a 4% risk) has a value of 10/10 on the DC scale and 1/10 on the energy loss scale and 10/10 on the energy loss scale.

(487 DC)	Germany	(4%)	
(456 DC)	UK	(8%)	
(281 DC)	Netherlands	(5%)	
(264 DC)	France	(6%)	
(172 DC)	Russia	(10%)	
(136 DC)	Poland	(6%)	
(131 DC)	Italy	(7%)	
(125 DC)	Spain	(10%)	
(104 DC)	Switzerland	(7%)	
(81 DC)	Sweden	(5%)	
(74 DC)	Belgium	(5%)	
(66 DC)	Austria	(5%)	
(57 DC)	Ukraine	(11%)	
(49 DC)	Ireland	(8%)	
(38 DC)	Denmark	(6%)	
(37 DC)	Norway	(6%)	
(34 DC)	Finland	(4%)	
(33 DC)	Czech Republic	(5%)	
(33 DC)	Turkey	(15%)	
(24 DC)	Bulgaria	(9%)	
(24 DC)	Latvia	(9%)	
(24 DC)	Slovenia	(5%)	
(26 DC)	Romania	(11%)	
(17 DC)	Lithuania	(22%)	
(17 DC)	Portugal	(10%)	
(13 DC)	Hungary	(12%)	
(13 DC)	Slovakia	(2%)	
(13 DC)	Greece	(8%)	
(9 DC)	Belarus	(9%)	
(8 DC)	Croatia	(12%)	
(8 DC)	Serbia	(15%)	
(6 DC)	Iceland	(3%)	
(6 DC)	Macedonia	(20%)	
(6 DC)	Moldova	(22%)	
(5 DC)	Cyprus	(4%)	
(5 DC)	Georgia	(6%)	
(4 DC)	Malta	(5%)	
(2 DC)	Bosnia and Herzegovina	(8%)	
(1 DC)	Albania	(24%)	
(1 DC)	Azerbaijan	(14%)	
· /	<u>-</u>	(

FIGURE 4. Number of data centers in Europe by country 2021 compared to electric power transmission and distribution losses (percentage of output in 2014) (adapted from Statista 2021 & The World Bank 2018)

5.2.8 Risk of catastrophic events

Another highly relevant comparison value is the risk of catastrophic events which is going to be analysed using the percentage of population who suffered from disaster events during 1990-2009. In Figure 5, we can observe a pattern of higher risk in countries where the industry was least developed, with Albania (1 DC and 5.3% risk) showing a value of 1/10 on DC scale and 10/10 on risk scale, as opposed to leading DC industry in Germany (487 DC and 0.0% risk) showing a value of 10/10 on DC scale and 11/10 on DC scale and 11/

(487 DC)	Germany	(0.0)					
(456 DC)	UK	(0.0)					
(281 DC)	Netherlands	(0.0)					
(264 DC)	France	(0.0)					
(172 DC)	Russia	(0.1)					
(136 DC)	Poland	(0.0)					
(131 DC)	Italy	(0.0)					
(125 DC)	Spain	(0.7)					
(104 DC)	Switzerland	(0.0)					
(81 DC)	Sweden	(0.0)					
(74 DC)	Belgium	(0.0)					
(66 DC)	Austria	(0.0)					
(57 DC)	Ukraine	(0.3)					
(49 DC)	Ireland	(0.0)					
(38 DC)	Denmark	(0.0)					
(37 DC)	Norway	(0.0)					
(34 DC)	Finland	(0.0)					
(33 DC)	Czech Republic	(0.2)					
(33 DC)	Turkey	(0.1)					
(24 DC)	Bulgaria	(0.0)					
(24 DC)	Latvia	(0.0)					
(24 DC)	Slovenia	(0.0)					
(26 DC)	Romania	(0.1)					
(17 DC)	Lithuania	(0.0)					
(17 DC)	Portugal	(0.0)					
(13 DC)	Hungary	(0.1)					
(13 DC)	Slovakia	(0.0)					
(13 DC)	Greece	(0.0)					
(9 DC)	Belarus	(0.0)					
(8 DC)	Croatia	(0.0)					
(8 DC)	Serbia	(0.0)					
(6 DC)	Iceland	(No data)					
(6 DC)	Macedonia	(0.3)					
(6 DC)	Moldova	(0.3)					
(5 DC)	Cyprus	(0.0)			i i		
(5 DC)	Georgia	(0.8)					<u> </u>
(4 DC)	Malta	(No data)					
(2 DC)	Bosnia and Herzegovina	(0.5)					
(1 DC)	Albania	(5.3)					
(1 DC)	Azerbaijan	(1.1)					
				_		 	 _

FIGURE 5. Number of data centers in Europe by country 2021 compared to risk of droughts, floods, extreme temperatures (% of population, average 1990-2009) (adapted from Statista 2021 & The World Bank 2009)

5.2.9 GDP

We continue by checking if economic factors such as GDP per capita, real estate costs, taxes, and electrical rates will also prove to show relevant international patterns and will continue by looking at these comparisons starting with GDP, illustrated in Figure 6. We can observe a clear correlation between the lower half of GDP values and lower half of existing DC industry, with the lowest GDP per capita in Moldova (6 DCs and 5,230.7 GDP) showing a value of 1/10 on DC scale and 1/10 on GDP scale, while the leading DC industry in Germany (487 DCs and 51,203.6 GDP) shows a 10/10 DC scale value and 5/10 GDP scale value. On average, the countries show a direct correlation with an average GDP value for a high DC value, but there are a few exceptions such as Ireland which holds the highest GDP per capita (100,172.1 USD) translated to a value of 10/10, and only 49 DCs, translated to a value of 1/10, or Russia which holds a low GDP (12,194.8 USD) translated to a value of 1/10 but holding the 5th place on current local expansion scale.



FIGURE 6. Number of data centers in Europe by country 2021 compared to GDP per capita in 2021 (adapted from Statista 2021 & The World Bank 2020)

5.2.10 Electricity rate

Another relevant economic factor is the local rate for electricity which is going to be analysed using current statistics on electricity prices for non-household customers including taxes. The Figure 7 compares these statistics and the current DC industry spread-out and it does not show a direct correlation as countries with a developed industry have average to above average electricity rates, while countries with the lowest rates do not show a high value for industry development. The country holding the
highest number of DCs (Germany, 487 DCs and 0.18 euro per KWh) has a value of 10/10 on the DC scale and 6/10 on the energy rate scale and the country with the lowest rate (Ukraine, 57 DCs and 0.03 euro per KWh) has a value of 2/10 on the DC scale and 1/10 on the energy rate scale. The conclusion is that an average rate seems to be the tendency but it does not seem to be a priority on the overall locating decision.

(487 DC)	Germany	(0.18)	
(456 DC)	UK	(0.20)	
(281 DC)	Netherlands	(0.17)	
(264 DC)	France	(0.13)	
(172 DC)	Russia	(0.06)	
(136 DC)	Poland	(0.15)	
(131 DC)	Italy	(0.25)	
(125 DC)	Spain	(0.19)	
(104 DC)	Switzerland	(0.22)	
(81 DC)	Sweden	(0.11)	
(74 DC)	Belgium	(0.19)	
(66 DC)	Austria	(0.16)	
(57 DC)	Ukraine	(0.03)	
(49 DC)	Ireland	(0.21)	
(38 DC)	Denmark	(0.16)	
(37 DC)	Norway	(0.15)	
(34 DC)	Finland	(0.08)	
(33 DC)	Czech Republic	(0.16)	
(33 DC)	Turkey	(0.19)	
(24 DC)	Bulgaria	(0.17)	
(24 DC)	Latvia	(0.12)	
(24 DC)	Slovenia	(0.16)	
(26 DC)	Romania	(0.23)	
(17 DC)	Lithuania	(0.18)	
(17 DC)	Portugal	(0.12)	
(13 DC)	Hungary	(0.15)	
(13 DC)	Slovakia	(0.22)	
(13 DC)	Greece	(0.30)	
(9 DC)	Belarus	(0.09)	
(8 DC)	Croatia	(0.16)	
(8 DC)	Serbia	(0.10)	
(6 DC)	Iceland	(0.13)	
(6 DC)	Macedonia	(0.10)	
(6 DC)	Moldova	(0.08)	
(5 DC)	Cyprus	(0.25)	
(5 DC)	Georgia	(0.08)	
(4 DC)	Malta	(0.14)	
(2 DC)	Bosnia and Herzegovina	(0.07)	
(1 DC)	Albania	(0.10)	
(1 DC)	Azerbaijan	(0.04)	
()		(0.0.1)	

FIGURE 7. Number of data centers in Europe by country 2021 compared to electricity prices for nonhousehold customers including taxes (euro per KWh in 2022) (adapted from Statista 2021 & Eurostat 2022)

5.2.11 Profit tax rate

Assuming that local taxes can have a high impact on the operation cost, the DC numbers will be compared to the current values of commercial profit tax. Looking at Figure 8, we see a slight correlation as while the top 3 countries within the industry have average to high tax rates, we can see a tendency for below average tax rates in the rest of the well developed countries. However, the general tendency is that countries with developed industry have both high and low tax scale values, while countries with the lowest tax don't show a high value for industry development, with the exception of France (265 DCs and 0.2% tax) which may not be relevant on its own as France showed favourable indicators on many of the other factors we have assessed so far (such as low service interruption or hazard risk, high GDP, moderate climate and average electrical rate). The country holding the highest numbers of DCs (Germany, 487 DCs and 23.2% tax) has a value of 10/10 on the DC scale and 8/10 on the commercial profit tax scale; and the country with the lowest tax (Croatia, 8 DCs and 0.0% tax) has a value of 1/10 on the DC scale and 1/10 on the tax scale.

(487 DC)	Germany	(23.2%)		
(456 DC)	UK	(16.6%)		
(281 DC)	Netherlands	(20.4%)		
(264 DC)	France	(0.2%)		
(172 DC)	Russia	(7.2%)		
(136 DC)	Poland	(14.5%)		
(131 DC)	Italy	(14.6%)		
(125 DC)	Spain	(10.6%)		
(104 DC)	Switzerland	(9.3%)		
(81 DC)	Sweden	(13.1%)		
(74 DC)	Belgium	(10.3%)		
(66 DC)	Austria	(17.1%)		
(57 DC)	Ukraine	(10.2%)		
(49 DC)	Ireland	(12.4%)		
(38 DC)	Denmark	(17.1%)		
(37 DC)	Norway	(20.0%)		
(34 DC)	Finland	(12.1%)		
(33 DC)	Czech Republic	(5.2%)		
(33 DC)	Turkey	(20.0%)		
(24 DC)	Bulgaria	(4.9%)		
(24 DC)	Latvia	(7.8%)		
(24 DC)	Slovenia	(12.7%)		
(26 DC)	Romania	(15.6%)		
(17 DC)	Lithuania	(5.9%)		
(17 DC)	Portugal	(12.5%)		
(13 DC)	Hungary	(9.4%)		
(13 DC)	Slovakia	(9.1%)		
(13 DC)	Greece	(23.0%)		
(9 DC)	Belarus	(11.1%)		
(8 DC)	Croatia	(0.0%)		
(8 DC)	Serbia	(13.0%)		
(6 DC)	Iceland	(8.5%)		
(6 DC)	Macedonia	(11.0%)		
(6 DC)	Moldova	(8.4%)		
(5 DC)	Cyprus	(8.3%)		
(5 DC)	Georgia	(7.8%)		
(4 DC)	Malta	(32.3%)		
(2 DC)	Bosnia and Herzegovina	(8.4%)		
(1 DC)	Albania	(14.1%)		
(1 DC)	Azerbaijan	(12.7%)		

FIGURE 8. Number of data centers in Europe by country 2021 compared to commercial profit tax (percentage in 2019) (adapted from Statista 2021 & The World Bank 2021)

5.2.12 PPP

In order to compare the DC industry to the level of international economic growth, we are going to look at the purchase power parity (PPP) for each country, which per Figure 9, reveals a pattern of average values (4-6 on the 1-10 scale) in the countries with the highest local industry development, while countries with very high PPP (8 to 10 values on the 1-10 scale) or very low PPP (1 to 2 values on the 1-10 scale) have a low existing development. The country holding the highest numbers of DCs (Germany, 487 DCs and 116.2 PPP) has a value of 10/10 on the DC scale and 5/10 on the PPP scale. The country with the lowest PPP (Bulgaria, 24 DCs and 36 PPP) has a value of 1/10 on the DC scale and 1/10 on the PPP scale. The country with the highest PPP (Switzerland, 104 DCs and 196.7 PPP) has a value of 3/10 on the DC scale and 10/10 on the PPP scale.



FIGURE 9. Number of data centers in Europe by country 2021 compared to purchasing power parities (PPPs) (price level indices in 2021) (adapted from Statista 2021 & Eurostat 2023)

5.2.13 Conclusions

In conclusion, the expansion of the DC industry across Europe seems to have been mostly influenced by a low risk of energy loss, moderately low temperature, low risk of catastrophic natural events, average-high GDP/capita and average PPP up until today.

5.3 Selecting the right location for a data center

We have learned that factors such as low risk of energy loss or natural disasters, moderately low temperature, average international values for PPP, taxes and GDP, have influenced the development of the European industry so far. We are going to continue the research by learning which factors are considered critical when selecting a new site. Selecting the right location for a DC requires careful consideration of a number of factors, such as the availability of power and telecommunications infrastructure, the cost of land and labour, the regulatory environment, and the risks and opportunities associated with different locations (Carter & Hines 2019). The key criteria that organizations should consider when selecting the location for a DC and how these criteria can impact the performance and cost of the DC will be next examined.

According to current trends and literature, the critical factors influencing site selection are mainly labour costs (analysing the cost of skilled labour in potential locations and assessing the availability of a qualified workforce to meet DC operational needs), accessibility (evaluating proximity to major airports and highways for efficient logistics and recognizing the importance of easy access during both construction and operation phases), real estate options (investigating the availability and cost of real estate options, deliberating the choice between retrofitting existing facilities and constructing purposebuilt data centers, and considering advantages and disadvantages of each option), telecommunications infrastructure (ensuring access to a robust telecommunications infrastructure for seamless connectivity, emphasizing the significance of high quality network connectivity for DC operations, analysing the presence of competitive markets in telecommunications), utilities costs (analysing the cost of utilities, specifically electricity and water, exploring access to renewable energy sources to promote sustainability, assessing the role of utility costs in total operational expenses), tax and regulatory climate (understanding the tax and regulatory environment in potential locations, identifying incentives or tax breaks that may apply to data center operations, evaluating the impact of regulations on data center construction and operation) and proximity to suppliers (assessing the proximity of suppliers for specialized equipment and parts, recognizing the value of a local supply chain for reduced lead times and cost efficiency, analysing the impact of supplier location on supply chain resilience) (Gigerich 2012).

Another aspect a business should consider when selecting a site location is whether to choose a retrofit or purpose-built project. While retrofit sites have the advantage of lower building cost and timeframe, utilizing vacant land offers the advantage of freely customizing the project to meet any preventive measures that may need to be taken due to various factors such as climate, natural hazards or specific client requirements. Additionally, it is easier to design a purpose-built data center (PBDC) properly equipped for high security standards and IT aspects such as high density equipment, easy facility navigation and efficient computer cooling systems (Carter & Hines 2019).

The research will continue with the assessment of the factors identified by theoretical research from the perspective of the author's work experience within the industry.

5.3.1 Energy efficiency

Seeing as DCs are virtually computer stacks, they are bound to consume a very high amount of power, not only for functioning, but also for computer cooling and cloud disconnection prevention, therefore, one of the most important design aspects to be considered is energy efficiency. Energy efficiency in buildings refers to the use of design, construction, and technology solutions to reduce the amount of energy needed to operate a building. The goal of energy efficiency in buildings is to reduce energy consumption, lower energy costs, and reduce greenhouse gas emissions, while maintaining or improving the comfort, convenience, and functionality of the building. In 2022, DCs were estimated to consume 3-4% of world's electricity (Iron Mountain 2023) which is extremely high comparing to other industries. The main mitigation practices within the industry include designing the building with economical energy consumption solutions, choosing a location with access to renewable energy and choosing a location with low electrical rates.

To achieve an energy-efficient design, a harmonious blend of creativity and innovative technology is required. A great example of such a design is currently operational in Hamina, Finland, where a DC was constructed on the site of a former paper mill by the Baltic Sea. The engineering team behind this project focused on refurbishing the mill's original pumps and pipes to channel cold seawater directly

into the computer rooms, naturally cooling the air containment. This clever design significantly reduced energy consumption while ensuring zero adverse environmental effects, as the water is simply filtered and returned to the sea (Baxtel 2017).

In addition to their significant power consumption, DCs cannot tolerate any electrical disruptions once they are operational. That is why most projects are designed with two separate energy pathways: a primary and a backup, with at least one of them being uninterruptible. The engineering and maintenance of these energy supplies can increase the already high electrical costs and environmental impact, making the utilization of renewable energy sources a preferred choice. This approach has the benefit of mitigating both issues long term. DCs are known to be major contributors to pollution, with a carbon footprint surpassing that of the entire airline industry as of 2022 (Monserrate 2022).

Although the first two options are preferable due to their environmental benefits and long-term advantages, the process of selecting an appropriate site can be challenging and time-consuming due to the scarcity of the necessary criteria required for such a design. Another solution for reducing the longterm operational cost of a DC is to select a building location in an area with lower electrical rates compared to other competing sites, but this solution does not address the sustainability issues.

5.3.2 Data center construction cost

DCs experience substantial operational costs when compared to other industrial buildings. However, it is crucial to take into account the construction expenses as well since the initial capital investment ranks among the highest in today's tech industries. In addition to standard building materials, the inclusion of specialized electrical and telecom equipment, along with shipping costs (which can vary based on the building's geographical location), significantly contribute to the overall budget, accounting for more than 60% of the total investment. Moreover, labour costs tend to be above average due to the requirement for field specialists, often requiring international outsourcing.



PICTURE 1. Initial capital investment to construct and equip a data center (Gigerich 2012)

The construction cost of a DC project can be categorized in four main parts: electrical systems (40-45% of the cost), HVAC/mechanical systems (15-20% of the cost), building fit-out (20-25% of the cost) and land and shell (15-20% of the cost) (DgtlInfra 2022).

Typically, the analysis of the amount is conducted per square meter and megawatts (MW), given the variable scale of buildings ranging from 1 to 50 MW on average. A proportional assessment offers specific advantages for buildings. Large-scale projects benefit from cost reductions as a result of procuring a substantial quantity of HVAC and electrical equipment. However, a drawback is that the increased power density in large projects requires a corresponding rise in cooling capacity, leading to higher long-term costs.

The project schedule can play a crucial role in influencing costs. In many cases, these projects prioritize rapid construction, often commencing when only 60% of the design is complete. While this approach saves time, it frequently results in conflicts during the detailed design phase, leading to rework and resource inefficiencies. Building such projects at a slower pace has been demonstrated to result in cost savings.

When considering an international investment, the differences between building costs should not take decisional priority as the real estate value only sums up to 5% on average and has minimal impact on the total budget. The human resources factor is similar as the construction teams are most often made

up from various international specialists with little differences in labour pay. According to current research, some intercontinental differences can be seen in the total building cost with North America rated at approximatively 9.5m/MW and \$1,000/sqft, Europe rated at 14m/MW and \$1,200/sqft, Asia-Pacific rated at 12m/MW and \$1,000/sqft and South America rated at 10m/MW and no data for sqft (Gigerich 2012). Based on these values we may conclude that the lowest building cost is on average in the United States while the highest one is in Europe.

5.3.3 Risk of natural disasters and weather events

When selecting a building location, it is crucial to consider local weather conditions and the risk of natural disasters, especially for DCs since computers are sensitive to factors like humidity, high temperatures, earthquakes, or any disruptive natural events. Service disruption is a significant threat that requires careful planning. In regions with extreme climates, adverse weather can impact the construction schedule from an early stage, therefore the building schedule should account for seasonal changes brought by the effects of heat, frost, or rain throughout each construction phase until the structure is weatherproof. Construction may then proceed indoors. Projects need to be meticulously planned based on the specific natural risks they might encounter. Additional measures, such as computer rack bracing in seismic areas, extra cooling in drought-prone regions, and structural reinforcement for potential destabilizing events, must be incorporated as well.

5.3.4 Access to infrastructure and supplies

Efficient business operations rely on site accessibility, including several crucial factors such as DC efficiency in order to ensure optimal performance, quality power supply that guarantees a reliable and high-quality source of power, construction facilitation for ease of construction processes and logistics, and attraction for IT talent recruitment in order to create an appealing environment for recruiting top IT talent.

It is essential for the potential location to have a high quality pre-existing telecommunications infrastructure, making markets with high competition among service carriers optimal. Such locations are known for cost-effective distribution systems, reliable conductivity, and low disruption in fiber network communication. Energy supply is also critical as it impacts the operation efficiency. It is often measured by the cost per watt of consumed electricity and some locations can reach levels as high as 1,000W/square meter consumed. Therefore, access to a supplier capable of providing the required energy capacity is an essential consideration (Carter & Hines 2019).

Access to hard infrastructure is crucial as well, especially during the construction phase, where a continuous influx of building materials and equipment is required within tight schedules. Proximity to a major airport is advantageous for site visits by international team members. Site accessibility is important not only for current operations but also for future business expansion. Additionally, it plays an essential role in preventing man-made hazards by enabling access for emergency service vehicles.

Another important consideration is supplier proximity to the DC, as its need for specialized equipment can often become a procurement challenge. Having a network of part suppliers nearby can reduce costs, lead times for sourcing parts, and mitigate the risk of supply chain disruptions. Local supply chains also contribute to sustainability by minimizing carbon emissions associated with transportation. Moreover, local suppliers are likely to have a better understanding of the specific needs and requirements of the DC thanks to their knowledge of local codes and regulations, which increases the reliability of the provided parts.

5.3.5 Conclusions

Selecting an optimal site for building a DC should prioritize: access to a stable energy supplier, minimal risks of natural disasters and a naturally cool climate. The access to a quality electric infrastructure is essential for an efficient DC as it avoids potential service disruption caused by power outages. Challenges caused by natural disasters such as floods, earthquakes or tornadoes could be mitigated through design features, however, it would still present a risk and an increased cost compared to other markets where this type of risk is minimal. A naturally cool environment is preferred due to its direct impact on minimizing cost and energy consumption used by computer cooling which is one of the main power consumers in a DC.

6 PRELIMINARY SCREENING AND PESTEL ANALYSIS FOR DATA CENTERS IN EU-ROPE

In this chapter the theoretical framework described in Chapters 2 and 3 will be applied while considering the findings on the DC industry specifics described in Chapter 5. The purpose of this data triangulation is to determine which European countries are the most favourable for local DC industry expansion in the most data backed way. The systematic market selection process begins per Root's strategy rule by considering all countries within the geographical segment of the European continent as potential markets. To conduct a preliminary screening, the PESTEL elements will be employed to eliminate markets that are least likely to be favourable potential locations.

6.1 Political preliminary screening

To conduct the initial preliminary screening of the European market, the PESTEL method will be used in order to gather background information on each country and exclude from further consideration the ones with elevated levels of risk or high lack of data transparency, as well as identify the markets with the most favourable environment for local future expansion of the DC industry. The research objectives will be aligned with the location selection criteria outlined in Chapter 5 based on the findings.

The preliminary market screening will be initiated by analysing indexes related to the political environment of all the European countries. The objective is to understand the local levels of political stability, assess potential risks, and understand the regulatory environments of each country, in order to identify which markets are optimal for a potential DC investment and which are not.

6.1.1 Political stability index

In order to evaluate and rank the European countries for market opportunities for future DC industry expansion from a political perspective, the process will be initiated by examining the political stability index. This index represents the average data available from 1996 to 2021, on a scale from -2.5 to 2.5 where '-2.5' indicates the weakest political stability level and '2.5' the strongest (Global Economy 2021). The political stability index evaluates governmental stability by analysing the likelihood of its

overthrow through unconstitutional means or political violence, including terrorism. The index is determined by averaging indexes from the 'Economist Intelligence Unit,' 'World Economic Forum,' and 'Political Risk Services' (Global Economy 2021).

This index is relevant to the DC industry because it could directly impact the safety and security of a DC, as well as indicate the ease of conducting business in a specific location. A country with high political stability is less likely to face civil unrest, political disruptions or government instability (Global Economy 2021). A politically stable country is more likely to have a predictable and transparent legal and regulatory framework.

TABLE 1. Ranking of European countries by Political Stability Index in 2021 (Adapted from GlobalEconomy 2021)

	Political Stability (from highest to lowest)											
1	1 Liechtenstein 1.64 18 Hungary		Hungary	0.86		35	Greece	0.15				
2	Andorra	1.63	19	Ireland	0.86		36	North Macedonia	0.12			
3	Iceland	1.37	20	Lithuania	0.82		37	Albania	0.11			
4	Luxembourg	1.21	21	Estonia	0.76		38	Serbia	-0.13			
5	Switzerland	1.13	22	Germany	0.76		39	Montenegro	-0.15			
6	Norway	1.1	23	Slovenia	0.76		40	Moldova	-0.21			
7	Sweden	1.03	24	Croatia	0.71		41	Kazakhstan	-0.25			
8	Finland	0.98	25	Belgium	0.61		42	Bosnia and Herzegovina	-0.38			
9	Malta	0.97	26	Italy	0.58		43	Georgia	-0.42			
10	Czechia	0.96	27	Spain	0.58		44	Russian Federation	-0.65			
11	Latvia	0.96	28	Slovak Republic	0.56		45	Belarus	-0.74			
12	Denmark	0.95	29	United Kingdom	0.54		46	Armenia	-0.84			
13	Portugal	0.95	30	Romania	0.53		47	Azerbaijan	-0.85			
14	Netherlands	0.92	31	Poland	0.51		48	Turkiye	-1.1			
15	Austria	0.91	32	Bulgaria	0.46		49	Ukraine	-1.1			
16	Monaco	0.91	33	Cyprus	0.44		50	Kosovo	no data			
17	San Marino	0.91	34	France	0.37							

Based on this ranking it is concluded that Serbia, Montenegro, Moldova, Kazakhstan, Bosnia and Herzegovina, Georgia, Russia, Belarus, Armenia, Azerbaijan, Turkey, Ukraine and Kosovo should be eliminated from consideration, while the leading markets are Lichtenstein, Andorra, Iceland, Luxembourg, Switzerland, Norway, Sweden, Finland, Malta and Czech Republic.

6.1.2 Restrictions on foreign investments

The European countries can also be evaluated based on restrictions on foreign investments, utilizing an OECD index that assesses limits on foreign ownership, preliminary market screening requirements and restrictions on foreign personnel and operational freedom. The index is based on a 0 to 1 value with 0 indicating fewer restrictions and 1 indicating more. Foreign investments are defined as the acquisition of assets or engagement in business operations by an entity from one country in another country (OECD 2002). These investments include stock purchase in a foreign company, real estate investment, or establishing a subsidiary abroad.

When evaluating a potential market for local future expansion of the DC industry, analysing the local restrictions on foreign investments is important because they can have a strong impact on the types of businesses permitted for foreign entities, limitations on the percentage of ownership foreign entities can hold in local companies, and requirements for local partnerships or joint ventures. These restrictions can potentially increase the difficulty and cost for a foreign company to build a DC in a particular country. Moreover, insight into the local laws and regulations is essential in order to ensure compliance with local laws, regulations, and taxation. It also helps recognize potential risks, including political instability, corruption, or other legal and regulatory issues that could impact the DC operations.

Due to the substantial number of countries with unavailable data, no countries were eliminated from consideration based on this index, a conclusion solely being drawn on the leading markets per the ranking illustrated in Appendix 5, which are: UK, Ireland, Netherlands, Germany, Denmark, Belgium, Italy, France, Greece and Sweden.

6.1.3 Difficulty index level for foreign controlled enterprises to invest locally

Data from the "OECD National Treatment for Foreign-Controlled Enterprises" (OECD 2002) was utilized in order to evaluate the political restrictions on foreign-controlled enterprises. Initially, the European countries were categorized by membership, and subsequently, examined by the specific restrictions applicable to each country. Based on the potential impact relevant specifically to a potential DC project, the countries were then ranked from the least to the most restrictive as potential markets for DC industry expansion. Restrictions on foreign controlled enterprises can include regulations on the types of businesses that foreign entities are allowed to engage in, as well as limits on the percentage of ownership they can have in a local company, and requirements for local partnerships or joint ventures. These restrictions can make it more difficult or costly for a foreign entity to establish a DC in the corresponding country, and may also limit the potential for future growth or expansion. A breakdown of the restrictions per country can be seen in Appendix 6.

Based on the resulted ranking (APPENDIX 6) it is concluded that Albania, Andorra, Armenia, Azerbaijan, Belarus, Bosnia and Herzegovina, Bulgaria, Croatia, Cyprus, Georgia, Kosovo, Lichtenstein, Malta, Moldova, Monaco, Montenegro, Macedonia, Russia, San Marino and Serbia should be eliminated from consideration, while the leading markets are Czech Republic, Romania, Portugal, Slovakia, Poland, Latvia, Ireland, Italy, Belgium, Slovenia, Hungary, Netherlands, Luxembourg and Norway.

6.1.4 Corruption index

The corruption index is measured on a 0 to 100 scale where 100 indicates the highest risk. It is a crucial factor to evaluate when considering a potential market for DC construction because it can impact the overall business environment, the ease of conducting business, and the potential for encountering legal and regulatory issues. (Global Risk Profile 2022.)

Corruption is defined as the misuse of power for personal gain and it manifests in various forms such as bribery, embezzlement, and nepotism. A high corruption index suggests a higher likelihood of corruption in a country, making business operations more challenging due to potential arbitrary and discriminatory actions by government officials, along with an increased likelihood of legal and regulatory issues (Global Risk Profile 2022). Moreover, corruption can increase business costs as companies may be compelled to pay bribes or engage in illicit payments to government officials, creating difficulties in competing with other DCs in the region.

	Corruption index (from least to most)											
1	Norway	7,12	18	8 Slovenia 24,64 35 Serbia		45,04						
2	Finland	7,14	19	Czechia	25,89		36	Moldova	49,19			
3	Sweden	8,8	20	Latvia	25,91		37	Kazakhstan	49,28			
4	Denmark	10,98	2	Poland	28,1		38	Turkiye	51,47			
5	Estonia	11,13	22	Italy	29,83		39	Russian Federation	51,88			
6	Netherlands	13,74	23	Greece	31,23		40	Ukraine	51,95			
7	Iceland	15,23	24	Cyprus	32,31		41	Albania	52,36			
8	Ireland	15,48	25	Malta	32,98		42	Belarus	53,16			
9	United Kingdom	15,5	20	Slovak Republic	34,21		43	Kosovo	53,61			
10	Germany	16,56	27	Georgia	34,57		44	Bosnia and Herzegovina	54,09			
11	Belgium	18,7	28	Croatia	35,47		45	Azerbaijan	56,06			
12	Austria	18,77	29	Bulgaria	37,61		46	Andorra	no data			
13	France	18,87	30	Armenia	38,8		47	Liechtenstein	no data			
14	Switzerland	18,87	3	Romania	38,89		48	Luxembourg	no data			
15	Lithuania	21,79	32	Hungary	40,89		49	Monaco	no data			
16	Portugal	al 22,22 33 North Macedon		North Macedonia	42,55		50	San Marino	no data			
17	17 Spain 22,66 34 Monter		Montenegro	44,41								

TABLE 2. Ranking of European countries by corruption in 2022 (Adapted from Global Risk Profile2022)

Based on this ranking it is concluded that Moldova, Kazakhstan, Turkey, Russia, Ukraine, Albania, Belarus, Kosovo, Bosnia and Herzegovina, Azerbaijan, Andorra, Liechtenstein, Luxembourg, Monaco and San Marino should be eliminated from consideration, while the leading markets are Norway, Finland, Sweden, Denmark, Estonia, Netherlands, Iceland, Ireland, UK and Germany.

6.1.5 Security threat index

The following index measures security threats faced by a country, including factors such as bombings, attacks, battle related deaths, rebel movements, mutinies, coups, and terrorism. It also considers significant criminal elements, including organized crime and homicides, as well as the perceived trust of citizens in domestic security (Global Economy 2022). A higher value on this index indicates an increased threat which is relevant to a DC because it provides an understanding of potential security risks and it allows DCs to assess the need for security measures and sensitive data protection.

TABLE 3. Ranking of European countries by security threats index in 2022 (Adapted from GlobalEconomy 2022)

		S	Security Threat Index (from least to most)										
1	Portugal	0.3	18	Belgium	2.3		35	Albania	4.5				
2	Slovenia	0.3	19	Croatia	2.3		36	Bosnia & Herz.	4.7				
3	Iceland	0.4	20	Czechia	2.3		37	Italy	4.8				
4	Luxembourg	0.4	21	Latvia	2.3		38	Moldova	4.8				
5	Slovakia	0.9	22	Ireland	2.4		39	Belarus	5.3				
6	Denmark	1.4	23	Sweden	2.4		40	Georgia	5.3				
7	Norway	1.5	24	Lithuania	2.5		41	Armenia	5.4				
8	Austria	1.7	25	Spain	3.1		42	Azerbaijan	6.1				
9	Switzerland	1.7	26	Cyprus	3.2		43	Ukraine	6.4				
10	Estonia	2	27	France	3.2		44	Turkey	6.9				
11	Germany	2	28	UK	3.2		45	Russia	7.8				
12	Hungary	2.1	29	Greece	3.6		46	Andorra	no data				
13	Malta	2.1	30	Kazakhstan	3.7		47	Kosovo	no data				
14	Netherlands	2.1	31	Bulgaria	4		48	Liechtenstein	no data				
15	Poland	2.1	32	Serbia	4.2		49	Monaco	no data				
16	Finland	2.2	33	Montenegro	4.3		50	San Marino	no data				
17	Romania	2.2	34	North Macedonia	4.4								

Based on this ranking it is concluded that Belarus, Georgia, Armenia, Azerbaijan, Ukraine, Turkey, Russia, Andorra, Kosovo, Lichtenstein, Monaco and San Marino should be eliminated from consideration, while the leading markets are Portugal, Slovenia, Iceland, Luxembourg, Slovakia, Denmark, Norway, Austria, Switzerland, Estonia and Germany.

6.1.6 Political preliminary screening conclusion

Based on the political preliminary market screening it is concluded that the countries with either the least amount of available data or the highest political instability would present the highest potential risk to the development of a DC industry and they are: Azerbaijan, Belarus, Kosovo, Russia, Andorra, Armenia, Bosnia and Herzegovina, Georgia, Lichtenstein, Moldova, Monaco, San Marino, Turkey and Ukraine. The countries that offer the highest degree of political stability are: Norway, Denmark, Germany, Iceland, Ireland, Luxembourg, Netherlands and Sweden.

6.2 Legal preliminary screening

Indexes related to the legal frameworks and conditions in the European countries will be screened next. The objective is to analyse various jurisdictions, identifying potential challenges and opportunities that may impact the potential future expansion of the local DC industry.

6.2.1 Trade freedom index

The trade freedom index is calculated by averaging the trade-weighted tariffs and non-tariff barriers and it measures the ease of trade within a country by taking into account both tariffs (taxes on imported goods) and non-tariff barriers (such as regulations, customs procedures, and government intervention in trade) on a 0 to 100 scale with 100 being the highest level of freedom (Global Economy 2022).

Analysing a country's trade freedom index is relevant when considering building a DC in that country because it can provide insight into the ease of doing business locally. A high trade freedom index indicates that the country has relatively low tariffs and minimal non-tariff barriers, which can make it easier and less costly to import the necessary equipment and supplies for the DC. Additionally, a high trade freedom index may indicate that the country has a more stable business environment, which can be beneficial for a foreign company looking to invest in building a DC.

TABLE 4. Ranking of European countries by trade freedom in 2022 (Adapted from Global Economy2022)

	Trade Freedom Index (from highest to lowest)											
1	Georgia	87	18 Germany		79		35	Ukraine	79			
2	Switzerland	87	19	Greece	79		36	North Macedonia	78			
3	Norway	85	20	Hungary	79		37	Moldova	77			
4	Spain	84	21	Ireland	79		38	Serbia	77			
5	United Kingdom	84	22	Italy	79		39	Belarus	76			
6	Albania	83	23	Latvia	79		40	Turkey	76			
7	Iceland	81	24	Lithuania	79		41	Kazakhstan	75			
8	Austria	79	25	Luxembourg	79		42	Armenia	74			
9	Belgium	79	26	Malta	79		43	Azerbaijan	67			
10	Bulgaria	79	27	Montenegro	79		44	Bosnia and Herzegovina	69			
11	Croatia	79	28	Netherlands	79		45	Russia	69			
12	Cyprus	79	29	Poland	79		46	Andorra	no data			
13	Czechia	79	30	Portugal	79		47	Kosovo	no data			
14	Denmark	79	31	Romania	79		48	Liechtenstein	no data			
15	Estonia	79	32	Slovakia	79		49	Monaco	no data			
16	Finland	79	33	Slovenia	79		50	San Marino	no data			
17	France	79	34	Sweden	79							

Based on this ranking it is concluded that North Macedonia, Moldova, Serbia, Belarus, Turkey, Kazakhstan, Armenia, Azerbaijan, Bosnia and Herzegovina, Russia, Andorra, Kosovo, Lichtenstein, Monaco and San Marino should be eliminated from consideration, while the leading markets are Georgia, Switzerland, Norway, Spain, UK, Albania and Iceland.

6.2.2 Taxes on the income or profits of corporations

The analysis of corporate income or profit taxes is based on the latest available data of the overall taxto-GDP ratio, calculated as the sum of taxes and net social contributions into a percentage of gross domestic product (Eurostat 2020). Evaluating this percentage is relevant when assessing a potential DC location as it offers valuable insights into the overall business climate and the associated costs of operating in that specific country. A high percentage can increase overall construction and maintenance costs as a result of increased operational costs. Additionally, a high tax rate may discourage foreign investment, making it more challenging for a foreign company to build a DC in that specific country.

TABLE 5. Ranking of European countries by taxes on the income or profits of corporations in 2019 as% of GDP (Adapted from Eurostat 2020, 5)

	Taxes on Income or Profit of Corporations (least to most)										
1	Latvia	0.2	18	France	2.8	35	Azerbaijan	no data			
2	Hungary	1.2	19	Slovak Republic	3	36	Belarus	no data			
3	Lithuania	1.6	20	Sweden	3	37	Bosnia and Herzegovina	no data			
4	Estonia	1.8	21	Denmark	3.1	38	Georgia	no data			
5	Italy	1.9	22	Ireland	3.1	39	Kazakhstan	no data			
6	Bulgaria	2	23	Portugal	3.1	40	Kosovo	no data			
7	Slovenia	2	24	Switzerland	3.1	41	Liechtenstein	no data			
8	Iceland	2.1	25	Czechia	3.3	42	Moldova	no data			
9	Romania	2.1	26	Belgium	3.7	43	Monaco	no data			
10	Spain	2.1	27	Netherlands	3.7	44	Montenegro	no data			
11	Greece	2.2	28	Malta	5.7	45	North Macedonia	no data			
12	Poland	2.2	29	Norway	5.8	46	Russian Federation	no data			
13	Croatia	2.4	30	Cyprus	5.9	47	San Marino	no data			
14	United Kingdom	2.4	31	Luxembourg	5.9	48	Serbia	no data			
15	Finland	2.5	32	Albania	no data	49	Turkiye	no data			
16	Germany	2.7	33	Andorra	no data	50	Ukraine	no data			
17	Austria	2.8	34	Armenia	no data						

Based on this ranking it is concluded that Denmark, Ireland, Portugal, Switzerland, Czech Republic, Belgium, Netherlands, Malta, Norway, Cyprus, Luxembourg and all the countries lacking data per Table 5 should be eliminated from consideration, while the leading markets are Latvia, Hungary, Lithuania, Estonia, Italy, Bulgaria, Slovenia, Iceland, Romania and Spain.

6.2.3 Investment freedom index

The following index evaluates various investment restrictions, including bureaucracy, land ownership, foreign exchange control, capital control, security issues, absence of investment infrastructure, and expropriation of investments without fair compensation (Global Economy 2022). A higher value on the

index indicates greater investment freedom which is relevant for a DC because it has a direct impact on the ability to secure funding, sustain growth and technological advancement.

TABLE 6. Ranking of European countries by Investment freedom index in 2022 (Adapted fromGlobal Economy 2022)

	Investment Freedom (from most to least)												
1	Luxembourg	95	1	18 Poland		80		35	Slovenia	70			
2	Denmark	90	1	9	UK	80		36	Turkey	70			
3	Estonia	90	2	0	Armenia	75		37	Kazakhstan	50			
4	Ireland	90	2	1	Croatia	75		38	Bosnia & Herz.	65			
5	Netherlands	90	2	2	Cyprus	75		39	North Macedonia	65			
6	Belgium	85	2	3	France	75		40	Bulgaria	60			
7	Finland	85	2	4	Montenegro	75		41	Greece	55			
8	Latvia	85	2	5	Norway	75		42	Moldova	55			
9	Spain	85	2	6	Slovakia	75		43	Ukraine	35			
10	Sweden	85	2	7	Albania	70		44	Belarus	30			
11	Switzerland	85	2	8	Azerbaijan	70		45	Russia	30			
12	Austria	80	2	9	Czechia	70		46	Andorra	no data			
13	Georgia	80	2	0	Lithuania	70		47	Kosovo	no data			
14	Germany	80	3	1	Malta	70		48	Liechtenstein	no data			
15	Hungary	80	-	2	Portugal	70		49	Monaco	no data			
16	Iceland	80	-	3	Romania	70		50	San Marino	no data			
17	Italy	80	1	4	Serbia	70							

Based on this ranking it is concluded that Bosnia and Herzegovina, North Macedonia, Bulgaria, Greece, Moldova, Ukraine, Belarus, Russia, Andorra, Kosovo, Liechtenstein, Monaco and San Marino should be eliminated from consideration, while the leading markets are Luxembourg, Denmark, Estonia, Ireland, Netherlands, Belgium, Finland, Latvia, Spain, Sweden and Switzerland.

6.2.4 Fiscal freedom index

The following index measures the government imposed taxes, considering factors such as the top marginal tax rate on individual income, the top marginal tax rate on corporate income, and the total tax burden as a percentage of GDP (Global Economy 2022). A higher percentage indicates greater fiscal freedom in the corresponding country which is relevant for a DC as it impacts costs, profitability, and the ability to retain both foreign investments and skilled workforce.

TABLE 7. Ranking of European countries by fiscal freedom index in 2022 (Adapted from GlobalEconomy 2022)

]	Fiscal	Freedom (from mos	t to le	east)		
1	Azerbaijan	99	18	Lithuania	87	35	Greece	68
2	Luxembourg	99	19	Moldova	87	36	Georgia	67
3	Russia	99	20	Malta	86	37	Turkey	59
4	Norway	97	21	Serbia	86	38	Hungary	58
5	Bosnia & Herz.	96	22	Finland	83	39	Belgium	50
6	Bulgaria	96	23	Poland	78	40	Italy	49
7	Denmark	96	24	Slovenia	77	41	Romania	43
8	Sweden	96	25	Iceland	76	42	France	39
9	Switzerland	96	26	Armenia	76	43	Spain	30
10	Belarus	95	27	Slovakia	76	44	UK	23
11	Kazakhstan	94	28	Croatia	75	45	Montenegro	12
12	Netherlands	94	29	Ukraine	74	46	Andorra	no data
13	Czechia	93	30	Portugal	73	47	Kosovo	no data
14	Estonia	93	31	Austria	72	48	Liechtenstein	no data
15	Latvia	91	32	Albania	71	49	Monaco	no data
16	Germany	90	33	Cyprus	71	50	San Marino	no data
17	Ireland	89	34	North Macedonia	70			

Based on this ranking it is concluded that North Macedonia, Greece, Georgia, Turkey, Hungary, Belgium, Italy, Romania, France, Spain, UK, Montenegro, Andorra, Kosovo, Lichtenstein, Monaco and San Marino should be eliminated from consideration, while the leading markets are Azerbaijan, Luxembourg, Russia, Norway, Bosnia, Bulgaria, Denmark, Sweden, Switzerland and Belarus.

6.2.5 Legal preliminary screening conclusion

Based on the legal preliminary screening it can be concluded that the countries with either the least amount of available data or the highest legal potential risk to the development of a DC industry are: Andorra, Kosovo, Lichtenstein, Monaco, North Macedonia, San Marino, Belarus, Bosnia and Herzegovina, Moldova, Russia and Turkey, while the countries that offer the highest degree of legal stability are Spain and Switzerland.

6.3 Ecological preliminary screening

The preliminary screening will continue by evaluating indexes related to the ecological sustainability and impact of the European countries. The objective is to identify environmentally conscious and sustainable markets for potential future expansion of the DC industry.

6.3.1 Share of renewable energy

The share of renewable energy indicates the proportion of energy generated from renewable sources within a country and it is measured as a percentage of the total electricity generation. Renewable energy sources include solar, wind, hydro, geothermal, and bio-energy (International Renewable Energy Agency 2022). Analysing this percentage is relevant when considering building a DC in a certain country because it can affect the overall energy costs, the energy reliability and the environmental impact of the DC.

DCs are substantial consumers of energy, and the cost and reliability of energy play an important role in influencing overall operating costs and profitability. A high proportion of renewable energy in a country's energy mix indicates potential benefits such as lower energy costs, a more reliable energy supply, and a reduced environmental impact. This not only increases the feasibility of a country for companies investing in and building DCs but also contributes to the building's overall profitability.

Contrarily, a low share of renewable energy in the country's energy mix indicates higher energy costs, a less stable energy supply, and an increased environmental impact. This makes it less appealing for companies to invest in the country and construct DCs. Furthermore, there is a growing emphasis on sustainability in DC operations, with companies actively seeking ways to minimize their carbon foot-print. Using renewable energy is a key strategy in achieving this objective, as it effectively reduces the carbon emissions associated with DC operations

TABLE 8. Ranking of European countries by share of renewable energy in 2020 (Adapted from Balkan Green Energy News 2022; Cahill 2022; Deutsche Energie Agentur GmbH 2021, 1; Eurostat 2020; IEA 2020; IEA 2021; IEA 2021, 7; International Renewable Energy Agency 2022, 1; International Trade Administration 2022; Macrotrends 2023; UNECE REUptake 2021, 1; The Ministry of Energy of The Republic of Azerbaijan 2022; Ritchie & Roser 2021; World Data 2020)

	Share of Energy from Renwable Sources (from highest to lowest)											
1	Iceland	83.70%	18	Kosovo	25.69%	35	Ireland	16.20%				
2	Norway	77.40%	19	Slovenia	25.00%	36	Poland	16.10%				
3	Switzerland	62.20%	20	United Kingdom	24.84%	37	Andorra	15%				
4	Sweden	60.01%	21	Romania	24.50%	38	Netherlands	14.00%				
5	Albania	45%	22	Liechtenstein	24.21%	39	Hungary	13.90%				
6	Finland	43.80%	23	Armenia	24.00%	40	Belgium	13.00%				
7	Latvia	42.10%	24	Bulgaria	23.30%	41	Ukraine	12.40%				
8	Montenegro	39.00%	25	Greece	21.70%	42	Luxembourg	11.70%				
9	Bosnia and Herzegovina	37.00%	26	Spain	21.20%	43	Malta	10.70%				
10	Austria	36.50%	27	Georgia	20.50%	44	Russian Federation	6.62%				
11	Portugal	34.00%	28	Italy	20.40%	45	Belarus	6.00%				
12	Denmark	31.60%	29	Germany	19.30%	46	Moldova	6.00%				
13	North Macedonia	31.40%	30	France	19.10%	47	Kazakhstan	0.60%				
14	Croatia	31.00%	31	Azerbaijan	17.30%	48	San Marino	0%				
15	Estonia	30.20%	32	Czechia	17.30%	49	Turkiye	0%				
16	Lithuania	26.80%	33	Slovak Republic	17.30%	50	Monaco	no data				
17	Serbia	26.30%	34	Cyprus	16.90%							

Based on this ranking it is concluded that Belgium, Ukraine, Luxembourg, Malta, Russia, Belarus, Moldova, Kazakhstan, San Marino, Turkey and Monaco should be eliminated from consideration, while the leading markets are Iceland, Norway, Switzerland, Sweden, Albania, Finland, Latvia, Montenegro, Bosnia and Herzegovina and Austria.

6.3.2 Emissions index

The emissions index is calculated based on the weighting of carbon and methane emissions and it is ranging from 0 to 100, with 0 indicating the highest risk and 100 the lowest (World Economics 2020). Analysing this index during a DC location selection process is important because it provides insights into the environmental impact of the energy utilized by the potential DC. The emissions index quanti-

fies the volume of GHG produced by a country, region, or sector, typically measured in CO2 equivalent (World Economics 2020). High emission values can mean a greater environmental impact caused by the DC's energy usage, making it less appealing for companies trying to reduce their carbon footprint. A high emissions index also implies that the country's electricity likely originates from sources emitting substantial GHG, such as coal, oil, or natural gas (World Economics 2020).

Operating a DC in a country with a high index value would result in a larger carbon footprint, contributing to climate change. A low index value indicates that the country generates a significant portion of electricity from renewable sources such as wind or solar power, resulting in a lower environmental impact. Considering these values is crucial when constructing a DC, as it profoundly influences its overall sustainability (Subchapter 532.1.). Moreover, many countries are starting to implement carbon taxation and regulations to minimize emissions and climate change. A high emission index in Table 9 can signify higher costs associated with these measures, potentially impacting the profitability of a DC.

TABLE 9. Ranking of European countries by emissions index in 2020 (Adapted from World Economics 2020)

	Environmental Susteinability Index (from highest to lowest)											
1	Iceland	100.00%		18	Finland	99.60%	35	Spain	97.20%			
2	Malta	100.00%		19	Norway	99.60%	36	Kazakhstan	96.80%			
3	Albania	99.90%		20	Sweden	99.60%	37	Italy	96.60%			
4	Armenia	99.90%		21	Switzerland	99.60%	38	Poland	96.50%			
5	Cyprus	99.90%		22	Bulgaria	99.50%	39	Ukraine	96.40%			
6	Estonia	99.90%		23	Hungary	99.50%	40	France	96.10%			
7	Latvia	99.90%		24	Austria	99.40%	41	Turkiye	96.10%			
8	Luxembourg	99.90%		25	Greece	99.30%	42	United Kingdom	96.10%			
9	North Macedonia	99.90%		26	Portugal	99.30%	43	Germany	94.40%			
10	Slovenia	99.90%		27	Belgium	99.20%	44	Russian Federation	57.30%			
11	Croatia	99.80%		28	Serbia	99.20%	45	Andorra	no data			
12	Lithuania	99.80%		29	Ireland	99.10%	46	Kosovo	no data			
13	Moldova	99.80%		30	Belarus	99.00%	47	Liechtenstein	no data			
14	Bosnia and Herzegovina	99.70%		31	Czechia	99.00%	48	Monaco	no data			
15	Georgia	99.70%		32	Romania	98.70%	49	Montenegro	no data			
16	Slovak Republic	99.70%		33	Netherlands	98.50%	50	San Marino	no data			
17	Denmark	99.60%		34	Azerbaijan	98.10%						

Based on this ranking it is concluded that Spain, Kazakhstan, Italy, Poland, Ukraine, France, Turkey, UK, Germany, Russia, Andorra, Kosovo, Lichtenstein, Monaco, Montenegro and San Marino should

be eliminated from consideration, while the leading markets are Iceland, Malta, Albania, Armenia, Cyprus, Estonia, Latvia, Luxembourg, North Macedonia and Slovenia.

6.3.3 Ecological preliminary screening conclusion

Based on the ecological preliminary screening it can be concluded that the countries with either the least amount of available data or the highest ecological risk to the development of a DC industry are: Belgium, followed by Kazakhstan, Monaco, Russia, San Marino, Turkey and Ukraine, while the countries that offer the highest degree of ecological stability are: Albania, Iceland and Latvia.

6.4 Technological preliminary screening

The preliminary screening will continue by assessing technological indicators related to the digital infrastructure and innovation readiness of the European countries. By evaluating these indexes, the objective is to identify markets that offer high quality technological environments, promoting a feasible location for the efficient and sustainable future expansion of the DC industry.

6.4.1 Internet speed

Internet speed is a relevant factor for the comparative analysis of European countries, particularly in the selection of DC locations because it significantly influences the performance and reliability of the DC. A country with high internet speed can facilitate faster data transfer and enhance connectivity for users accessing the data, ultimately improving the user experience and overall operational efficiency.

In today's highly interconnected and fast paced digital environment, building a DC in a country with high internet speed offers a competitive advantage thanks to the ability to achieve faster data transfer and processing, which is vital for most businesses. A DC in a country with low internet speed may lead to delays and disruptions in data transfer, negatively impacting the performance and reliability of the DC.

		Inte	ernet	Speed (from highest m	egabits/sec	cor	nd t	o lowest)	
1	Monaco	261.82	18	Germany	136.66		35	Croatia	69.09
2	Romania	232.17	19	Latvia	135.22		36	Kazakhstan	63.28
3	Switzerland	229.96	20	Finland	135.05		37	Montenegro	60.03
4	Denmark	227.91	21	Belgium	125.12		38	Cyprus	55.11
5	France	214.04	22	Ireland	122.55		39	Bosnia and Herzegovina	48.51
6	Hungary	210.55	23	Slovenia	115.61	4	40	Albania	45.55
7	Spain	201.47	24	Slovak Republic	115.11	4	41	Turkiye	43.48
8	Sweden	174.9	25	Italy	104.09	4	42	North Macedonia	43.24
9	Luxembourg	172.05	26	United Kingdom	102.24	4	43	Greece	39.08
10	Andorra	171.44	27	Russian Federation	95.96	4	44	Armenia	38.95
11	Norway	168.25	28	Austria	94.31	4	45	Iceland	32.13
12	Netherlands	165	29	Estonia	88.49	4	46	Georgia	28.61
13	Portugal	162.55	30	Serbia	86.77	4	47	Azerbaijan	25.1
14	Poland	156.33	31	Bulgaria	86.75	4	48	Kosovo	no data
15	Moldova	154.56	32	Ukraine	76.82	4	49	Liechtenstein	no data
16	Lithuania	147.05	33	Czechia	74.8	:	50	San Marino	no data
17	Malta	144.06	34	Belarus	69.41				

TABLE 10. Ranking of European countries by internet speed in 2023 (Adapted from World Popula-tion Review 2023)

Based on this ranking it is concluded that Cyprus, Bosnia and Herzegovina, Albania, Turkey, North Macedonia, Greece, Armenia, Iceland, Georgia, Azerbaijan, Kosovo, Lichtenstein and San Marino should be eliminated from consideration, while the leading markets are Monaco, Romania, Switzer-land, Denmark, France, Hungary, Spain, Sweden, Luxembourg and Andorra.

6.4.2 Internet bandwidth

The internet bandwidth indicator represented in Table 11 is the sum of the capacity of all internet exchanges offering international bandwidth and is measured in kilobits per second (kb/s) (Global Economy 2016). Bandwidth and internet speed are often mistaken but they are different concepts, as bandwidth refers to the volume of information that can be sent over a connection in a given time, measured in megabits per second (Mbps) (Verizon 2022). On the other hand, internet speed measures how fast information can be received or downloaded (Parrish, Holslin & Armstrong 2024). Internet bandwidth is an important factor to consider during the DC location selection process because it affects performance, scalability, reliability and user experience for the DC' services and operations.

Internet Bandwidth (from best to worst)											
Luxembourg	6887.71	18	Lithuania	125.45		35	Bosnia & Herz.	43			
Malta	1178.76	19	Slovenia	121.14		36	Turkey	42.91			
Sweden	527.45	20	Romania	117.32		37	North Macedonia	41.81			
Iceland	519.87	21	Czechia	116.81		38	Ukraine	40.7			
UK	429.83	22	Serbia	112.37		39	Hungary	37.03			
Switzerland	352.24	23	Spain	111.55		40	Azerbaijan	32.22			
Denmark	341.71	24	Greece	99.51		41	Albania	32.08			
Netherlands	281.06	25	Latvia	93.68		42	Russia	29.86			
Belgium	263.91	26	Italy	92.5		43	Estonia	28.67			
France	221.66	27	Poland	90.36		44	Slovakia	11.46			
Portugal	218.88	28	Austria	79.64		45	Andorra	no data			
Finland	218.74	29	Montenegro	77.02		46	Belarus	no data			
Norway	203.94	30	Cyprus	75.05		47	Kosovo	no data			
Ireland	160.97	31	Georgia	70.97		48	Liechtenstein	no data			
Moldova	152.36	32	Croatia	58.03		49	Monaco	no data			
Germany	145.99	33	Kazakhstan	51.49		50	San Marino	no data			
Bulgaria	135.11	34	Armenia	44.53							
	Luxembourg Malta Sweden Iceland UK Switzerland Denmark Denmark Belgium France Portugal France Finland Norway Ireland Iceland Moldova Bulgaria	Luxembourg 6887.71 Malta 1178.76 Sweden 527.45 Iceland 519.87 UK 429.83 Switzerland 352.24 Denmark 341.71 Netherlands 281.06 Belgium 263.91 France 221.66 Portugal 218.88 Finland 218.74 Norway 203.94 Ireland 160.97 Moldova 152.36 Germany 145.99 Bulgaria 135.11	Luxembourg 6887.71 18 Malta 1178.76 19 Sweden 527.45 20 Iceland 519.87 21 UK 429.83 22 Switzerland 352.24 23 Denmark 341.71 24 Netherlands 281.06 25 Belgium 263.91 26 France 221.66 27 Portugal 218.88 28 Finland 218.74 29 Norway 203.94 30 Ireland 160.97 31 Moldova 152.36 32 Germany 145.99 33	Luxembourg6887.7118LithuaniaMalta1178.7619SloveniaSweden527.4520RomaniaIceland519.8721CzechiaUK429.8322SerbiaSwitzerland352.2423SpainDenmark341.7124GreeceBelgium263.9126ItalyFrance221.6627PolandPortugal218.8828AustriaFinland218.7429MontenegroNorway203.9430CyprusIreland160.9731GeorgiaMoldova152.3632CroatiaBulgaria135.1134Armenia	Luxembourg 6887.71 18 Lithuania 125.45 Malta 1178.76 19 Slovenia 121.14 Sweden 527.45 20 Romania 117.32 Iceland 519.87 21 Czechia 116.81 UK 429.83 22 Serbia 112.37 Switzerland 352.24 23 Spain 111.55 Denmark 341.71 24 Greece 99.51 Netherlands 281.06 25 Latvia 93.68 Belgium 263.91 26 Italy 92.5 France 221.66 27 Poland 90.36 Portugal 218.74 29 Montenegro 77.02 Norway 203.94 30 Cyprus 75.05 Ireland 160.97 31 Georgia 58.03 Germany 145.99 33 Kazakhstan 51.49 Bulgaria 135.11 34 Armenia 44.53	Luxembourg 6887.71 18 Lithuania 125.45 Malta 1178.76 19 Slovenia 121.14 Sweden 527.45 20 Romania 117.32 Iceland 519.87 21 Czechia 116.81 UK 429.83 22 Serbia 112.37 Switzerland 352.24 23 Spain 111.55 Denmark 341.71 24 Greece 99.51 Netherlands 281.06 25 Latvia 93.68 Belgium 263.91 26 Italy 92.5 France 221.66 27 Poland 90.36 Portugal 218.74 29 Montenegro 77.02 Norway 203.94 30 Cyprus 75.05 Ireland 160.97 31 Georgia 70.97 Moldova 152.36 32 Croatia 58.03 Germany 145.99 33 Kazakhstan 51.49 <	Luxembourg 6887.71 18 Lithuania 125.45 35 Malta 1178.76 19 Slovenia 121.14 36 Sweden 527.45 20 Romania 117.32 37 Iceland 519.87 21 Czechia 116.81 38 UK 429.83 22 Serbia 112.37 39 Switzerland 352.24 23 Spain 111.55 40 Denmark 341.71 24 Greece 99.51 41 Netherlands 281.06 25 Latvia 93.68 42 France 221.66 27 Poland 90.36 44 Portugal 218.88 28 Austria 79.64 45 Finland 216.097 31 Georgia 70.97 48 Moldova 152.36 32 Croatia 58.03 49 Germany 145.99 33 Kazakhstan 51.49 50 <td>Internet Bandwidth (from best to worst)Luxembourg6887.7118Lithuania125.4535Bosnia & Herz.Malta1178.7619Slovenia121.1436TurkeySweden527.4520Romania117.3237North MacedoniaIceland519.8721Czechia116.8138UkraineUK429.8322Serbia112.3739HungarySwitzerland352.2423Spain111.5540AzerbaijanDenmark341.7124Greece99.5141AlbaniaNetherlands281.0625Latvia93.6842RussiaBelgium263.9126Italy92.543EstoniaFrance221.6627Poland90.3644SlovakiaPortugal218.8828Austria79.6445AndorraFinland218.7429Montenegro77.0246BelarusNorway203.9430Cyprus75.0547KosovoIreland160.9731Georgia70.9748LiechtensteinMoldova152.3632Croatia58.0349MonacoGermany145.9933Kazakhstan51.4950San Marino</td>	Internet Bandwidth (from best to worst)Luxembourg6887.7118Lithuania125.4535Bosnia & Herz.Malta1178.7619Slovenia121.1436TurkeySweden527.4520Romania117.3237North MacedoniaIceland519.8721Czechia116.8138UkraineUK429.8322Serbia112.3739HungarySwitzerland352.2423Spain111.5540AzerbaijanDenmark341.7124Greece99.5141AlbaniaNetherlands281.0625Latvia93.6842RussiaBelgium263.9126Italy92.543EstoniaFrance221.6627Poland90.3644SlovakiaPortugal218.8828Austria79.6445AndorraFinland218.7429Montenegro77.0246BelarusNorway203.9430Cyprus75.0547KosovoIreland160.9731Georgia70.9748LiechtensteinMoldova152.3632Croatia58.0349MonacoGermany145.9933Kazakhstan51.4950San Marino			

TABLE 11. Ranking of European countries by internet bandwidth in 2016 (Adapted from GlobalEconomy 2016)

Based on this ranking it is concluded that Bosnia and Herzegovina, Turkey, North Macedonia, Ukraine, Hungary, Azerbaijan, Albania, Russia, Estonia, Slovakia, Andorra, Belarus, Kosovo, Lichtenstein, Monaco and San Marino should be eliminated from consideration, while the leading markets are Luxembourg, Malta, Sweden, Iceland, UK, Switzerland, Denmark, Netherlands, Belgium and France.

6.4.3 Electricity production capacity

Another relevant technological index is the country's electricity production capacity because a high value can indicate a local good quality, reliable electrical source which is one of the most important factors to consider during the DC location selection process (Subchapter 5.3.1.). Table 12 ranks the European countries based on their capacity, measured in million kilowatts.

TABLE 12. Ranking of European countries by electricity production capacity in 2020 (Adapted fromGlobal Economy 2020)

		Elec	tric	city]	Production Capacity (from h	nighest to	o lo	wes	t)	
1	Russia	276.9		18	Romania	20.93		35	Latvia	3.07
2	Germany	242.24		19	Greece	20.61		36	Estonia	3.02
3	France	138.39		20	Finland	19.79		37	Iceland	2.98
4	Italy	120.72		21	Denmark	17.15		38	Albania	2.33
5	United Kingdom	112.34		22	Bulgaria	11.32		39	North Macedonia	1.86
6	Spain	111.18		23	Ireland	11.28		40	Cyprus	1.83
7	Turkiye	92.04		24	Hungary	10.32		41	Luxembourg	1.82
8	Ukraine	55.19		25	Belarus	10.24		42	Montenegro	1
9	Poland	44.42		26	Serbia	7.9		43	Malta	0.75
10	Sweden	42.58		27	Slovak Republic	7.85		44	Moldova	0.58
11	Netherlands	38.25		28	Azerbaijan	7.66		45	Andorra	no data
12	Norway	36.86		29	Croatia	4.85		46	Kazakhstan	no data
13	Austria	27.82		30	Bosnia and Herzegovina	4.76		47	Kosovo	no data
14	Belgium	25.16		31	Georgia	4.35		48	Liechtenstein	no data
15	Czechia	22.5		32	Slovenia	3.87		49	Monaco	no data
16	Portugal	22.29		33	Armenia	3.59		50	San Marino	no data
17	Switzerland	21.88		34	Lithuania	3.47				

Based on this ranking it is concluded that Latvia, Estonia, Iceland, Albania, North Macedonia, Cyprus, Luxembourg, Montenegro, Malta, Moldova, Andorra, Kazakhstan, Kosovo, Lichtenstein, Monaco and San Marino should be eliminated from consideration, while the leading markets are Russia, Germany, France, Italy, UK, Spain, Turkey, Ukraine, Poland and Sweden.

6.4.4 Renewable power capacity

The following index measures the European countries' overall capacity to generate electricity from renewable sources and is measured in million kilowatts (Global Economy 2020). Access to renewable power allows DCs to minimize their environmental impact, reduce costs, increase reliability, meet regulatory requirements and demonstrate social responsibility.

	Renwable Power Capacity (from most to least)										
1	Germany	131.4	18	Denmark	9.68	35	Azerbaijan	1.28			
2	Spain	56.78	19	Finland	8.73	36	Lithuania	0.95			
3	Italy	52.47	20	Switzerland	7.85	37	North Macedonia	0.82			
4	Russia	52.35	21	Ireland	4.72	38	Montenegro	0.78			
5	France	51.61	22	Czechia	4.35	39	Estonia	0.73			
6	Turkey	49.56	23	Bulgaria	3.59	40	Luxembourg	0.45			
7	UK	48.05	24	Georgia	3.47	41	Belarus	0.43			
8	Norway	35.85	25	Serbia	2.93	42	Cyprus	0.37			
9	Sweden	33.54	26	Hungary	2.89	43	Malta	0.19			
10	Netherlands	18.11	27	Croatia	2.88	44	Moldova	0.12			
11	Austria	16.4	28	Iceland	2.85	45	Andorra	no data			
12	Ukraine	13.61	29	Albania	2.41	46	Kazakhstan	no data			
13	Poland	11.85	30	Slovakia	2.38	47	Kosovo	no data			
14	Belgium	11.47	31	Bosnia & Herz.	2.22	48	Liechtenstein	no data			
15	Portugal	11.38	32	Latvia	1.82	49	Monaco	no data			
16	Romania	10.77	33	Slovenia	1.68	50	San Marino	no data			
17	Greece	10.2	34	Armenia	1.39						

TABLE 13. Ranking of European countries by renewable power capacity in 2020 (Adapted fromGlobal Economy 2020)

Based on this ranking it is concluded that Lithuania, North Macedonia, Montenegro, Estonia, Luxembourg, Belarus, Cyprus, Malta, Moldova, Andorra, Kazakhstan, Kosovo, Lichtenstein, Monaco and San Marino should be eliminated from consideration, while the leading markets are Germany, Spain, Italy, Russia, France, Turkey, UK, Norway, Sweden and Netherlands.

6.4.5 Road quality index

The following index assesses the quality of the roads in the European countries based on data from the WEF Executive Opinion Survey, including the opinions of over 14,000 business leaders in 144 countries. The score for road quality is based on the respondents' 1 to 7 ranking of the roads in their country of operation, with 1 meaning underdeveloped and 7 meaning efficient by international standards. The individual responses were compiled to produce a country score. (Global Economy 2019.)

Evaluating the access to quality road infrastructure is relevant because it supports the operational efficiency, it facilitates transportation of equipment and personnel and it supports emergency responses, making it an important consideration in site selection and infrastructure planning.

TABLE 14. Ranking of European countries by roads quality in 2019 (Adapted from Global	Economy
2019)	

	Road Infrastructure Quality (from best to worst)										
1	Netherlands	6.4		18	Lithuania	4.8		35	Latvia	3.6	
2	Switzerland	6.3		19	Estonia	4.7		36	Russia	3.5	
3	Austria	6		20	Greece	4.6		37	Serbia	3.5	
4	Portugal	6		21	Norway	4.5		38	Bulgaria	3.4	
5	Spain	5.7		22	Belgium	4.4		39	North Macedonia	3.4	
6	Croatia	5.6		23	Ireland	4.4		40	Malta	3.3	
7	Denmark	5.6		24	Italy	4.4		41	Romania	3	
8	Luxembourg	5.5		25	Poland	4.3		42	Ukraine	3	
9	France	5.4		26	Iceland	4.1		43	Bosnia & Herz.	2.8	
10	Finland	5.3		27	Hungary	4		44	Moldova	2.6	
11	Germany	5.3		28	Slovakia	4		45	Andorra	no data	
12	Sweden	5.3		29	Albania	3.9		46	Belarus	no data	
13	Azerbaijan	5.2		30	Czechia	3.9		47	Kosovo	no data	
14	Cyprus	5.1		31	Montenegro	3.9		48	Liechtenstein	no data	
15	Turkey	5		32	Georgia	3.8		49	Monaco	no data	
16	Slovenia	4.9		33	Armenia	3.6		50	San Marino	no data	
17	UK	4.9		34	Kazakhstan	3.6					

Based on this ranking it is concluded that Russia, Serbia, Bulgaria, North Macedonia, Malta, Romania, Ukraine, Bosnia and Herzegovina, Moldova, Andorra, Belarus, Kosovo, Lichtenstein, Monaco and San Marino should be eliminated from consideration, while the leading markets are Netherlands, Switzerland, Austria, Portugal, Spain, Croatia, Denmark, Luxembourg, France, Finland, Germany and Sweden.

6.4.6 Railroad quality index

The following index assesses the quality of the railroads in the European countries based on data from the WEF Executive Opinion Survey, including the opinions of over 14,000 business leaders in 144 countries. The score for railroad quality is based on the respondents' 1 to 7 ranking of the railroads in their country of operation, with 1 meaning underdeveloped and 7 meaning efficient by international standards. The individual responses were compiled to produce a country score. (Global Economy 2019.)

Evaluating the access to quality railroad infrastructure is relevant because it supports the operational efficiency and global connectivity of DCs, making it an important consideration in site selection and infrastructure planning.

TABLE 15. Ranking of European countries by railroads quality in 2019 (Adapted from Global Economy 2019)

	R	ailroa	d Infr	astructure Qua	lity (fro	om b	est to worst)	
1	Switzerland	6.4	18	Ukraine	4.2		35	Romania	2.8
2	Netherlands	5.7	19	Belgium	4.1		36	Serbia	2.6
3	Finland	5.5	20	Italy	4.1		37	Croatia	2.4
4	Spain	5.4	21	Ireland	4		38	Bosnia & Herz.	2.2
5	Austria	5.3	22	Slovakia	4		39	North Macedonia	2.1
6	France	5	23	Sweden	4		40	Albania	1.2
7	Luxembourg	5	24	Georgia	3.9		41	Andorra	no data
8	Germany	4.9	25	Poland	3.9		42	Azerbaijan	no data
9	Russia	4.9	26	Hungary	3.8		43	Belarus	no data
10	Latvia	4.6	27	Turkey	3.5		44	Cyprus	no data
11	Lithuania	4.6	28	Armenia	3.1		45	Iceland	no data
12	Czechia	4.5	29	Bulgaria	3.1		46	Kosovo	no data
13	Denmark	4.5	30	Estonia	3.1		47	Liechtenstein	no data
14	Norway	4.5	31	Montenegro	3.1		48	Malta	no data
15	UK	4.3	32	Slovenia	3.1		49	Monaco	no data
16	Kazakhstan	4.2	33	Greece	3		50	San Marino	no data
17	Portugal	4.2	34	Moldova	3				

Based on this ranking it is concluded that Montenegro, Slovenia, Greece, Moldova, Romania, Serbia, Croatia, Bosnia and Herzegovina, North Macedonia, Albania, Andorra, Azerbaijan, Belarus, Cyprus, Iceland, Kosovo, Lichtenstein, Malta, Monaco and San Marino should be eliminated from consideration, while the leading markets are Switzerland, Netherlands, Finland, Spain, Austria, France, Luxembourg, Germany, Russia, Latvia and Lithuania.

6.4.7 Port infrastructure quality index

The following index assesses the quality of the ports in the European countries based on data from the WEF Executive Opinion Survey, including the opinions of over 14,000 business leaders in 144 countries. The score for port infrastructure quality is based on the respondents' 1 to 7 ranking of the port facilities and inland waterways in their country of operation, with 1 meaning underdeveloped and 7 meaning efficient by international standards. For landlocked countries, the respondents were asked to rate the access to port facilities and inland waterways on a scale from 1 meaning impossible, to 7 meaning easy. The individual responses were compiled to produce a country score. (Global Economy 2019.)

Evaluating the access to quality port infrastructure is relevant because it supports the operational efficiency and global connectivity of DCs, making it an important consideration in site selection and infrastructure planning.

]	Port Infra	stru	cture Quality (f	rom be	est to	worst)	
1	Finland	6.4	16	Latvia	4.9	31	Montenegro	4.2
2	Netherlands	6.4	17	Portugal	4.9	32	Romania	3.9
3	Denmark	5.8	18	Greece	4.8	33	Ukraine	3.9
4	Belgium	5.6	19	Lithuania	4.8	34	Georgia	3.8
5	Estonia	5.6	20	Croatia	4.7	35	Austria	3.7
6	Iceland	5.4	21	Italy	4.7	36	Kazakhstan	3.3
7	Spain	5.4	22	Russia	4.7	37	Czechia	3.2
8	Sweden	5.3	23	Slovenia	4.7	38	Hungary	3.2
9	France	5.2	24	Turkey	4.7	39	Serbia	3.1
10	Germany	5.2	25	Poland	4.5	40	Slovakia	3.1
11	UK	5.2	26	Switzerland	4.5	41	Armenia	2.4
12	Azerbaijan	5.1	27	Luxembourg	4.4	42	North Macedonia	2.4
13	Malta	5.1	28	Albania	4.3	43	Moldova	2.3
14	Norway	5.1	29	Bulgaria	4.3	44	Bosnia & Herz.	2.1
15	Ireland	5	30	Cyprus	4.3			

TABLE 16. Ranking of European countries by port infrastructure quality in 2019 (Adapted fromGlobal Economy 2019)

Based on this ranking it is concluded that Georgia, Austria, Kazakhstan, Czech, Hungary, Serbia, Slovakia, Armenia, North Macedonia, Moldova and Bosnia and Herzegovina should be eliminated from consideration, while the leading markets are Finland, Netherlands, Denmark, Belgium, Estonia, Iceland, Spain, Sweden, France, Germany and UK.

6.4.8 Air infrastructure quality index

The following index assesses the quality of the airports in the European countries based on data from the WEF Executive Opinion Survey, including the opinions of over 14,000 business leaders in 144 countries. The score for air transport infrastructure quality is based on the respondents' 1 to 7 ranking of the passenger air transport in their country of operation, with 1 meaning underdeveloped and 7 meaning efficient by international standards. The individual responses were compiled to produce a country score. (Global Economy 2019.)

Evaluating the access to quality air infrastructure is relevant because it supports the operational efficiency and global connectivity of DCs, making it an important consideration in site selection and infrastructure planning.

TABLE 17. Ranking of European countries by air infrastructure quality in 2019 (Adapted from GlobalEconomy 2019)

		Air I	nfra	structure Qual	ity (f	ro	m b	est to worst)	
1	Netherlands	6.4	18	Turkey	5.4		35	Bulgaria	4.5
2	Finland	6.3	19	UK	5.3		36	Georgia	4.4
3	Switzerland	6.1	20	Austria	5.2		37	Moldova	4.4
4	Azerbaijan	5.8	21	Cyprus	5.1		38	Kazakhstan	4.3
5	Denmark	5.8	22	Czechia	5		39	North Macedonia	4.3
6	Latvia	5.7	23	Portugal	5		40	Serbia	4.3
7	Sweden	5.7	24	Russia	5		41	Albania	4.1
8	Belgium	5.6	25	Italy	4.9		42	Ukraine	4
9	Iceland	5.6	26	Lithuania	4.9		43	Slovakia	3.8
10	Luxembourg	5.6	27	Croatia	4.8		44	Bosnia & Herzegovina	3.5
11	Norway	5.6	28	Poland	4.8		45	Andorra	no data
12	Spain	5.6	29	Armenia	4.6		46	Belarus	no data
13	France	5.5	30	Estonia	4.6		47	Kosovo	no data
14	Germany	5.5	31	Hungary	4.6		48	Liechtenstein	no data
15	Ireland	5.5	32	Montenegro	4.6		49	Monaco	no data
16	Malta	5.5	33	Romania	4.6		50	San Marino	no data
17	Greece	5.4	34	Slovenia	4.6				

Based on this ranking it is concluded that Bulgaria, Georgia, Moldova, Kazakhstan, North Macedonia, Serbia, Albania, Ukraine, Slovakia, Bosnia and Herzegovina, Andorra, Belarus, Kosovo, Lichtenstein, Monaco and San Marino should be eliminated from consideration, while the leading markets are Netherlands, Finland, Switzerland, Azerbaijan, Denmark, Latvia, Sweden, Belgium, Iceland, Luxembourg, Norway and Spain.

6.4.9 Technological preliminary screening conclusion

Based on the technological preliminary screening it can be concluded that the countries with either the least amount of available data or lowest level of technological development relevant to the DC industry are: North Macedonia, Kosovo, Lichtenstein, San Marino, followed by Andorra, Bosnia and Herzegovina, Moldova, Monaco, Albania and Belarus, while the countries that offer the highest degree of technological development are: France, Spain, Sweden, followed by Netherlands.

6.5 Socio-cultural preliminary screening

The preliminary screening will continue by assessing socio-cultural indicators to evaluate the social and cultural environment of the European countries. The objective is to identify markets that offer a supportive and inclusive socio-cultural environment for the successful establishment and operation of new DCs.

6.5.1 Rule of law

The first socio-cultural index to be investigated is the rule of law which is a statistic based on the average available data between 1996 and 2021, with '-2.5' as the weakest value and '2.5' as the strongest value. According to Global Economy (2021), this index shows the likelihood of a society to abide by rules and laws and it can give a good indication of the quality of contract enforcements, rights, courts, police and likelihood of crime.

A country with a strong rule of law should have a predictable legal framework that can be relied upon to protect property rights, enforce contracts and provide security for investments. This will make it more feasible for foreign companies to invest and operate in the country, as they will have a higher degree of confidence in the security of their investments. A strong rule of law also promotes stability by reducing the risk of arbitrary or discriminatory actions by government officials or other entities. On the other hand, a weak rule of law can lead to instability, unpredictability and insecurity in the business environment. This can create uncertainty and increase the risk of arbitrary or discriminatory actions by government officials, which can lead to delays or cancellations of potential projects, loss of property rights and increased business risks (Global Economy 2021). In the context of a DC, the rule of law plays a critical role in protecting the DC and the data stored in it. It ensures that the DC is protected by a legal framework, the rights and obligations of the parties are clearly defined and that disagreements are resolved fairly and in a timely manner.

TABLE 18. Ranking of European countries by rule of law in 2021 (Adapted from Global Economy2021)

			Ru	e of Law Index (fro	m high	est	to lo	owest)	
1	Finland	2.06	18	Czechia	1.13		35	Montenegro	-0.06
2	Norway	1.95	19	Portugal	1.13		36	North Macedonia	-0.08
3	Denmark	1.94	20	Lithuania	1.11		37	Serbia	-0.09
4	Switzerland	1.81	21	Slovenia	1.03		38	Armenia	-0.1
5	Austria	1.79	22	Latvia	0.98		39	Albania	-0.26
6	Luxembourg	1.79	23	Spain	0.88		40	Bosnia and Herzegovina	-0.28
7	Iceland	1.75	24	Malta	0.86		41	Moldova	-0.33
8	Liechtenstein	1.74	25	Slovak Republic	0.71		42	Turkiye	-0.42
9	Netherlands	1.74	26	o Cyprus	0.64		43	Kazakhstan	-0.49
10	Sweden	1.73	27	Hungary	0.53		44	Azerbaijan	-0.58
11	Andorra	1.72	28	Poland	0.44		45	Ukraine	-0.66
12	Germany	1.61	29	Romania	0.41		46	Russian Federation	-0.87
13	Ireland	1.53	30	Greece	0.35		47	Belarus	-1.1
14	Estonia	1.43	3	Croatia	0.3		48	Kosovo	no data
15	United Kingdom	1.43	32	Italy	0.27		49	Monaco	no data
16	Belgium	1.33	33	Georgia	0.17		50	San Marino	no data
17	France	1.29	34	Bulgaria	-0.04				

Based on this ranking it is concluded that Bulgaria, Montenegro, North Macedonia, Serbia, Armenia, Albania Bosnia and Herzegovina, Moldova, Turkey, Kazakhstan, Azerbaijan, Ukraine, Russia, Belarus, Kosovo, Monaco and San Marino should be eliminated from consideration, while the leading markets are Finland, Norway, Denmark, Switzerland, Austria, Luxembourg, Iceland, Lichtenstein, Netherlands and Sweden.

6.5.2 Globalization index

Another important socio-cultural evaluation is the social globalization index, shown in Table 19 as percentages of the average data available in 2020. Global Economy indicates that social globalization

is the process by which people, information and culture become more interconnected and interdependent. It can be broken down into three dimensions: personal contacts, information flows and cultural proximity. Analysing this indicator is relevant when considering building a DC in a certain country because it can affect the accessibility and connectivity of the location.

TABLE 19. Ranking of European countries by social globalization in 2020 (Adapted from GlobalEconomy 2020)

	Social Globalization Index (from highest to lowest)											
1	Luxembourg	90	18	Malta	85		35	Poland	77			
2	Monaco	90	19	Croatia	84		36	Serbia	77			
3	Switzerland	90	20	Czechia	84		37	Romania	76			
4	Liechtenstein	89	21	Estonia	84		38	Georgia	73			
5	Norway	89	22	Greece	84		39	Bosnia and Herzegovina	69			
6	United Kingdom	89	23	Iceland	84		40	Moldova	69			
7	Austria	88	24	Lithuania	84		41	Ukraine	69			
8	Germany	88	25	San Marino	84		42	Russia	68			
9	Ireland	88	26	Slovakia	84		43	North Macedonia	67			
10	Sweden	88	27	Portugal	83		44	Albania	66			
11	Andorra	87	28	Spain	83		45	Kazakhstan	66			
12	Denmark	87	29	Latvia	82		46	Armenia	65			
13	Netherlands	87	30	Hungary	80		47	Turkey	65			
14	Cyprus	86	31	Italy	80		48	Belarus	64			
15	Finland	86	32	Slovenia	80		49	Azerbaijan	57			
16	France	86	33	Bulgaria	77		50	Kosovo	no data			
17	Belgium	85	34	Montenegro	77							

Based on this ranking it is concluded that Bosnia and Herzegovina, Moldova, Ukraine, Russia, North Macedonia, Albania, Kazakhstan, Armenia, Turkey, Belarus, Azerbaijan and Kosovo should be eliminated from consideration, while the leading markets are Luxembourg, Monaco, Switzerland, Lichtenstein, Norway, UK, Austria, Germany, Ireland and Sweden.

6.5.3 Positive attitude towards immigrants index

In order to measure the local level of positive attitude towards immigrants, statistics from a report that gathered data on a 0 to 10 scale by asking locals "Do immigrants make the country a better place to
live in?" will be used (Ruelens & Nicaise 2022). This is relevant when considering building a DC in a certain country because it can affect the availability and diversity of the workforce. The attitude towards immigrants in a country can indicate the level of acceptance and inclusion of foreign workers in the local market. A positive attitude towards immigrants can result in a more diverse and inclusive workforce, which can provide a wider range of skill sets and perspectives, and can also improve the ability to attract and retain employees from different countries and backgrounds.

On the other hand, a negative attitude towards immigrants can make it more difficult for employers to retain foreign skilled workers, which can limit the skill sets and perspectives available to the company. Additionally, the local attitude towards immigrants can also affect the ability of employees to integrate into the local community, which can have an impact on their mental and emotional well being, and their ability to perform their job.

TABLE 20. Ranking of European countries by positive general attitude towards immigrants in 2022(Adapted from Goubin, Ruelens & Nicaise 2022, 16)

Attitude Towards Immigrants (from best to worst)														
1	Iceland	7,2	18	Lithuania	5	35	Bosnia and Herzegovina	no data						
2	Ireland	6,3	19	Montenegro	4,6	36	Georgia	no data						
3	Sweden	6,3	20	Serbia	4,5	37	Greece	no data						
4	Norway	5,9	21	Austria	4,4	38	Kazakhstan	no data						
5	Denmark	5,85	22	Cyprus	4,3	39	Kosovo	no data						
6	Portugal	5,8	23	Estonia	4,3	40	Liechtenstein	no data						
7	United Kingdom	5,8	24	Slovenia	4,3	41	Luxembourg	no data						
8	Netherlands	5,7	25	Italy	4,1	42	Malta	no data						
9	Finland	5,5	26	Bulgaria	3,9	43	Moldova	no data						
10	Poland	5,5	27	Hungary	3,9	44	Monaco	no data						
11	Spain	5,5	28	Slovak Republic	3,9	45	North Macedonia	no data						
12	Switzerland	5,5	29	Czechia	3,8	46	Romania	no data						
13	Germany	5,3	30	Albania	no data	47	Russian Federation	no data						
14	Latvia	5,3	31	Andorra	no data	48	San Marino	no data						
15	Belgium	5,2	32	Armenia	no data	49	Turkiye	no data						
16	Croatia	5	33	Azerbaijan	no data	50	Ukraine	no data						
17	France	5	34	Belarus	no data									

Based on this ranking it is concluded that Montenegro, Serbia, Austria, Cyprus, Estonia, Slovenia, Italy, Bulgaria, Hungary, Slovakia, Czech and all the countries with no available data should be eliminated from consideration, while the leading markets are Iceland, Ireland, Sweden, Norway, Denmark, Portugal, UK, Netherlands, Finland and Poland.

6.5.4 Socio-cultural preliminary screening conclusion

Based on the socio-cultural preliminary screening it can be concluded that the countries with the highest level of socio-cultural risks relevant to the DC industry are: Albania, Armenia, Azerbaijan, Belarus, Bosnia and Herzegovina, Kazakhstan, Kosovo, Moldova, North Macedonia, Russia, Turkey and Ukraine; while the countries with the lowest level of socio-cultural risks are: Norway and Sweden.

6.6 Economic preliminary screening

Lastly, the preliminary screening will be finalized by assessing economic indexes to evaluate the economic environment of all the European countries. The objective is to identify markets where the current state of their economic factors create a favourable environment for building local DCs, as well as identify the markets that are least feasible.

6.6.1 Labour cost index

The first economic index evaluated is the labour cost which measures the total compensation of employees, including salaries, employers' social security contributions and employment taxes (Global Economy 2022). Analysing local labour costs is relevant when considering building a DC in a certain country because it can affect the overall operating costs and profitability. High labour costs can make it more expensive to hire and retain employees, as well as lower regional competitiveness. On the other hand, low labour costs can make it more affordable to hire and retain employees, which can lower the overall operating costs of a DC, and make it more competitive in the region. TABLE 21. Ranking of European countries by labour cost in 2022 (Adapted from Global Economy2022)

			L	abor Cost (from lowest to hi	ghest)			
1	Finland	107.2	18	Croatia	128.1	35	North Macedonia	230.08
2	Italy	109.1	19	Greece	129.2	36	Moldova	247.62
3	Spain	112.4	20	Slovenia	130.4	37	Belarus	249.85
4	Denmark	112.7	21	Albania	134.05	38	Azerbaijan	270.3
5	France	113.7	22	Czechia	139.9	39	Turkiye	554.34
6	Belgium	115.2	23	Iceland	144.4	40	Ukraine	4541.22
7	Montenegro	115.49	24	Estonia	144.8	41	Andorra	no data
8	Germany	116.5	25	Latvia	148.7	42	Armenia	no data
9	Sweden	116.7	26	Slovak Republic	149.5	43	Kazakhstan	no data
10	Luxembourg	116.9	27	Poland	159.2	44	Kosovo	no data
11	Russian Federation	117.09	28	Hungary	170.2	45	Liechtenstein	no data
12	Malta	118	29	Lithuania	170.2	46	Monaco	no data
13	Netherlands	118.8	30	Bosnia and Herzegovina	177.3	47	Norway	no data
14	Portugal	121.6	31	Bulgaria	181.5	48	San Marino	no data
15	Ireland	121.8	32	Romania	191	49	Serbia	no data
16	Cyprus	123.3	33	United Kingdom	198.04	50	Switzerland	no data
17	Austria	124	34	Georgia	200.51			

Based on this ranking it can be concluded that Bulgaria, Romania, UK, Georgia, North Macedonia, Moldova, Belarus, Azerbaijan, Turkey, Ukraine and all the countries with no available data should be eliminated from consideration, while the leading markets are Finland, Italy, Spain, Denmark, France, Belgium, Montenegro, Germany, Sweden and Luxembourg.

6.6.2 Transport prices

The following index evaluates the cost of transportation, including road, rail, and air transport expenses (Global Economy 2017). By comparing values across countries, relative price levels can be determined which is relevant for a DC as transport prices have a high impact on the building's financial feasibility, including its operational efficiency and sustainability.

TABLE 22. Ranking of European countries by transport prices in 2017 (Adapted from Global Econ
omy 2017)

		Cost of	fТ	Trans	sport (lowest to	highest)			
1	Azerbaijan	52.79		18	Serbia	98.5	35	Germany	141.33
2	Kazakhstan	55.99		19	Latvia	102.02	36	France	141.89
3	Georgia	59.28		20	Lithuania	102.02	37	Ireland	142.3
4	Belarus	59.62		21	Slovakia	105.37	38	Austria	143.7
5	Armenia	64.05		22	Hungary	107.04	39	Finland	152.76
6	Ukraine	65.43		23	Estonia	107.95	40	Netherlands	154.57
7	Moldova	67.75		24	Croatia	112.89	41	Sweden	160.4
8	Russia	71.51		25	Malta	114.37	42	Switzerland	160.56
9	North Macedonia	82.59		26	Cyprus	116.86	43	Denmark	173.28
10	Romania	87.59		27	Spain	117.72	44	Iceland	194.11
11	Bulgaria	89.35		28	Slovenia	119.8	45	Norway	194.38
12	Bosnia and Herzegovina	89.74		29	Greece	120.91	46	Andorra	no data
13	Poland	93.44		30	Portugal	123.44	47	Kosovo	no data
14	Turkey	95.17		31	Luxembourg	124.81	48	Liechtenstein	no data
15	Albania	95.5		32	Italy	135.37	49	Monaco	no data
16	Montenegro	97.47		33	UK	138.72	50	San Marino	no data
17	Czechia	97.56		34	Belgium	141.05			

Based on this ranking it is concluded that France, Ireland, Austria, Finland, Netherlands, Sweden, Switzerland, Denmark, Iceland, Norway, Andorra, Kosovo, Lichtenstein, Monaco and San Marino data should be eliminated from consideration, while the leading markets are Azerbaijan, Kazakhstan, Georgia, Belarus, Armenia, Ukraine, Moldova, Russia, North Macedonia and Romania.

6.6.3 Business freedom index

The business freedom index presented in Table 23 encompasses ten different indicators, including the aspect of starting a business (number of procedures, time measured in days, cost measured in percentage of income per capita, and minimum capital measured in percentage of income per capita), ease of obtaining a license (number of procedures, time measured in days, and cost measured in percentage of income per capita) and ease of closing a business (time measured in years, cost measured in percentage of estate, and recovery rate measured in cents on the dollar). (Global Economy 2022.) The business freedom index is relevant to the DC industry because it provides insights into the business environment, its regulations, and investment opportunities, which can help make an informed decision on optimizing business operations, mitigating risks and capitalizing on growth opportunities.

TABLE 23. Ranking of European countries by business freedom index in 2022 (Adapted from GlobalEconomy 2022)

	Business Freedom (from most to least)														
1	Norway	91	1	8	Slovenia	80		35	Greece	70					
2	Denmark	89	1	9	Poland	79		36	Bosnia & Herz.	67					
3	Finland	89	2	0	UK	79		37	Montenegro	67					
4	Luxembourg	89	2	1	Hungary	77		38	Armenia	65					
5	Netherlands	88	2	2	Malta	76		39	Azerbaijan	65					
6	Estonia	87	2	3	Portugal	76		40	Kazakhstan	65					
7	Germany	87	2	4	Slovakia	76		41	Moldova	64					
8	Ireland	87	2	5	Cyprus	75		42	Russia	63					
9	Lithuania	87	2	6	Spain	75		43	Turkey	63					
10	Sweden	85	2	27	Georgia	74		44	Ukraine	61					
11	Switzerland	84	2	8	Italy	74		45	Belarus	55					
12	Belgium	83	2	9	North Macedonia	74		46	Andorra	no data					
13	Iceland	83	3	0	Serbia	74		47	Kosovo	no data					
14	Austria	82	3	1	Bulgaria	72		48	Liechtenstein	no data					
15	France	82	3	2	Croatia	72		49	Monaco	no data					
16	Latvia	82	3	3	Albania	71		50	San Marino	no data					
17	Czechia	81	3	4	Romania	71									

Based on this ranking it is concluded that Bosnia and Herzegovina, Montenegro, Armenia, Azerbaijan, Kazakhstan, Moldova, Russia, Turkey, Ukraine, Belarus, Andorra, Kosovo, Lichtenstein, Monaco and San Marino should be eliminated from consideration, while the leading markets are Norway, Denmark, Finland, Luxembourg, Netherlands, Estonia, Germany, Ireland, Lithuania and Sweden.

6.6.4 Communication prices

The following communication services index measures the price for internet connection, mobile plans and all other communication services. This factor is relevant to the DC industry because it has a big impact on the operational costs as well as global connectivity and operational efficiency of DCs. TABLE 24. Ranking of European countries by communication prices in 2017 (Adapted from GlobalEconomy 2017)

Communication Prices (from lowest to highest)													
Ukraine	22.55	18	Finland	91.86	3	5	Luxembourg	130.98					
Azerbaijan	28.7	19	Turkey	91.97	3	6	Portugal	131.18					
Belarus	29.95	20	Bosnia & Herz.	93.35	3	7	Sweden	136.86					
Russia	34.81	21	Slovakia	95.08	3	8	Norway	139.76					
Moldova	37.92	22	Austria	95.98	3	9	Switzerland	140.61					
Armenia	38.79	23	Cyprus	100.74	4	0	Netherlands	144.37					
Kazakhstan	41.97	24	Hungary	101.74	4	1	Ireland	152					
Poland	51.3	25	Croatia	103.02	4	2	Spain	155.23					
Romania	56.75	26	Albania	108.62	4	3	Belgium	164.29					
Georgia	57	27	France	109.61	4	4	Iceland	170.54					
Lithuania	68.8	28	Malta	111.12	4	5	Greece	171.14					
North Macedonia	69.13	29	Czechia	112.12	4	6	Andorra	no data					
Serbia	72.23	30	Germany	113.92	4	7	Kosovo	no data					
Estonia	73.92	31	Denmark	114.5	4	8	Liechtenstein	no data					
Bulgaria	74.02	32	UK	122.19	4	9	Monaco	no data					
Latvia	79.46	33	Slovenia	124.17	5	50	San Marino	no data					
Montenegro	90.75	34	Italy	124.31									
	Ukraine Azerbaijan Belarus Belarus Russia Moldova Armenia Armenia Cadata Poland Poland Poland Caorgia Caorgia Caorgia Lithuania Serbia Serbia Serbia Estonia Bulgaria Latvia Montenegro	CommunicationUkraine22.55Azerbaijan28.7Belarus29.95Russia34.81Moldova37.92Armenia38.79Kazakhstan41.97Poland51.3Georgia56.75Georgia56.75Lithuania68.8North Macedonia69.13Serbia72.23Estonia73.92Bulgaria74.02Latvia90.75	Ukraine 22.55 18 Azerbaijan 28.7 19 Belarus 29.95 20 Russia 34.81 21 Moldova 37.92 23 Armenia 38.79 23 Kazakhstan 41.97 24 Poland 51.3 25 Romania 56.75 26 Georgia 57 27 Lithuania 68.8 28 North Macedonia 69.13 29 Serbia 72.23 30 Estonia 73.92 31 Bulgaria 74.02 32 Montenegro 90.75 34	Communication Prices (from 1Ukraine22.5518FinlandAzerbaijan28.719TurkeyBelarus29.9520Bosnia & Herz.Russia34.8121SlovakiaMoldova37.9223CyprusArmenia38.7924HungaryPoland51.325CroatiaGeorgia56.7526AlbaniaNorth Macedonia69.1328MaltaSerbia72.2330GermanyEstonia74.0233SloveniaMontenegro90.7534Italy	Communication Prices (from lowest to 1Ukraine22.5518Finland91.86Azerbaijan28.719Turkey91.97Belarus29.9520Bosnia & Herz.93.35Russia34.8121Slovakia95.08Moldova37.9222Austria95.08Armenia38.7923Cyprus100.74Kazakhstan41.9724Hungary101.74Poland51.326Albania103.02Georgia56.7527France109.61Lithuania68.829Czechia111.12North Macedonia69.1329Czechia112.12Serbia72.2331Denmark114.5Bulgaria79.4633Slovenia124.17Montenegro90.7534Italy124.31	Ukraine22.5518Finland91.863Azerbaijan28.719Turkey91.973Belarus29.9520Bosnia & Herz.93.353Russia34.8121Slovakia95.083Moldova37.9222Austria95.083Armenia38.7923Cyprus100.744Kazakhstan41.9724Hungary101.744Poland51.325Croatia103.024Georgia56.7526Albania108.624Lithuania68.828Malta111.124North Macedonia69.1329Czechia112.124Bulgaria74.0233Slovenia124.174Montenegro90.7534Italy124.315	Communication Prices (from Vest to	Communication Prices (from Users to subsect to su					

Based on this ranking it is concluded that Portugal, Sweden, Norway, Switzerland, Netherlands Ireland, Spain, Belgium, Iceland, Greece, Andorra, Kosovo, Lichtenstein, Monaco and San Marino should be eliminated from consideration, while the leading markets are Ukraine, Azerbaijan, Belarus, Russia, Moldova, Armenia, Kazakhstan, Poland, Romania and Georgia.

6.6.5 Machinery and equipment prices

The following index measures the general price for machinery and equipment. This factor is relevant to the DC industry because it has a big impact on the initial investment costs as well as operational expenses, technology upgrades and efficiency of DCs.

Machinery and Equipment Prices (form lowest to highest)													
1	Russia	79.67	18	Romania	88	35	Netherlands	96.12					
2	Ukraine	80.99	19	Slovenia	88.44	36	Austria	96.18					
3	Georgia	81.55	20	Estonia	88.72	37	Sweden	97.51					
4	Poland	81.64	21	Moldova	89.12	38	Malta	97.99					
5	Hungary	83.07	22	Slovakia	89.42	39	Greece	98.29					
6	UK	83.89	23	Czechia	89.57	40	Switzerland	100.03					
7	Lithuania	84.18	24	Latvia	90.49	41	Finland	100.44					
8	Bulgaria	84.29	25	Azerbaijan	90.62	42	Portugal	103.13					
9	North Macedonia	85.25	26	Cyprus	90.77	43	Denmark	106.04					
10	Belarus	85.69	27	Armenia	91.34	44	Norway	110.82					
11	Turkey	85.88	28	Serbia	91.81	45	Iceland	114.04					
12	Albania	86.72	29	Bosnia & Herz.	92.55	46	Andorra	no data					
13	Croatia	86.83	30	Kazakhstan	92.64	47	Kosovo	no data					
14	Italy	86.97	31	Ireland	92.83	48	Liechtenstein	no data					
15	Montenegro	87.25	32	Belgium	92.95	49	Monaco	no data					
16	Spain	87.41	33	Luxembourg	94.31	50	San Marino	no data					
17	Germany	87.62	34	France	95.16								

TABLE 25. Ranking of European countries by machinery and equipment prices in 2017 (Adapted from Global Economy 2017)

Based on this ranking it is concluded that Austria, Sweden, Malta, Greece, Switzerland, Finland, Portugal, Denmark, Norway, Iceland, Andorra, Kosovo, Lichtenstein, Monaco, San Marino should be eliminated from consideration, while the leading markets are Russia, Ukraine, Georgia, Poland, Hungary, UK, Lithuania, Bulgaria, North Macedonia and Belarus.

6.6.6 Economic preliminary screening conclusion

Based on the economic preliminary screening it can be concluded that the countries with the highest level of economic risks relevant to the DC industry are: Andorra, Kosovo, Lichtenstein, Monaco, San Marino, followed by Norway, Switzerland, Iceland and Sweden; while the countries with the lowest level of socio-cultural risks are: Belarus and Russia.

6.7 Preliminary screening conclusion

Based on the data gathered in this chapter, it can be concluded that the countries with the highest overall number of risks, as illustrated in Figure 10, include Albania, Andorra, Armenia, Austria, Azerbaijan, Belarus, Belgium, Bosnia and Herzegovina, Bulgaria, Cyprus, Estonia, Georgia, Greece, Hungary, Iceland, Kazakhstan, Kosovo, Liechtenstein, Luxembourg, Malta, Moldova, Monaco, Montenegro, North Macedonia, Norway, Romania, Russian Federation, San Marino, Serbia, Slovakia, Switzerland, Turkey, and Ukraine. Therefore, these countries will be excluded from further consideration.

On the contrary, the markets leading in most ranks, as illustrated in Figure 11, are: Sweden being number 1, Luxembourg number 2, Denmark number 3, Netherlands number 4, Norway number 5, Switzerland number 6, Finland number 7, Iceland number 8, Germany number 9, Spain number 10, United Kingdom number 11 and France number 12.

	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18	19	20	21	22	23	24
Albania	х	х	х	х	х	х	х	х	х	х	х													
Andorra	х	х	х	х	х	х	х	х	х	х	х	х	х	х	х	х	х	х	х	х				
Armenia	х	х	х	х	х	х	х	х	х	х	х	х												
Austria	х	х	х	x																				
Azerbaijan	х	х	х	х	х	х	х	х	х	х	х	х	х	х										
Belarus	х	х	х	х	х	х	х	х	х	х	х	х	х	х	х	х	х	х						
Belgium	х	х	х	х																				
Bosnia and Herzegovina	х	х	х	х	х	х	х	х	х	х	х	х	х	х	х	х								
Bulgaria	x	х	х	х	х	х	х																	
Croatia	x	х																						
Cyprus	х	х	х	х	х	х	х																	
Czechia	х	х	х																					
Denmark	x	x	x																					
Estonia	x	x	x	x																				
Finland	x	x																						
France	x	x	x																					
Georgia	x	x	x	х	х	x	x	x	x	x														
Germany	x																							
Greece	x	х	х	х	х	x	x					-						-			-	-	-	-
Hungary	x	x	x	x	-																			
Iceland	x	x	x	x	х	х																		
Ireland	x	x	x																		-			
talv	x	x	x																		-			
Kazakhstan	x	x	x	x	x	x	x	x	x	x	x	x	x	x	x							-	-	
Kosovo	x	x	x	x	x	x	x	x	x	x	x	x	x	x	x	x	x	x	x	x	x	x	x	x
atvia	x	~	~	~	~	~	~	~	~	~	~	n	~	~	~	~	~	~	~	~	~	~	~	~
iechtenstein	x	x	x	x	x	x	x	x	x	x	x	x	x	x	x	x	x	×	x	x	x		-	-
Lithuania	x	~	~	^	~	~	^	^	^	^	~	~	^	^	^	~	^	^	~	~	~			
Luxemboura	x	х	х	x	х	х						-						-		-	-	-	-	-
Malta	x	x	x	x	x	x	x	x	x															
Moldova	x	x	x	x	x	x	x	x	x	x	x	x	x	x	x	x	x	x		-	-	-	-	-
Monaco	v	v	x	×	×	×	×	x	×	×	×	x	×	×	×	×	×	×	x	×	×	v		
Montenearo	x	×	x	x	x	×	x	x	x	x	x	^	^	^	^	~	~	~	~	~	~	~		-
Netherlands	x	x	x	^	^	^	^	^	^	^	^	-						-		-	-	-	-	-
North Macedonia	×	×	×	v	×	v	v	×	×	×	×	×	×	×	×	×	×			-			-	
Norway	×	×	×	×	v	^	^	^	^	^	^	^	^	~	~	~	~	-		-	-	-	-	-
Poland	×	~	~	^	~							-						-		-	-	-	-	-
Portugal	x	x	x		-							-						-		-	-	-	-	-
Romania	×	×	×	×	×															-				
Russian Federation	×	×	×	×	×	v	v	×	×	×	×	v	×	×	×					-			-	
San Marino	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	v	×	×	
Serbia	x	×	×	×	×	×	×	×	×	×	×	^	^	^	^	^	~	~	~	~	~	^	~	-
Slovak Republic	×	×	×	×	^	^	^	^	^	^	~	-						-		-	-	-	-	-
Slovenia	~	×	~	~	-							-						-		-	-	-	-	-
Sioverila	X	×			-							-						-		-	-	-	-	
Sweden	X	X	X	H	-							-						-		-	-	-	-	-
Sweden	X	X	X									-						-		-	-	-	-	-
Switzenand	X	X	X	X	X													-		-	-	-	-	-
i urkiye	X	X	X	X	X	X	X	X	X	X	X	X	X	X	X	-		-		-	-	-	-	-
Ukraine	х	х	х	X	X	X	X	X	X	X	X	X	X	X	х	_		-		-	-	-	-	-
United Kingdom	х	х	Х															_						

FIGURE 10. Preliminary screening conclusion – number and type of identified risks for European countries

	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18
Albania	х	х	х															
Andorra	х	x																
Armenia	х	x	х															
Austria	х	x	x	х	x	х												
Azerbaijan	х	x	х	x														
Belarus	х	х	х	х														
Belgium	х	x	х	x	x	х	х											
Bosnia and Herzegovina	х	х																
Bulgaria	x	x	x															
Croatia	x																	
Cyprus	х																	
Czechia	x	х																
Denmark	x	x	х	х	x	х	х	х	х	x	x	х	x	x				
Estonia	x	x	x	x	x	x	x											
Finland	x	x	x	x	x	x	x	х	х	x	x	х						
France	х	x	x	x	x	x	x	x	x									
Georgia	x	x	x	x														
Germany	x	x	x	x	x	x	x	x	х	x	x							
Greece	x	~	~	^	~	~	A	~	~	~	~					-		
Hungary	x	x	х	х					-									
Iceland	x	x	x	x	x	x	x	x	x	x	x	x		-	-			
Ireland	x	x	x	x	x	x	×	^	^	^	^	~						
Italy	x	×	v	×	×	×	~			-					-	-		
Kazakhstan	x	x	^	^	^	^			-	-				-	-			
Kosovo	~	~	-															
Latvia	x	v	v	x	v	x	×											
Liechtenstein	×	~	×	^	~	^	^	-	-	-				-	-	-		
Lithuania	×	~	×	v		-	-		-	-				-	-	-		
Luxembourg	×	×	×	x	x	x	×	×	×	×	v	v	×	×	×			
Malta	×	~	×	~	*	~	^	*	~	^	~	~	~	*	~	-		
Moldova	×	~	~	-		-	-		-	-				-	-	-		
Monaco	×	×	-			-		-	-	-					-	-		
Montenegro	×	×	-			-	-	-	-	-				-	-			
Netherlande	×	×	¥	v	v	v	v	v	v	~	v	v	v	-	-	-		
North Magadania	X	X	X	X	X	X	×	X	X	×	X	X	X	-	-			
North Macedonia	X	X	X												-	-		
Reland	X	X	X	X	X	X	X	X	X	×	X	X	X	_	-	-		
Poland	X	X	X	X	X		-		-	-						-		
Ponugai	×	X	×	X			-		-									
Runnian Enderation	X	X	X	X	X				-	-				-	-	-		
San Marina	X	X	X	X	X	X	X	-	-	-				-	-	-		
Sarhia						-	-		-	-					-	-		
Slovak Republic	v	~	_			-	-			-					-	-		
Slovan Republic	X	X	v			-	-	-	-	-					-	-		
Sioverna	X	X	X	X											-	-		
Spain	X	X	X	X	X	X	X	X	X	X	X							
Sweden	X	X	X	X	X	X	X	X	X	X	X	X	X	X	х	x	X	X
Switzenand	X	X	X	X	X	X	X	X	X	x	X	X	X		-	-		
i urkiye	X	X				_	-		-	-					_	-		
Ukraine	х	x	x	х												_		
United Kingdom	х	х	х	Х	Х	х	х	х	х	х								

FIGURE 11. Preliminary screening conclusion - ranking of leading European countries

The markets that we will continue to consider for the market opportunity analysis are: Croatia, Czechia, Denmark, Finland, France, Germany, Ireland, Italy, Latvia, Lithuania, Netherlands, Poland, Portugal, Slovenia, Spain, Sweden and United Kingdom.



FIGURE 12. The most viable markets after preliminary screening

7 MARKET ENTRY CONSIDERATIONS FOR DATA CENTER EXPANSION

Europe offers a wide range of opportunities for local DC industry expansion thanks to its well-developed digital infrastructure. An effective market opportunity analysis is crucial to navigate the wide range of local market specifics and identify the optimal ones. Based on the results of the market selection analysis, the most viable identified options will be considered, excluding the markets that are already part of the most developed industries in order to achieve the thesis purpose of continuing the industry expansion in an uniform, equitable way. Therefore the countries analysed in this chapter will be: Croatia, Czechia, Denmark, Finland, Ireland, Latvia, Lithuania, Portugal and Slovenia. For each of these countries, considerations regarding market entry and marketing mix strategy will be discussed.

7.1 Croatia

Croatia's market has been experiencing significant growth due to several key factors including its developing digital infrastructure, its strategic geographic location between Western Europe and the Balkans and its sustainable energy resources (Statista 2023). While the current local DC industry is not as well developed as in other European countries (Figure 1), there has been a rising demand for DC services, especially in large cities, like Zagreb and Split, from companies looking to outsource their digital infrastructure and storage needs (Statista 2023). Croatia's growing touristic industry could also lead to a future increased need for cloud services to support the hospitality, leisure and related sectors (Datacenters 2022).

Partnerships or joint ventures could be the best market entry mode thanks to the wide range of available local expertise. Focusing on reliability and security in data handling could be a key selling point for Croatia, as the tourism industry is highly dependent on customer trust and reputation regarding secure data handling (Wiboonrat 2014). A local DC with high quality data security measures would help build trust amongst local businesses handling sensitive customer information, such as hotels, tour operators and travel agencies. Ensuring a stable and secure data handling environment minimizes the risk of system failures or data loss, thereby ensuring smoother business operations and continuity. In a market where digital services are growing but might not yet be fully matured, highlighting a strong emphasis on data reliability and security can be a competitive advantage as it allows DC service providers to create a niche by offering a higher level of assurance compared to in house management (Statista 2023).

The product offerings should be focused on cloud services that ensure secure and reliable data management for hospitality businesses. The pricing model should be competitive and flexible enough to accommodate the financial constraints of smaller businesses within the tourism industry (Mohamed 2014). The promotional strategies should focus on highlighting the reliability and security of the data management, including targeted digital campaigns.

7.2 Czechia

Czechia is reported to have one of the most powerful IT sectors in the world (ReportLinker 2022) and a growing demand for DC services (Arizton 2021). Its strategic location in Central Europe makes for a good opportunity to open a hub supporting future regional expansion, as the country is already connected to the major European computing nodes via fiber networks (Arizton 2021). Industries like automotive, manufacturing, gaming and IT could be potential beneficiaries of local DCs, as these sectors require a data infrastructure capable of handling complex technological demands.

The best market entry modes would be a direct investment approach, possibly establishing whollyowned subsidiaries due to the country's developed tech infrastructure, as well as local partnerships or joint ventures thanks to its low amount of restrictions per "OECD National Treatment for Foreign-Controlled Enterprises" (Appendix 6). The pricing strategies should be flexible and accommodate the wide range of demands of the industries on the Czech market. Promotional efforts should focus on cutting-edge technology and data security through industry specific events and targeted advertising. Leveraging Czechia's reputation as a technology driven nation provides a competitive advantage (Embassy of Czech Republic 2024) as it positions DC service providers as industry leaders capable of offering advanced data solutions.

Czechia's main selling point as a connecting IT hub in Central Europe with one of the best IT sectors and technological expertise in Europe (Embassy of Czech Republic 2024), provide a strong foundation for DC service providers to offer cutting-edge solutions tailored to the various demands of the market and establish a distinct competitive advantage.

7.3 Denmark

Denmark has a focus on sustainability and innovation, constantly ranking first in the environmental performance index (Ministry of Foreign Affairs of Denmark 2022) and the European innovation scoreboard (Ministry of Foreign Affairs of Denmark 2023). A DC designed for energy efficiency and high speed connectivity would align well with its environmentally conscious and tech savvy population (Chapter 7). Creating a digital hub in the Copenhagen region would be a compelling market opportunity considering the favourable infrastructure and the government's support for local tech innovation (International Trade Administration 2024).

Entering the Danish market could be done through a combination of wholly-owned subsidiaries and partnerships thanks to the local abundance of know-how and quality infrastructure (Barnett & Casasbuenas 2019). The DC design should focus on sustainability, energy efficiency, high speed connectivity and data security. Pricing strategies should reflect the value of a reliable, sustainable design. The promotion should be done via targeted digital campaigns and industry forums and they should focus on the use of green technology and high speed connectivity.

The Danish government actively supports technological innovation and provides a business friendly environment (APPENDIX 6), favourable tax policies and supportive regulations for the tech industry, encouraging business investment and growth. Denmark's strategic geographic location in Northern Europe offers good connectivity with both the Scandinavian countries and continental European markets. Its proximity to major European cities makes a good opportunity for establishing a DC that could serve a broad regional market.

Overall, Denmark's competitive advantage is its commitment to sustainability, tech savvy population, supportive government policies, strategic geographic location, and reliable infrastructure. These factors make Denmark a good opportunity for local DC industry expansion.

7.4 Finland

Finland is renowned for its tech industry driven by innovation (World Intellectual Property Organization 2022) and it has a high demand for strong data infrastructure focused on innovation, security (Table 18) and sustainability (Table 8). Its tech industry is thriving thanks to its innovative yet sustainable practices and the Helsinki digital hub would provide a great opportunity for DC expansion. Focusing on tailored solutions for the gaming and IoT sectors could be its main selling point (Angesleva 2023).

Finland's large share of renewable energy sources (mainly hydro and wind power) could significantly reduce DC carbon footprints which is a great competitive advantage when it comes to offering sustainable and economic energy solutions for DCs (Statistics Finland 2021). Its commitment to eco-friendly practices and design is well suited to the increasing global demand for sustainable data solutions (Munde 2024). The Finnish population is one of the most tech literate populations globally, which also plays an important role in facilitating the adoption and implementation of advanced technologies within the DC sector and it provides a pool of skilled workforce as well as a receptive market for innovative data services (Sear 2023).

Entering the Finnish market could be done through a direct investment strategy thanks to its techsavvy population. The new DCs should focus on innovation, security, sustainability and efficient cooling technologies. Furthermore, the Finnish government actively supports technology and innovation, offering incentives, funding, and favourable policies for tech related ventures (Technology Industries of Finland 2020) which contributes to an optimal environment for technological advancements and it attracts businesses looking for supportive regulations (Technology Industries of Finland 2023). Product offerings should be focused on innovation and security for industries like gaming and IoT (Neogames 2018). The pricing model should be flexible and reflect scalability and innovation. Promotional strategies should also revolve around innovation and security features in tech oriented events and targeted advertising.

Finland's main selling point in the DC industry is its innovative technological environment, well developed infrastructure, commitment to sustainability, tech-savvy population and government support, making it an optimal market for advanced DCs. The country is home to a wide range of startups as well as renowned tech companies thanks to its innovative culture, making it an ideal location for cutting-edge DC technologies (Rihti 2019). Finland's expertise in tech innovation positions it as a hub for pioneering solutions in data management and infrastructure and its high-quality infrastructure, including reliable energy sources and advanced telecommunications networks, along with its strategic geographic location provides favourable access to both European and global markets, making it an appealing choice for businesses looking for efficient connectivity (Haaramo 2015).

7.5 Ireland

Ireland has a favourable business environment thanks to its pool of skilled workforce and favourable tax policies proven by the fact that it already hosts multiple major tech companies in its existing DC hub (Stenson 2019). The local DC market is known to be on a continuous growth track (Miller 2013) which offers great expansion opportunities focused on the scalability of the data infrastructure. New DCs would be required to respond to local needs of rapid growth in order to keep up with the evolving cloud services. Ireland's business-friendly environment encourages expansion and innovation (Tech Central 2020) as both local companies and MNCs are looking for data solutions that would be capable of scaling along with their business growth. Scalable data services allow companies to innovate without being limited by the infrastructure (GemPool 2021).

Entering the market could be done through wholly-owned subsidiaries that could benefit from Ireland's well established infrastructure as well as local partnerships or joint ventures thanks to its low amount of restrictions per "OECD National Treatment for Foreign-Controlled Enterprises" (Appendix 6). The marketing mix could centre on providing high quality services, secure data management and scalability to meet the demands of tech companies and cloud services. A reliable value based pricing would be recommended for this market considering the local competition. Promotion strategies that could work best should be done through digital marketing campaigns and at industry specific events.

Ireland's tech hub position, its prioritization of cloud services and the digital economy, its business friendly environment, and the diverse industry presence collectively show the importance of scalability (Savvas 2021). Offering scalable data services addresses the varying and growing needs of businesses, making Ireland an optimal market for companies looking for flexibility and adaptability.

7.6 Latvia

Positioned between the Nordic and Eastern European countries, Latvia's strategic location in the Baltic area makes it an ideal regional hub for DCs that could bridge the neighbouring areas and cover a wide customer base (Dolzhenkova & Mokhorov 2020). The geographic advantages facilitate efficient connectivity and data transmission, making Latvia an optimal market for expanding the DC industry in the region, allowing opportunities for future growth. Furthermore, the country's technological industry has

been thriving in recent years and the e-government initiatives (OECD Library 2021) offer strong regional connectivity and innovation opportunities for potential DC operations (Institute of Economics of the Latvian Academy of Sciences 2023).

Entering the market with new DCs may work best through joint ventures or partnerships thanks to Latvia's national programmes focused on facilitating international partnership for the ICT sector (Digital Skills & Jobs Platform 2024) and its low amount of restrictions per "OECD National Treatment for Foreign-Controlled Enterprises" (Appendix 6). Collaborating with local tech firms and infrastructure providers would facilitate access to local expertise. Wholly-owned subsidiaries could also be a viable market entry, as Latvia is ranking 14th amongst 192 countries on the ease of doing business index (Multiplier 2024). The pricing strategy should match the regional market levels and the promotional strategy could be realised through campaigns focused on local technological innovation and secure data management.

DCs focused on reliability and strict security regulations offering secure data management in both the public and private sectors coincides with general governmental priorities. Latvia's focus on e-government initiatives proves its commitment to digital transformation which would create demand for secure data services specialised in support of governmental operations, presenting an opportunity for DCs to market to the public sector (European Commission 2022). Complying with the local regulations and ensuring secure, reliable data handling could be the key selling point of a Latvian DC.

7.7 Lithuania

Lithuania invested significantly in developing its ICT infrastructure including high-speed connectivity, reliable digital networks, cutting-edge technology and is now home to the tech hub ranking first globally (Invest Lithuania 2022). These technological advancements have turned Lithuania into an optimal market for DC operations that would require advanced infrastructure and connectivity. The Lithuanian government actively supports technological initiatives and its proactive attitude in offering incentives, funding, and regulations for tech-related ventures encourages investment and growth in the tech sector, making Lithuania an optimal market for international businesses needing governmental backing (Euro-Start Entreprises 2023). Entering Lithuania's DC market could involve joint ventures or partnerships with local tech companies as it's currently home to over 150 MNC (Ruzgys 2022). Given Lithuania's well developed ICT infrastructure and its famous Vilnius innovation hub (Vilnius City Innovation Industrial Park 2022), a wholly-owned subsidiary could work best with the local advanced tech environment (Dimoska 2020).

Lithuania's focus on R&D initiatives supports technological advancements and presents a good opportunity for DCs looking to establish a leading position in technological innovation and offer cuttingedge solutions with a focus on education in technology-related fields, training workforce and providing opportunities for tech talents. This skilled network could be advantageous for DCs needing technological expertise and innovation. The pricing models should be flexible and the promotion could be done through events and targeted advertising. (Interreg Europe 2018.)

7.8 Portugal

Portugal is known for its impressive progress with renewable power, particularly wind and solar energy, as it reached the point where 95% of its consumption is covered by sustainable sources (Eurostat 2019). This achievement makes Portugal a leading country in sustainability, opening it to DC construction opportunities that could operate using eco-friendly practices (Euronews Green 2024). Leveraging renewable energy sources allows DCs to reduce their carbon footprints which helps with global sustainability goals. Also, its strategic location at the crossroads between Europe, Africa, and the Americas provides advantageous global connectivity as Portugal is one of the few countries in the world that has direct connections with 5 different continents (White Papers 2022). Portuguese DCs could serve as strategic hubs for global data distribution, targeting global markets and promoting global connectivity.

Expanding the DC industry in Portugal could involve joint ventures to leverage local partnerships or subsidiaries thanks to its established tech infrastructure and strategic connectivity (Goncalves 2021) (Tech Explore 2021) thanks to its low amount of restrictions per "OECD National Treatment for Foreign-Controlled Enterprises" (Appendix 6). Portugal's emphasis on renewable energy and strategic connectivity opens up opportunities for eco-friendly DCs, highlighting sustainable practices, global connectivity, and digital transformation initiatives (López-Dóriga 2022). The recommended pricing models would be value-based and the promotion should be focused on the sustainable infrastructure. The country's strategic location makes it an attractive destination for MNCs looking to centralize their data operations in a place with efficient global connectivity. Local DCs could capitalize on this advantage to attract international businesses. Portugal is known to actively pursue digital transformation initiatives, digital economy and innovative technological solutions (International Trade Administration 2024) which would allow local DCs to focus their operations on advanced digital infrastructure and digitalization.

7.9 Slovenia

Slovenia's strategic location in central Europe presents an advantage as it could form a bridge with Central and Southeastern European markets (International Trade Administration 2024). This position provides efficient connectivity and proximity to a multitude of customer bases which would allow a DC to serve as a central point for businesses focused on access to these regions. Slovenia's increasing technological innovation makes it an ideal home for startups, tech initiatives and innovation development (Slovenia Business Development Agency 2022). This type of environment is optimal for technological advancement, making the Slovenian market a favourable place for DCs looking for collaboration opportunities and access to growing tech talent.

Entering Slovenia's DC market might involve partnerships with local tech companies or wholly-owned subsidiaries, given its central European location and growing technological innovation supported by governmental initiatives and its low amount of restrictions per "OECD National Treatment for Foreign-Controlled Enterprises" (Appendix 6). Slovenia's location and tech drive present market potential for tailored data services for regional access, focusing on innovation (Vekic, Djakovic, Borocki, Sroka, Popp & Olàh 2020). The pricing should take into account regional market demands and implement promotional strategies focused on technological innovation.

Slovenia is known to be focused on R&D initiatives proven by its allocation of over 2% of its GDP to R&D out of which 75 % were provided by the business sector contributing to technological advancements (Slovenia Business 2023). DCs aligning with these government supported initiatives could leverage R&D investments and collaborations and become leaders in technological innovation in the DC industry. In conclusion, the European markets considered for future DC industry expansion, Croatia, Czechia, Denmark, Finland, Ireland, Latvia, Lithuania, Portugal, and Slovenia, have a different range of strengths and local market opportunities. Leveraging these distinct advantages, including technological innovation or sustainability initiatives, regional connectivity, and government support, would allow a DC to design their approach in order to tap into locally specific market demands and build a strong position within these regions. Understanding each country's unique selling point and aligning the DC operation with it, is key to a successful market expansion.

8 CONCLUSIONS

Expanding a business globally has endless potential, but success depends on careful planning and analysis. The theoretical framework of this thesis focuses on the essential role of market opportunity analysis and market research for international expansion. The DC industry is becoming a foundation of the digital environment and its evolution shows a mix of technological advancement and increasing data demand. Cloud computing and edge computing's rise continue to redefine operational standards, while sustainability and cybersecurity remain important concerns within the industry. The European DC market has been continuously growing and incorporating an increasing amount of sustainable initiatives and data protection regulations, however, the inconsistencies that can be seen amongst its countries call for a need for strategic site selection when considering the future of the European DC industry.

8.1 Discussion and summarizing conclusion

The main purpose of this thesis was to understand the DC industry in Europe and to explore opportunities for its future expansion. In order to do so, an overview of the industry and how it came to function as it does today was developed by researching the reasons why the DC industry developed unevenly across Europe and uncovering what were the most important expansion deciding factors, which helped provide an industry guideline on the priorities that need to be considered in order for the best possible outcome to be reached. Learning the driving factors of the expansion of this industry by researching factors that ensure suitability for DC development design uncovered the most significant strengths and weaknesses that need to be considered in the decision-making process. (Chapter 5.)

In order to strategically choose a site suitable for building a new DC, there are a wide range of factors that must be evaluated ranging from labour and utilities cost to infrastructure accessibility. One of the main concerns is the energy efficiency which most DC designers are mitigating by implementing innovative designs based on renewable energy and carefully assessing factors that could potentially have a negative impact on the environment. Another concern would be the construction cost and performing a comparative analysis on a certain geographic segment allows the opportunity to make an informed decision when selecting a new market. Power efficiency, natural disaster risks, infrastructure accessibil-

ity, and environmental concerns remain key elements when considering the construction of a cost-effective and sustainable DC. Since multiple factors need to be investigated in the strategic selection process, an integral approach would work best and it would allow the company to make an optimal, informed decision that could be both efficient operationally and sustainable long term. (Chapter 5.)

The political preliminary screening revealed many differences amongst the European countries when it comes to political stability, foreign investment restrictions, corruption risks, security threats, and limitations on foreign-controlled enterprises. Norway, Denmark, Germany, Iceland, Ireland, Luxembourg, Netherlands, and Sweden benefit from higher political stability, fewer restrictions on foreign investments and enterprises, lower corruption risks, and reduced security threats, making them optimal for potential DC investments from a political perspective. Contrarily, Moldova, Kazakhstan, Turkey, Russia, Ukraine, Albania, Belarus, Kosovo, Bosnia and Herzegovina, Azerbaijan, Andorra, Liechtenstein, Luxembourg, Monaco, and San Marino present higher political risks and could pose challenges with political instability, restrictions, corruption, and security threats. (Subchapter 6.1.)

The legal preliminary screening also revealed a wide variety of differences amongst the European countries when it comes to trade freedom, corporate income taxes, investment freedom, fiscal freedom, and legal stability. Spain and Switzerland have the highest legal stability, lowest corporate income taxes, high investment and fiscal freedom, and better trade conditions compared to other countries, making them optimal for potential DC investments from a legal perspective. Contrarily, Andorra, Kosovo, Liechtenstein, Monaco, North Macedonia, San Marino, Belarus, Bosnia and Herzegovina, Moldova, Russia, and Turkey present higher legal potential risks and could pose challenges with trade restrictions, high corporate income taxes, limited investment and fiscal freedom, and potential legal instabilities. (Subchapter 6.2.)

The ecological preliminary screening also revealed a wide range of differences amongst the European countries when it comes to the share of renewable energy and the emissions index. Belgium, Ukraine, Luxembourg, Malta, Russia, Belarus, Moldova, Kazakhstan, San Marino, Turkey, and Monaco have higher potential ecological risks due to lower shares of renewable energy and higher emissions indexes. Contrarily, Iceland, Norway, Switzerland, Sweden, Albania, Finland, Latvia, Montenegro, Bosnia and Herzegovina, Austria, Spain, Kazakhstan, Italy, Poland, Ukraine, France, and the UK benefit from better ecological stability with higher shares of renewable energy and lower emissions indexes, making them optimal for local potential DC development from an ecological perspective. (Subchapter 6.3.)

The technological preliminary screening of the European market showed diverse technological landscapes across its countries as well. Cyprus, Bosnia and Herzegovina, Albania, Turkey, North Macedonia, Greece, Armenia, Iceland, Georgia, Azerbaijan, Kosovo, Lichtenstein, and San Marino, indicate potential risks when it comes to internet speed, bandwidth, and electricity production capacity, suggesting potential limitations for DC efficiency. Contrarily, Monaco, Romania, Switzerland, Denmark, France, Hungary, Spain, Sweden, Luxembourg, Andorra, Luxembourg, Malta, Sweden, Iceland, the UK, and the Netherlands benefit from higher technological readiness across multiple indexes, making them optimal for local potential DC development from an ecological perspective. (Subchapter 6.4.)

The socio-cultural preliminary screening also revealed a wide variety of differences amongst the European countries when it comes to the rule of law, globalisation index, and attitude towards immigrants. Bulgaria, Montenegro, North Macedonia, Serbia, Armenia, Albania, Bosnia and Herzegovina, Moldova, Turkey, Kazakhstan, Azerbaijan, Ukraine, Russia, Belarus, Kosovo, Monaco, and San Marino are the countries indicating the highest potential risks, determined by weak rule of law and limited social globalisation. Contrarily, Finland, Norway, Denmark, Switzerland, Austria, Luxembourg, Iceland, Lichtenstein, the Netherlands, and Sweden are the leading markets benefiting from a strong rule of law, higher social globalization, and positive attitudes towards immigrants, making them optimal for local potential DC development from a socio-cultural perspective. (Subchapter 6.5.)

Finally, the economic preliminary screening revealed that Bulgaria, Romania, the UK, Georgia, North Macedonia, Moldova, Belarus, Azerbaijan, Turkey, Ukraine present higher potential risks dealing with high labour costs. Contrarily, Finland, Italy, Spain, Denmark, France, Belgium, Montenegro, Germany, Sweden, and Luxembourg have optimal labour cost conditions within the region. Transport prices, business freedom, communication costs, and machinery prices revealed France, Ireland, Austria, Finland, Netherlands, Sweden, Switzerland, Denmark, Iceland, Norway to have optimal conditions for local DC expansion. Additionally, Ukraine, Azerbaijan, Belarus, Russia, Moldova have some of the lowest communication and machinery prices which brings potential cost-effective opportunities for DC construction. While Andorra, Kosovo, Lichtenstein, Monaco, San Marino, Norway, Switzerland, Iceland, and Sweden have high economic risks that could potentially impact the DC industry, Belarus and Russia show a lower level of economic risk. (Subchapter 6.6.)

The overall conclusions of the PESTEL based preliminary screening and market opportunity analysis for future expansion of the DC industry in Europe indicated that Croatia, Czechia, Denmark, Finland, Ireland, Latvia, Lithuania, Portugal, and Slovenia are the countries with the biggest strengths making them optimal potential locations. Each of these countries have specific advantages and selling points that DC operators could leverage for a successful market entry. (Subchapter 6.7.)

Croatia is positioned in a well-developed touristic area as the most visited country in Central and Eastern Europe (CEE) (Statista 2023) and accentuating its reliability when it comes to handling data securing is its main selling point. Czechia's IT sector and technological expertise are one of the best in Europe (Embassy of Czech Republic 2024), making it an ideal location for a DC focused on advanced technology and innovation as its key selling points. Denmark has a focus on sustainability and innovation, constantly ranking first in the environmental performance index (Ministry of Foreign Affairs of Denmark 2022) and the European innovation scoreboard (Ministry of Foreign Affairs of Denmark 2023) and a DC designed for energy efficiency and high-speed connectivity would align well with its environmentally conscious and tech savvy population. (Chapter 7.)

Finland is renowned for its tech industry driven by innovation (World Intellectual Property Organization 2022) and it has a high demand for strong data infrastructure focused on innovation, security and sustainability. Its main selling point would be tailored solutions for the gaming and IoT sectors (Angesleva 2023). Ireland's scalability stands out as a strong selling point in Ireland's tech industry (Deloitte 2021), supporting changing business demands when it comes to data needs, especially for startups and multinational companies. Latvia's strategic geographic location along with its thriving tech industry and e-government initiatives (OECD Library 2021) offers strong regional connectivity and innovation opportunities for potential DC operations (Institute of Economics of the Latvian Academy of Sciences 2023). Lithuania's strong ICT infrastructure, innovation hubs, and government support (Ruzgys 2022), as well as its position make it an optimal location for a DC focused on an environment driven by tech development and innovation. (Chapter 7.)

Portugal focuses on renewable energy (Euronews Green 2024), global connectivity (AICEP Portugal Global 2023), and digital transformation initiatives (Portugal Digital 2020) which makes it a sustainable and globally connected location for data operations. Slovenia's central European positioning along with its growing tech innovation and governmental initiatives (Slovenia Business Development Agency 2022) offer strategic advantages for DCs focused on regional connectivity and innovation hubs. (Chapter 7.)

A complex range of influential factors need to be investigated when researching potential market opportunities for the future of the European DC industry expansion. Since most countries present a certain set of advantages as well as disadvantages, it is important to first learn which factors are the most relevant for the industry in question, and how they need to be prioritised in order to be able to best interpret the research discoveries. Factors such as cost effectiveness, technological development, sustainable infrastructure, political stability, supportive legal framework or high social globalization become more relevant when assessed as an overall regional performance rather than on their own and this type of balance is crucial for optimal construction and operation of a DC.

8.2 Ethics and reliability of the thesis

Ensuring ethical integrity within academic research is essential conduct amongst the research community and this master's thesis complies with all the established recommendations and guidelines as advised by the Finnish National Board on Research Integrity (TENK) to hold ethical standards. Throughout the author's research process, ethical considerations such as ascertaining of conflicts of interest, acquainting with the thesis topic, ascertaining of resources (Nareaho, Kettunen, Karki, Paallysaho 2020) were prioritized to ensure adherence to the community's responsible conduct.

Given that all the data utilized for this research was secondary, there was no requirement for individual data usage consent. However, ethical principles were sustained through careful examination of the sources that were used, their quality and compliance to the principles of data privacy and confidentiality. The selection and use of secondary data were conducted conforming with the relevant regulations and guidelines to ensure the integrity and reliability of the research findings. Moreover, efforts were made to critically assess the ethical implications of utilizing secondary data, including considerations of data ownership, copyright, and potential biases within the sources (Nareaho, Kettunen, Karki, Paal-lysaho 2020). The reliability of this thesis is guaranteed by the quality of the secondary data that was used which was only selected from organizations that are generally considered reliable (e.g. World Bank, Statista, Eurostat). The reliability is also enhanced by the level of depth into which the analysis was performed. In conclusion, this thesis was written with commitment to the ethical research practices, ensuring that all the aspects of the study comply with the recognized ethical recommendations and guidelines.

REFERENCES

Albaum, G., Duerr, E & Josiassen, A. 2011. *International marketing and export management*. Eighth edition. Harlow: Pearson.

Angesleva, J.,P. 2023. Finland's Innovation Odyssey: A Global Competitor in the Making. Available at: https://www.linkedin.com/pulse/finlands-innovation-odyssey-global-competitor-making-%C3%A4ngeslev%C3%A4. Accessed 09 March 2024

AICEP Portugal Global. 2023. Infrastructure. Available at: https://www.portugalglobal.pt/en/investment/why-portugal/reasons-to-invest/infrastructure/#:~:text=The%20country%20also%20boasts%20an,all%20continents%20(excluding%20Antarctica). . Accessed 09 March 2024

Arizton. 2021. CZECH REPUBLIC DATA CENTER MARKET - INVESTMENT ANALYSIS & GROWTH OPPORTUNITIES 2021-2026. Available at: https://www.arizton.com/market-re-ports/czech-republic-data-center-market . Accessed 07 March 2024

Barnett G., Casasbuenas J. 2019. Digital Frontrunners Spotlight: Denmark. Available at: https://www.nesta.org.uk/blog/digital-frontrunners-spotlight-denmark/ . Accessed 08 March 2024

Balkan Green Energy News. 2022. North Macedonia's green energy output rose 14.7% in 2021. Available at: https://balkangreenenergynews.com/north-macedonias-green-energy-output-rose-14-7-in-2021/. Accessed 15 January 2023

Balkan Green Energy News. 2022. Serbia's ambitious energy security plan targets 50% renewables share by 2040. Available at: https://balkangreenenergynews.com/serbias-ambitious-energy-security-plan-targets-50-renewables-share-by-2040/. Accessed 15 January 2023

Balodis R. & Opmane I. History of Data Centre Development. 2019. Available at: https://dl.ifip.org/db/series/ifip/ifip387/BalodisO12.pdf . Accessed at 10 December 2023

Bankwatch Network. 2020. The energy sector in Kosovo. Available at: https://bankwatch.org/beyond-fossil-fuels/the-energy-sector-in-kosovo . Accessed 15 January 2023

Baxtel. 2020. Google Hamina Finland. Available at: https://baxtel.com/data-center/google-hamina-finland . Accessed 19 December 2022

Butcher M. 2021. Locals share why Vilnius, Lithuania is becoming an international startup hub. Available at: https://techcrunch.com/2021/09/07/locals-share-why-vilnius-lithuania-is-becoming-an-international-startup-hub/?guccounter=1 . Accessed 10 March 2024

Cahill B. 2022. Developing Renewable Energy in Ukraine. Available at: https://www.csis.org/analy-sis/developing-renewable-energy-ukraine . Accessed 15 January 2023

Carter, M. & Hines, R. 2019. Data Centers: Choosing an optimal site location. Questline. Available at: https://aeped.com/wp-content/uploads/2019/12/Choosing-an-Optimal-Data-Center-Site.pdf . Accessed 19 December 2022

Chong, K. W. 2003. The Role of Pricing in Relationship Marketing - A Study of the Singapore Heavy Equipment Spare Parts Industry. PhD Dissertation. International Graduate School of Management. University of South Australia.

Cohen, L., Manion, L, & Morrison, K. 2018. *Research Methods in Education*. Eighth edition. London: Routledge.

Constantinides, E. 2006. The Marketing Mix Revisited: Towards the 21st Century Marketing by E. Constantinides. In Möller, K. (ed) *Journal of Marketing Management*. Oxfordshire: Taylor & Francis, 407-438.

Datacenters. 2022. About Croatia Data Centers Market. Available at: https://www.datacenters.com/locations/croatia . Accessed 07 March 2024

De Sousa D. 2018. Examining Cloud Computing Technologies Through the Internet of Things. Available at:https://www.linkedin.com/pulse/examining-cloud-computing-technologies-through-things-de-sousa-mba . Accessed 12 January 2024

Deutsche Energie-Agentur GmbH. 2021. Factsheet: Renewable Energy in Albania. Available at: https://unece.org/sites/default/files/2021-07/UNECE-RE_Uptake_Factsheet_Albania.pdf . Accessed 15 January 2023

Deloitte. 2021. Looking to the clouds: are Irish organisations amplifying the transformational value of cloud technology? Available at: https://www.deloitte.com/ie/en/services/consulting/analysis/cloud-in-dex-survey.html . Accessed 09 March 2024

Device42. 2023. Best Practices for Data Center Sustainability. Available at: https://www.de-vice42.com/data-center-infrastructure-management-guide/data-center-sustainability/ . Accessed 12 January 2024

Digital Skills & Jobs Platform. 2024. Latvia - eSkills Partnership. Available at: https://digital-skills-jobs.europa.eu/en/about/national-coalitions/latvia-eskills-partnership . Accessed 10 March 2024

Dimoska E. 2020. The self-titled "G-spot of Europe": Lithuania's startup ecosystem at a glance. Available at: https://www.eu-startups.com/2020/07/the-self-titled-g-spot-of-europe-lithuanias-startup-eco-system-at-a-glance/ . Accessed 10 March 2024

Directorate-General for Energy. 2024. Commission adopts EU-wide scheme for rating sustainability of data centres. Available at: https://energy.ec.europa.eu/news/commission-adopts-eu-wide-scheme-rat-ing-sustainability-data-centres-2024-03-15_en. Accessed 20 March 2024

Doole, I., Lowe, R & Kenyon, A. 2017. *International marketing strategy: Analysis, development and implementation*. Ninth edition. Hampshire: Cengage Learning.

Dolzhenkova E. & Mokhorov D. 2020. Latvia's role in the Arctic. Cooperation prospects. Available at: https://iopscience.iop.org/article/10.1088/1755-1315/539/1/012046/pdf . Accessed 09 March 2024

Embassy of Czech Republic. 2024. ICT and Data Centers. Available at: https://mzv.gov.cz/beijing/en/trade_and_economy/doing_business_in_cz/investing_in_the_cr/ict_and_data_centers/index.html#:~:text=The%20Czech%20Republic%20is%20one,tradition%20of%20outstanding%20technical%20fields . Accessed 09 March 2024

Euronews Green. 2024. Renewables are meeting 95% of Portugal's electricity needs. How did it become a European leader? Available at: https://www.euronews.com/green/2024/05/10/renewables-aremeeting-95-of-portugals-electricity-needs-how-did-it-become-a-european-lead#:~:text=Portugal%20has%20made%20some%20huge,fleet%20with%20added%20storage%20capacity . Accessed 15 May 2024

EuroStart Entreprises. 2023. Why entrepreneurs are starting up in Lithuania. Available at: https://www.eurostartentreprises.com/en/business-advice/why-entrepreneurs-are-starting-up-in-lithuania . Accessed 10 March 2024

Eurostat. 2022. Electricity price statistics. Available at: https://ec.europa.eu/eurostat/statistics-ex-plained/index.php?title=Electricity_price_statistics . Accessed 03 January 2023

Eurostat. 2023. Purchasing power parities (PPPs), price level indices and real expenditures for ESA 2010 aggregates. Available at: https://ec.europa.eu/eurostat/databrowser/view/PRC_PPP_IND__cus-tom_3623484/bookmark/table?lang=en&bookmarkId=1e3001b1-5091-4486-a53d-c8bb41f322d2 . Accessed 03 January 2023

Eurostat. 2019. Share of renewable energy in the EU up to 17.5% in 2017. Available at: https://ec.europa.eu/eurostat/documents/2995521/9571695/8-12022019-AP-EN.pdf/b7d237c1-ccea-4adc-a0ba-45e13602b428 . Accessed 10 March 2024

Eurostat. 2020. Tax-to-GDP ratio at 41.1% in EU?. Available at: https://ec.europa.eu/eurostat/documents/2995521/11469100/2-29102020-BP-EN.pdf/ . Accessed 15 January 2023

Eurostat. 2020. What is the share of renewable energy in the EU?. Available at: https://ec.europa.eu/eurostat/cache/infographs/energy/bloc-4c.html. Accessed 15 January 2023.

European Commission. 2022. Latvia in the Digital Economy and Society Index. Available at: https://digital-strategy.ec.europa.eu/en/policies/desi-latvia . Accessed 10 March 2024

Faist Group. 2023. Why are data centres so important?. Available at: https://www.faistgroup.com/news/data-centres/#:~:text=Data%20centres%20are%20designed%20to,with%20other%20IT%20infrastructure%20components . Accessed at 10 December 2023

Fortinet. 2023. How Does Data Center Security Work?. Available at: https://www.fortinet.com/re-sources/cyberglossary/data-center-security. Accessed 12 January 2024

GemPool. 2021. Data Centres In Ireland: The Ever-growing Industry. Available at: https://www.gem-pool.ie/insights/data-centres-in-ireland-the-ever-growing-industry . Accessed 09 March 2024

Gigerich, L. 2012. Critical Factors: The Data Center Location Decision. Available at: https://www.ar-eadevelopment.com/siteSelection/April2012/data-center-location-decision-criteria-26255554.shtml . Accessed 17 October 2022

Gillis A.S. 2023. Edge data center. Available at: https://www.techtarget.com/searchdatacenter/definition/edge-data-center. Accessed 12 January 2024

Global Economy. 2019. Air infrastructure quality - Country rankings. Available at: https://www.theg-lobaleconomy.com/rankings/air_transport_infrastructure/ . Accessed 26 January 2023

Global Economy. 2022. Business freedom - Country rankings. Available at: https://www.theglobale-conomy.com/rankings/herit_business_freedom/ . Accessed 19 January 2023

Global Economy. 2017. Communication prices- Country rankings. Available at: https://www.theglobaleconomy.com/rankings/communications_prices_wb/ . Accessed 18 January 2023

Global Economy. 2020. Electricity production capacity - Country rankings. Available at: https://www.theglobaleconomy.com/rankings/electricity_production_capacity/Europe/ . Accessed 16 January 2023

Global Economy. 2022. Fiscal freedom- Country rankings. Available at: https://www.theglobaleconomy.com/rankings/herit_fiscal_freedom/Europe/ . Accessed 19 January 2023

Global Economy. 2016. Internet bandwidth - Country rankings. Available at: https://www.theglobaleconomy.com/rankings/Internet_bandwidth/Europe/ . Accessed 19 January 2023

Global Economy. 2022. Investment freedom- Country rankings. Available at: https://www.theglobale-conomy.com/rankings/herit_investment_freedom/ . Accessed 19 January 2023

Global Economy. 2022. Labor cost by country: the latest data. Available at: https://www.theglobaleconomy.com/rankings/labor_cost . Accessed 16 January 2023

Global Economy. 2017. Machinery and equipment prices- Country rankings. Available at: https://www.theglobaleconomy.com/rankings/machinery_and_equipment_prices_wb/ . Accessed 18 January 2023

Global Economy. 2021. Political stability - Country rankings. Available at: https://www.theglobaleconomy.com/rankings/wb_political_stability/Europe/ . Accessed 05 January 2023

Global Economy. 2019. Port infrastructure quality - Country rankings. Available at: https://www.theg-lobaleconomy.com/rankings/seaports_quality/ . Accessed 26 January 2023

Global Economy. 2020. Renewable power capacity - Country rankings. Available at: https://www.theglobaleconomy.com/rankings/renewable_power_capacity/Europe/ . Accessed 19 January 2023

Global Economy. 2019. Railroads quality - Country rankings. Available at: https://www.theglobale-conomy.com/rankings/railroad_quality/ . Accessed 26 January 2023

Global Economy. 2019. Roads quality - Country rankings. Available at: https://www.theglobaleconomy.com/rankings/roads_quality/Europe/ . Accessed 26 January 2023

Global Economy. 2021. Rule of law - Country rankings. Available at: https://www.theglobaleconomy.com/rankings/wb_ruleoflaw/ . Accessed 16 January 2023 Global Economy. 2022. Security threats index- Country rankings. Available at: https://www.theglobaleconomy.com/rankings/security_threats_index/ . Accessed 18 January 2023

Global Economy. 2020. Social globalization - Country rankings. Available at: https://www.theglobale-conomy.com/rankings/kof_soc_glob/Europe/ . Accessed 16 January 2023

Global Economy. 2022. Trade freedom - Country rankings. Available at: https://www.theglobaleconomy.com/rankings/herit_trade_freedom/Europe/ . Accessed 14 January 2023.

Global Economy. 2017. Transport prices- Country rankings. Available at: https://www.theglobaleconomy.com/rankings/transport_prices_wb/ . Accessed 18 January 2023

Global Risk Profile. 2022. Global Corruption Index 2022. Available at: https://risk-indexes.com/global-corruption-index / . Accessed 15 January 2023

Grönroos, C. 1994. From Marketing Mix to Relationship Marketing: Towards A Paradigm Shift in Marketing. West Yorkshire: Emerald Group Publishing.

Goncalves S. 2021. Davidson Kempner, Pioneer Point to invest \$4.2 billion in Portugal data centre. Available at: https://www.streetinsider.com/Reuters/Davidson+Kempner%2C+Pioneer+Point+to+in-vest+%244.2+billion+in+data+centre+in+Portugal/18299572.html . Accessed 10 March 2024

Gooding M. 2024. CBRE: Data center demand in Europe outstrips supply. Available at: https://www.datacenterdynamics.com/en/news/cbre-data-center-demand-in-europe-outstrips-supply/. Accessed 30 January 2024

Goubin S., Ruelens A. and Nicaise I.. 2022. Trends in attitudes towards migration in Europe. A comparative analysis. Available at: https://www.opportunitiesproject.eu/media/attachments/2022/04/20/changing-attitudes-towards-migration.pdf . Accessed 16 January 2023

Haaramo E. 2015. Finland: The world's next datacentre powerhouse?. Available at: https://www.zdnet.com/article/finland-the-worlds-next-datacentre-powerhouse/ . Accessed 9 March 2024

Howard M. 2024. Cloud Computing Market: Revolutionizing the Digital Landscape with Growth, Size, Share, Trends, and Future Insights. Available at: https://www.linkedin.com/pulse/cloud-computing-market-revolutionizing-digital-landscape-maria-howard-9kbzf . Accessed 12 January 2024

Hollensen, S. 2014. Global marketing. Sixth edition. Harlow: Pearson.

IBM. 2024. What is a data center? Available at: https://www.ibm.com/topics/data-centers . Accessed 15 February 2024

IEA. 2021. Armenia energy profile. Available at: https://www.iea.org/reports/armenia-energy-profile . Accessed 15 January 2023

IEA. 2020. Belarus energy profile. Available at: https://www.iea.org/reports/belarus-energy-profile . Accessed 15 January 2023

IEA. 2021. Republic of Moldova Energy Profile. Available at: https://iea.blob.core.windows.net/as-sets/a6dd7ac3-8955-41f9-8971-d7c09ccff6d9/MoldovaEnergyProfile.pdf . Accessed 15 January 2023

Institute of Economics of the Latvian Academy of Sciences. 2023. Latvia external relations briefing: Latvia: A Growing Geopolitical Power in the European Union. Available at: https://china-cee.eu/wp-content/uploads/2023/11/2023er10_Latvia.pdf . Accessed 09 March 2024

International Renewable Energy Agency. 2022. Montenegro Energy Profile. Available at: https://www.irena.org/-/media/Files/IRENA/Agency/Statistics/Statistical_Profiles/Europe/Montenegro_Europe_RE_SP.pdf . Accessed 15 January 2023

International Trade Administration. 2022. Bosnia and Herzegovina - Country Commercial Guide. Available at: https://www.trade.gov/country-commercial-guides/bosnia-and-herzegovina-energy . Accessed 15 January 2023

International Trade Administration. 2024. Denmark - Renewable Energy Products. Available at: https://www.trade.gov/country-commercial-guides/denmark-renewable-energy-products . Accessed 09 March 2024

International Trade Administration. 2024. Portugal - Country Commercial Guide. Available at: https://www.trade.gov/country-commercial-guides/portugal-information-and-communications-technology . Accessed 09 March 2024

International Trade Administration. 2024. Slovenia - Country Commercial Guide. Available at: https://www.trade.gov/country-commercial-guides/slovenia-information-and-communications-technology . Accessed 09 March 2024

Interreg Europe. 2018. Open R&D centres in Lithuania. Available at: https://www.interregeurope.eu/good-practices/open-rd-centres-in-lithuania . Accessed 07 March 2024

Invest Lithuania. 2022. Lithuania's tech sector at a glance. Available at: https://investlithuania.com/tech/ . Accessed 07 March 2024

Iron Mountain. 2023. Data center power consumption 101: What to know about powering data centers. Available at: https://www.ironmountain.com/resources/blogs-and-articles/d/data-centers-power-con-sumption-101-what-to-know-about-powering-data-centers . Accessed 14 March 2023

Kent, R. A. 1986. Faith in the four Ps: An alternative. *Management Decision*. Abingdon: Journal of Marketing Management, 145-154.

Lettire, C. 2021. Hot or Not? How Data Center Thermal Standards Impact Energy Use and Costs. Available at: https://www.coresite.com/blog/hot-or-not-how-data-center-thermal-standards-impact-energy-use-and-costs . Accessed 17 October 2022

López-Dóriga E. 2022. Portugal, a strong bet on renewables to reduce its energy dependency. Available at: https://www.unav.edu/en/web/global-affairs/portugal-a-strong-bet-on-renewables-to-reduce-its-energy-dependency . Accessed 10 March 2024

Macrotrends. 2023. San Marino Renewable Energy 1990-2023. Available at: https://www.macro-trends.net/countries/SMR/san-marino/renewable-energy-statistics . Accessed 15 January 2023

ManageEngine. 2023. Data center security and privacy guidelines 101. Available at: https://www.site24x7.com/learn/datacenter/data-center-security-and-privacy-for-europe.html . Accessed 20 March 2024

Miller R. 2013. Microsoft Invests \$230 Million to Expand Dublin Data Center. Available at: https://www.datacenterknowledge.com/archives/2013/12/03/microsoft-expands-dublin-data-center#close-modal . Accessed 09 March 2024

Ministry of Foreign Affairs of Denmark. 2022. What makes Denmark named World's most Sustainable Country. Available at: https://indien.um.dk/en/news/what-makes-denmark-named-worlds-most-sustainable-country . Accessed 09 March 2024

Ministry of Foreign Affairs of Denmark. 2023. Denmark #1 in European innovation. Available at: https://investindk.com/insights/denmark-number-1-in-european-innovation . Accessed 09 March 2024

Mitchell, T.D., Carter, T.R., Jones, P.D., Hulme, M., New, M., 2003: A Comprehensive Set of High-Resolution Grids of Monthly Climate for Europe and the Globe: the Observed Record (1901-2000) and 16 Scenarios (2001-2100). Available at: https://web.archive.org/web/20150905135247/http://lebanese-economy-forum.com/wdi-gdf-advanced-data-display/show/EN-CLC-AVRT-C/ . Accessed 03 January 2023

Mohamed, D. F. W. 2014. Using Cloud Computing in Supporting the Management of Travel Agencies. *Computer Science and Information Technology*. California: Horizon Research Publishing, 323-330.

Monserrate, S. G. 2022. The Staggering Ecological Impacts of Computation and the Cloud. Available at: https://thereader.mitpress.mit.edu/the-staggering-ecological-impacts-of-computation-and-the-cloud/ . Accessed 15 December 2022

Mordor Intelligence. 2023. Europe Data Center Market Size. Available at: https://www.mordorintelligence.com/industry-reports/europe-colocation-market-industry. Accessed 20 March 2024

Multiplier. 2024. Subsidiaries in Latvia. Available at: https://www.usemultiplier.com/latvia/setting-up-a-subsidiary-company . Accessed 10 March 2024

Munde, S. 2024. Green data center market. Available at: https://www.marketresearchfuture.com/reports/green-data-center-market-1534?utm_term=&utm_campaign=&utm_source=adwords&utm_medium=ppc&hsa_acc=2893753364&hsa_cam=20817028056&hsa_grp=160107916030&hsa_ad=68312 7039302&hsa_src=g&hsa_tgt=dsa-2160750282649&hsa_kw=&hsa_mt=&hsa_net=adwords&hsa_ver=3&gad_source=1. Accessed 18 May 2024

Nareaho S., Kettunen J., Karki A. & Paallysaho S. 2020. Responsible thesis: Ethical recommendations for thesis writing at universities of applied sciences. Available at: https://www.arene.fi/wp-content/up-loads/Raportit/2018/ETHICAL%20RECOMMENDATIONS%20FOR%20THESIS%20WRIT-ING%20AT%20UNIVERSITIES%20OF%20APPLIED%20SCIENCES.pdf . Accessed 29 April 2024

Neogames. 2018. Finnish game industry: Policy recommendations for 2019-2024. Available at: https://www.neogames.fi/wp-content/uploads/2019/01/Finnish-Game-industry-policy-recommendations-for-2019%E2%80%932024.pdf . Accessed 07 March 2024

OECD. 2002. Foreign Direct Investment Restrictions in OECD Countries. Available at: https://www.oecd.org/economy/reform/2956455.pdf . Accessed 14 January 2023

OECD. 2017. National Treatment for Foreign-Controlled Enterprises. Available at: https://www.oecd.org/investment/investment-policy/national-treatment-instrument-english.pdf . Accessed 14 January 2023

OECD Library. 2021. Going digital in Latvia. Available at: https://www.oecd-ilibrary.org/sites/8eec1828-en/index.html?itemId=/content/publication/8eec1828-en . Accessed 09 March 2024

Palmer, A. 2004. Introduction to Marketing - Theory and Practice. UK: Oxford University Press. Parrish, K., Holslin, P. & Armstrong, R.L. 2024. Internet Speed vs. Bandwidth: What's the Difference?. Available at: https://www.highspeedinternet.com/resources/internet-speed-vs-bandwidth . Accessed 05 May 2024

Portugal Digital. 2020. Portugal's Action Plan for Digital Transition. Available at: https://portugaldigital.gov.pt/wp-content/uploads/2022/01/Portugal_Action_Plan_for_Digital_Transition.pdf . Accessed 09 March 2024

Radiant. 2022. IoT Data Center Monitoring: How IoT Is Changing The Way Data Centers Operate. Available at: https://radiantrfid.com/blog/iot-data-center-monitoring-how-iot-is-changing-the-way-data-centers-operate/#:~:text=Data%20centers%20are%20increasingly%20adopting,energy%20us-age%2C%20and%20asset%20utilization. Accessed at 10 December 2023

Rana, S & Shrivastava, A. K. 2007. Doing business in emerging markets. London: Routledge.

ReportLinker. 2022. The Czech Republic ICT Industry Outlook 2022 - 2026. Available at: https://www.reportlinker.com/clp/country/597055/726391 . Accessed 07 March 2024

Ribokas M. 2021. TECH sector in Lithuania shows resilience and strong growth. Available at: https://www.linkedin.com/pulse/tech-sector-lithuania-shows-resilience-strong-growth-marius-ribokas/. Accessed 10 March 2024

Rihti M. 2019. A datacenter in Finland is all you need – safe, stable and energy-efficient. Available at: https://www.telia.fi/business/article/datacenter-in-finland-is-safe-stable-and-energy-efficient . Accessed 09 March 2024

Ritchie H. and Roser M. 2021. Russia: Energy Country Profile. Available at: https://our-worldindata.org/energy/country/russia . Accessed 15 January 2023

Ruzgys, T. 2022. Lithuania - a country that has already earned its name in the IT sector. Available at: https://www.crayon.com/lt/resources/news2/Lithuania-and-IT/ . Accessed 09 March 2024

Saunders, M., Lewis, P & Thornhill, A. 2006. *Research methods for business students*. Fourth edition. Harlow: Pearson Education.

Saunders, M., Lewis, P & Thornhill, A. 2016. *Research methods for business students*. Seventh edition. Harlow: Pearson Education.

Savvas A. 2021. The data centre Ireland paradise continues to grow. Available at: https://www.capaci-tymedia.com/article/29otd6mddjpstgh0mo0sg/feature/the-data-centre-ireland-paradise-continues-to-grow . Accessed 09 March 2024

Sears, M.J. 2023. Digital Finland is one of the leaders in the EU regarding human capital. Available at: https://www.linkedin.com/pulse/digital-finland-one-leaders-eu-regarding-human-capital-sear-idklf/ . Accessed 07 January 2024

Slovenia Business. 2023. Research, development and innovation: Top tallent, excellent support. Available at: https://www.sloveniabusiness.eu/business-environment/research-development-and-innovation . Accessed 07 March 2024

Slovenia Business Development Agency. 2022. Smart Slovenia: How Research And Development Is Positioning The Country As A Global Innovation Hub. Available at: https://sponsorcontent.cnn.com/int/spirit-slovenia/how-research-and-development-is-positioning-the-country-as-a-globalinnovation-hub/ . Accessed 09 March 2024

Stackscale. 2024. Cloud adoption among EU enterprises in 2023. Available at: https://www.stackscale.com/blog/cloud-adoption-eu-enterprises/. Accessed 18 February 2024

Statista. 2022. Number of data centers in Europe by country 2021. Available at: https://www.sta-tista.com/statistics/878621/european-data-centers-by-country/ . Accessed 17 October 2022

Statista. 2023. Travel and tourism in Croatia - statistics & factsTravel and tourism in Croatia - statistics & facts. Available at: https://www.statista.com/topics/7340/travel-and-tourism-in-croatia/#topicOverview . Accessed 09 March 2024

Statista. 2023. Data Center - Croatia. Available at: https://www.statista.com/outlook/tmo/data-center/croatia . Accessed 07 March 2024

Statistics Finland. 2021. Finland exceeded its target for 2020 – share of renewable energy in final consumption rose to 44.6 per cent. Available at: https://www.stat.fi/til/ehk/2020/ehk_2020_2021-12-16_tie_001_en.html . Accessed 09 March 2024

Stenson J. 2019. 'The new oil': Dublin strikes it rich as Europe's data hub. Available at: https://phys.org/news/2019-01-oil-dublin-rich-europe-hub.html . Accessed 09 March 2024

Stevens, R. E., Sherwood, P. K., Dunn, J. P & Loudon, D. L. 2002. *Market opportunity analysis*. First edition. New York: Routledge.

Team Evoque. 2021. A brief history of data centers and modern colocation. Available at: https://www.evoquedcs.com/blog/a-brief-history-of-data-centers-and-modern-colocation . Accessed 17 October 2022

Tech Central. 2020. Dublin to enter top five of data centre cities in Europe. Available at: https://www.techcentral.ie/dublin-to-enter-top-five-of-data-centre-cities-in-europe/. Accessed 09 March 2024

Tech Explore.2021. US-British firm to build 3.5 bn euro data centre in Portugal. Available at: https://techxplore.com/news/2021-04-us-british-firm-bn-euro-centre.html . Accessed 10 March 2024

Tech News And Updates. 2023. Data Centre (Data Centers) Market | Expected To Be the Fastest Growing Industry 2031. Available at: https://www.linkedin.com/pulse/data-centre-centers-market-expected-fastest-growing-r4ohf#:~:text=The%20global%20Data%20Centre%20(Data,USD%20200.34%20million%20by%202028. Accessed 12 January 2024

Technology Industries of Finland. 2023. Strategy. Available at: https://teknologiateollisuus.fi/en/about/strategy . Accessed 09 March 2024

Technology Industries of Finland. 2020. Technology Industries of Finland's low-carbon roadmap: solutions to the climate challenge. Available at: https://teknologiateollisuus.fi/en/focus/environment-andsustainability/technology-industries-finlands-low-carbon-roadmap-solutions . Accessed 09 March 2024

TechTarget. 2022. Data Center. Available at: https://www.techtarget.com/searchdatacenter/definition/data-center#:~:text=The%20origins%20of%20the%20first,complex%20to%20maintain%20and%20operate . Accessed 17 October 2022.

The Ministry of Energy of The Republic of Azerbaijan. 2022. The Use of Renewable Energy Resources in Azerbaijan. Available at: https://minenergy.gov.az/en/alternativ-ve-berpa-olunan-enerji/azerbaycanda-berpa-olunan-enerji-menbelerinden-istifade . Accessed 15 January 2023

The World Bank. 2020. Access to electricity (% of population). Available at: https://data.worldbank.org/indicator/EG.ELC.ACCS.ZS?name_desc=false . Accessed 02 January 2023

The World Bank. 2009. Droughts, floods, extreme temperatures (% of population, average 1990-2009). Available at: https://data.worldbank.org/indicator/EN.CLC.MDAT.ZS?name_desc=false . Accessed 02 January 2023

The World Bank. 2018. Electric power transmission and distribution losses (% of output). Available at: https://data.worldbank.org/indicator/EG.ELC.LOSS.ZS?name_desc=false . Accessed 02 January 2023

The World Bank. 2021. GDP per capita (current US\$). Available at: https://data.worldbank.org/indica-tor/NY.GDP.PCAP.CD?name_desc=false . Accessed 02 January 2023

The World Bank. 2020. Individuals using the Internet (% of population). Available at: https://data.worldbank.org/indicator/IT.NET.USER.ZS?name_desc=false . Accessed 02 January 2023

The World Bank. 2020. Population density (people per sq. km of land area). Available at: https://data.worldbank.org/indicator/EN.POP.DNST?name_desc=false . Accessed 02 January 2023

The World Bank. 2021. Population, total. Available at: https://data.worldbank.org/indica-tor/SP.POP.TOTL?name_desc=false . Accessed 02 January 2023

The World Bank. 2021. Surface area (sq. km). Available at: https://data.worldbank.org/indica-tor/AG.SRF.TOTL.K2?name_desc=false . Accessed 19 December 2022

The World Bank. 2020. Profit tax (% of commercial profits). Available at: https://data.worldbank.org/indicator/IC.TAX.PRFT.CP.ZS?name_desc=false . Accessed 03 January 2023 UNECE RE-Uptake. 2021. Factsheet: Renewable Energy in Georgia. Available at: https://www.ren21.net/wp-content/uploads/2019/05/Factsheet_Georgia-HardTalk-2021.pdf . Accessed 15 January 2023

Vekic, A., Djakovic, V., Borocki, J., Sroka, W., Popp, J & Olàh, J. 2020. The Importance of Academic New Ventures for Sustainable Regional Development. *Amfiteatru Economic*, 22(54), 533-550.

Verizon. 2022. Bandwidth. Available at: https://www.verizon.com/articles/internet-essentials/band-width-definition/ . Accessed 03 February 2023

Vilnius City Innovation Industrial Park. 2022. Innovative business development in the Lithuanian capital. Available at: https://vciip.com/ . Accessed 10 March 2024

Wheelen, T.L., Hunger, J.D., Hoffman, A.N & Bamford, C.E. 2018. *Strategic Management and Business Policy: Globalization, Innovation and Sustainability.* Fifteenth edition. Boston: Pearson.

White Papers. 2022. Portugal: A Hidden Gem of Connectivity. Available at: https://www.startcampus.pt/en/media/white-papers/portugal-a-hidden-gem-of-connectivity/ . Accessed 09 March 2024

Wiboonrat M. 2014. Cloud Computing in Budget Hotel Management. *IEEE International Conference on Management of Innovation and Technology*, 2014, 327-332, doi: 10.1109/ICMIT.2014.6942447.

World Data. 2020. Production capacities per energy source. Available at: https://www.worlddata.info/europe/andorra/energy-consumption.php . Accessed 15 January 2023

World Economics. 2020. Environmental Factors. Available at: https://www.worldeconomics.com/Indicator-Data/ESG/Environment.aspx . Accessed 15 January 2023

World Intellectual Property Organization. 2022. Global Innovation Index 2022. Available at: https://www.wipo.int/edocs/pubdocs/en/wipo_pub_2000_2022/fi.pdf . Accessed 09 March 2024

World Population Review. 2023. Internet Speeds by Country 2023. Available at: https://worldpopula-tionreview.com/country-rankings/internet-speeds-by-country . Accessed 15 January 2023

Yin, R. K. 2009. *Case study research: Design and methods*. Fourth edition. California: Sage Publications.

Young A. 2018. Equinix investing in new green Helsinki data centre. Available at: https://datacenternews.asia/story/equinix-investing-in-new-green-helsinki-data-centre Accessed 09 March 2024

Zhang, M. 2022. How Much Does it Cost to Build a Data Center?. Available at: https://dgtlin-fra.com/how-much-does-it-cost-to-build-a-data-center/ . Accessed 19 December 2022
(487 DC) Germany	(83,196 ppl)			
(456 DC) UK	(67,327 ppl)			
(281 DC) Netherlands	(17,533 ppl)			
(264 DC) France	(67,750 ppl)			
(172 DC) Russia	(143,449 ppl)			
(136 DC) Poland	(37,747 ppl)			
(131 DC) Italy	(59,110 ppl)			
(125 DC) Spain	(47,416 ppl)			
(104 DC) Switzerland	(8,703 ppl)			
(81 DC)	Śweden	(10,416 ppl)			
(74 DC)	Belgium	(11,593 ppl)			
(66 DC)	Austria	(8,956 ppl)			
(57 DC)	Ukraine	(43,793 ppl)			
(49 DC)	Ireland	(5,033 ppl)			
(38 DC)	Denmark	(5,857 ppl)			
(37 DC)	Norway	(5,408 ppl)			
(34 DC)	Finland	(5,541 ppl)			
(33 DC)	Czech Republic	(10,506 ppl)			
(33 DC)	Turkey	(84,775 ppl)			
(24 DC)	Bulgaria	(6,878 ppl)			
(24 DC)	Latvia	(1,884 ppl)			
(24 DC)	Slovenia	(2,108 ppl)			
(26 DC)	Romania	(19,120 ppl)			
(17 DC)	Lithuania	(2,801 ppl)			
(17 DC)	Portugal	(10,325 ppl)			
(13 DC)	Hungary	(9,710 ppl)			
(13 DC)	Slovakia	(5,447 ppl)			
(13 DC)	Greece	(10,641 ppl)			
(9 DC)	Belarus	(9,340 ppl)			
(8 DC)	Croatia	(3,899 ppl)			
(8 DC)	Serbia	(6,834 ppl)			
(6 DC)	Iceland	(373 ppl)			
(6 DC)	Macedonia	(2,065 ppl)			
(6 DC)	Moldova	(2,615 ppl)			
(5 DC)	Cyprus	(1,244 ppl)			
(5 DC)	Georgia	(3,709 ppl)			
(4 DC)	Malta	(519 ppl)			
(2 DC)	Bosnia and Herzegovina	(3,271 ppl)			
(1 DC)	Albania	(2,812 ppl)			
(1 DC)	Azerbaijan	(10,138 ppl)			

Number of data centers in Europe by country 2021 compared to total population (2021 value in thousands) (adapted from Statista 2021 & The World Bank 2021)

(487 DC)	Germany	(238 ppl/km2)			
(456 DC)	UK	(277 ppl/km2)			
(281 DC)	Netherlands	(518 ppl/km2)			
(264 DC)	France	(123 ppl/km2)			
(172 DC)	Russia	(9 ppl/km2)			
(136 DC)	Poland	(124 ppl/km2)			
(131 DC)	Italv	(201 ppl/km2)			-
(125 DC)	Spain	(95 ppl/km2)			
(104 DC)	Switzerland	(219 ppl/km2)			
(81 DC)	Sweden	(25 ppl/km2)			-
(74 DC)	Belaium	(381 ppl/km2)			
(66 DC)	Austria	(108 ppl/km2)			
(57 DC)	Ukraine	(76 ppl/km2)	1	i i	
(49 DC)	Ireland	(72 pp/km2)			=
(38 DC)	Denmark	(146 ppl/km2)			=
(37 DC)	Norway	(15 ppl/km2)			
(34 DC)	Finland	(18 ppl/km2)			_
(33 DC)	Czech Republic	(139 ppl/km2)			_
(33 DC)	Turkev	(109 ppl/km2)			
(24 DC)	Bulgaria	(64 ppl/km2)			
(24 DC)	Latvia	(31 ppl/km2)			
(24 DC)	Slovenia	(104 ppl/km2)			
(26 DC)	Romania	(84 ppl/km2)			
(17 DC)	Lithuania	(45 ppl/km2)			
(17 DC)	Portugal	(112 ppl/km2)			
(13 DC)	Hungary	(107 ppl/km2)			
(13 DC)	Slovakia	(114 ppl/km2)			
(13 DC)	Greece	(83 ppl/km2)			
(9 DC)	Belarus	(46 ppl/km2)			
(8 DC)	Croatia	(72 ppl/km2)			
(8 DC)	Serbia	(79 ppl/km2)			
(6 DC)	Iceland	(4 ppl/km2)			
(6 DC)	Macedonia	(82 ppl/km2)			
(6 DC)	Moldova	(92 ppl/km2)			
(5 DC)	Cyprus	(134 ppl/km2)			
(5 DC)	Georgia	(65 ppl/km2)			
(4 DC)	Malta	(1610 ppl/km2)			
(2 DC) Bo	snia and Herzegovina	(65 ppl/km2)			
(1 DC)	Albania	(104 ppl/km2)			
(1 DC)	Azerbaijan	(122 ppl/km2)			

Number of data centers in Europe by country 2021 compared to population density (2022 number of people/km2) (adapted from Statista 2021 & The World Bank 2020)

(487 DC)	Germany	(100%)	
(456 DC)	UK	(100%)	
(281 DC)	Netherlands	(100%)	
(264 DC)	France	(100%)	
(172 DC)	Russia	(100%)	
(136 DC)	Poland	(100%)	
(131 DC)	Italy	(100%)	
(125 DC)	Spain	(100%)	
(104 DC)	Switzerland	(100%)	
(81 DC)	Sweden	(100%)	
(74 DC)	Belgium	(100%)	
(66 DC)	Austria	(100%)	
(57 DC)	Ukraine	(100%)	
(49 DC)	Ireland	(100%)	
(38 DC)	Denmark	(100%)	
(37 DC)	Norway	(100%)	
(34 DC)	Finland	(100%)	
(33 DC)	Czech Republic	(100%)	
(33 DC)	Turkey	(100%)	
24 DC)	Bulgaria	(100%)	
(24 DC)	Latvia	(100%)	
(24 DC)	Slovenia	(100%)	
(26 DC)	Romania	(100%)	
(17 DC)	Lithuania	(100%)	
(17 DC)	Portugal	(100%)	
(13 DC)	Hungary	(100%)	
(13 DC)	Slovakia	(100%)	
(13 DC)	Greece	(100%)	
(9 DC)	Belarus	(100%)	
(8 DC)	Croatia	(100%)	
(8 DC)	Serbia	(100%)	
(6 DC)	Iceland	(100%)	
(6 DC)	Macedonia	(100%)	
(6 DC)	Moldova	(100%)	
(5 DC)	Cyprus	(100%)	
(5 DC)	Georgia	(100%)	
(4 DC)	Malta	(100%)	
(2 DC)	Bosnia and Herzegovina	(100%)	
(1 DC)	Albania	(100%)	
(1 DC)	Azerbaijan	(100%)	

Number of data centers in Europe by country 2021 compared to percentage of population with access to electricity (2020) (adapted from Statista 2021 & The World Bank 2020)

(487 DC)	Germany	(91%)	
(456 DC)	UK	(95%)	
(281 DC)	Netherlands	(92%)	
(264 DC)	France	(86%)	
(172 DC)	Russia	(88%)	
(136 DC)	Poland	(85%)	
(131 DC)	Italy	(75%)	
(125 DC)	Spain	(94%)	
(104 DC)	Switzerland	(96%)	
(81 DC)	Sweden	(88%)	
(74 DC)	Belgium	(93%)	
(66 DC)	Austria	(93%)	
(57 DC)	Ukraine	(75%)	
(49 DC)	Ireland	(92%)	
(38 DC)	Denmark	(99%)	
(37 DC)	Norway	(99%)	
(34 DC)	Finland	(93%)	
(33 DC)	Czech Republic	(83%)	
(33 DC)	Turkev	(81%)	
(24 DC)	Bulgaria	(75%)	
(24 DC)	Latvia	(91%)	
(24 DC)	Slovenia	(89%)	
(26 DC)	Romania	(84%)	
(17 DC)	Lithuania	(87%)	
(17 DC)	Portugal	(82%)	
(13 DC)	Hungary	(89%)	
(13 DC)	Slovakia	(89%)	
(13 DC)	Greece	(78%)	
(9 DC)	Belarus	(87%)	
(8 DC)	Croatia	(81%)	
(8 DC)	Serbia	(81%)	
(6 DC)	Iceland	(100%)	
(6 DC)	Macedonia	(81%)	
(6 DC)	Moldova	(76%)	
(5 DC)	Cyprus	(91%)	
(5 DC)	Georgia	(76%)	
(4 DC)	Malta	(87%)	
(2 DC)	Bosnia and Herzegovina	(76%)	
(1 DC)	Albania	(79%)	
(1 DC)	Azerbaijan	(85%)	
(1DC)	Azerbaljan	(05%)	

Number of data centers in Europe by country 2021 compared to percentage of population using internet (2020) (adapted from Statista 2021 & The World Bank 2020)

	Restrictions on Foreign Investments (from least to most)								
1	United Kingdom	0,06	18	Poland	0,21	35	Latvia	no data	
2	Ireland	0,07	19	Austria	0,27	36	Liechtenstein	no data	
3	Netherlands	0,08	20	Turkiye	0,34	37	Lithuania	no data	
4	Germany	0,08	21	Iceland	0,39	38	Luxembourg	no data	
5	Denmark	0,085	22	Albania	no data	39	Malta	no data	
6	Belgium	0,09	23	Andorra	no data	40	Moldova	no data	
7	Italy	0,10	24	Armenia	no data	41	Monaco	no data	
8	France	0,11	25	Azerbaijan	no data	42	Montenegro	no data	
9	Greece	0,13	26	Belarus	no data	43	North Macedonia	no data	
10	Sweden	0,14	27	Bosnia and Herzegovina	no data	44	Romania	no data	
11	Portugal	0,15	28	Bulgaria	no data	45	Russian Federation	no data	
12	Hungary	0,16	29	Croatia	no data	46	San Marino	no data	
13	Spain	0,165	30	Cyprus	no data	47	Serbia	no data	
14	Czechia	0,17	31	Estonia	no data	48	Slovak Republic	no data	
15	Switzerland	0,17	32	Georgia	no data	49	Slovenia	no data	
16	Finland	0,18	33	Kazakhstan	no data	50	Ukraine	no data	
17	Norway	0,185	34	Kosovo	no data				

Ranking of European countries by restrictions on foreign investments in 2002 (OECD 2002, 5)

APPENDIX 6/1

No #	Country (ranked by ascending difficulty level for foreign controlled eneterprises to invest locally)	Member of National Treatment for Foreign-Controlled Enterprises	Exceptions (at national level or by territorial subdivisions)	Conclusions	
1	Czechia	Yes	None/ Not relevant to DC Industry	Lack of restrictions makes it easy for foreign controlled companies to invest in this location	
2	Portugal	Yes	None/ Not relevant to DC Industry	Lack of restrictions makes it easy for foreign controlled companies to invest in this location	
3	Romania	Yes	None/ Not relevant to DC Industry	Lack of restrictions makes it easy for foreign controlled companies to invest in this location	
4	Slovak Republic	Yes	None/ Not relevant to DC Industry	Lack of restrictions makes it easy for foreign controlled companies to invest in this location	
5	Poland	Yes	-Foreign-controlled enterprises require authorisation for the acquisition of real estate (35)	Possible impact on: -schedule -operation	
6	Latvia	Yes	-Purchase of land in some locations is restricted (25)	Impact on: -limits site location selection	
7	Ireland	Yes	-Air transport limited to EU carriers (22)	Possible impact on: -schedule -cost	
8	Italy	Yes	-Air transport limited to EU carriers (23)	Possible impact on: -schedule -cost	
9	Belgium	Yes	-Transport of goods on inland waterway is limited to EU carriers (3)	Possible impact on: -schedule -cost	
10	Slovenia	Yes	-Maritime transport limited to EU carriers (36)	Possible impact on: -schedule -cost	
11	Hungary	Yes	-Air transport limited to EU carriers (17) -Inland waterways transport limited to EU carriers (18)	Possible impact on: -schedule -cost	
12	Netherlands	Yes	-Air transport limited to EU carriers (30) -Maritime transport limited to EU carriers (31) -Inland waterways transport limited to EU carriers (32)	Possible impact on: -schedule -cost	
13	Luxembourg	Yes	-Restrictions on air and land transport schedule (29)	Possible impact on: -schedule -cost -operation	
14	Norway	Yes	-Maritime transport limited to EEA carriers (33) -Road transport limited to Norwegian carriers (34)	Possible impact on: -schedule -cost -operation	
15	United Kingdom	Yes	-Maritime transport limited to British carriers (49) -Inland waterways transport limited to EU carriers (50)	Possible impact on: -schedule -cost	
16	Denmark	Yes	-Air transport limited to Danish carriers (4)	Possible impact on: -schedule -cost	
17	Finland	Yes	-Air transport limited to Finnish carriers (7)	Possible impact on: -schedule -cost	
18	Spain	Yes	-Air transport limited to Spanish carriers (37)	Possible impact on: -schedule -cost	
19	Turkiye	Yes	-Maritime transport is limited to Turkish carriers (45)	Possible impact on: -schedule -cost	
20	Greece	Yes	-Purchase of land is limited to EU members (15) -Maritime transport is limited to Greek carriers (16)	Possible impact on: -schedule -cost -operation	
21	Austria	Yes	-Air transport limited to Austrian carriers (1) -Real estate acquisition requires authorization (2)	Possible impact on: -schedule -cost -operation	
22	Sweden	Yes	-Air transport limited to Swedish carriers (38) -Air routes restrictions for foreign enterprises (39) -Maritime transport limited to Swedish carriers (40)	Possible impact on: -schedule -cost	
23	France	Yes	-Transport of goods on inland waterway limited to EU carriers (8) -Foreign participation in newly privatised companies is limited (9)	Impact on: -access to the local market opportunitites Possible impact on: -schedule -cost	
24	Lithuania	Yes	-Restrictions on land acquisition (26) -Maritime and inland waterway transport is limited to Lithuanian carriers (27) -Restrictions for foreign legal persons receiving state subsidies (28)	Impact on: -limits site location selection Possible impact on: -schedule -cost -operation	
25	Estonia	Yes	-Purchase of real estate in some locations is restricted to EEA nationals with minimum 3 years residence (5) -Maritime transport limited to EU carriers (6)	Possible impact on: -schedule -cost -operation	

APPENDIX 6/2

No #	Country (ranked by ascending difficulty level for foreign controlled eneterprises to invest locally)	Member of National Treatment for Foreign-Controlled Enterprises	Exceptions (at national level or by territorial subdivisions)	Conclusions
26	Switzerland	Yes	-Air transport is limited to Swiss carriers (41) -Inland waterways transport is limited to Swiss carriers (42) -Maritime transport is limited to Swiss carriers (43) -Real estate acquisition requires authorization (44)	Possible impact on: -schedule -cost -operation
27	Ukraine	Yes	-Restrictions on purchases of state-owned land by foreign legal entities (46) -Maritime transport limited to Ukrainian carriers (47) -Foreign firms participation to procurment biddings is forbidden (48)	Possible impact on: -schedule -cost -operation -future expansion and development
28	Germany	Yes	-Air transport limited to EEA carriers (10) -Maritime transport limited to EU carriers (11) -Inland waterways transport limited to EU carriers (12) -Rail transport limited to German carriers (13) -Disadvantages for enterprises not established as legally independent (14)	Possible impact on: -schedule -cost -operation
29	Iceland	Yes	-Restrictions on foreign investments. (19) -Restrictions and prohibitions on investments by foreign states or state- owned enterprises without authorization (20) -Foreign investment in real estate is limited to EEA nationals (21)	Highest number of restrictions on foreign investments amongst member states. Possible impact on: -schedule -cost -operation
30	Kazakhstan	Yes	-Governmental approval needed for owning and using Telecommunications services (24)	Hiighest level of impact to project cost, schedule and operation.
31	Albania	No	N/A	Lack of membership makes it more difficult for foreign controlled companies to invest in this location
32	Andorra	No	N/A	Lack of membership makes it more difficult for foreign controlled companies to invest in this location
33	Armenia	No	N/A	Lack of membership makes it more difficult for foreign controlled companies to invest in this location
34	Azerbaijan	No	N/A	Lack of membership makes it more difficult for foreign controlled companies to invest in this location
35	Belarus	No	N/A	Lack of membership makes it more difficult for foreign controlled companies to invest in this location
36	Bosnia and Herzegovina	No	N/A	Lack of membership makes it more difficult for foreign controlled companies to invest in this location
37	Bulgaria	No	N/A	Lack of membership makes it more difficult for foreign controlled companies to invest in this location
38	Croatia	No	N/A	Lack of membership makes it more difficult for foreign controlled companies to invest in this location
39	Cyprus	No	N/A	Lack of membership makes it more difficult for foreign controlled companies to invest in this location
40	Georgia	No	N/A	Lack of membership makes it more difficult for foreign controlled companies to invest in this location
41	Kosovo	No	N/A	Lack of membership makes it more difficult for foreign controlled companies to invest in this location
42	Liechtenstein	No	N/A	Lack of membership makes it more difficult for foreign controlled companies to invest in this location
43	Malta	No	N/A	Lack of membership makes it more difficult for foreign controlled companies to invest in this location
44	Moldova	No	N/A	Lack of membership makes it more difficult for foreign controlled companies to invest in this location
45	Monaco	No	N/A	Lack of membership makes it more difficult for foreign controlled companies to invest in this location
46	Montenegro	No	N/A	Lack of membership makes it more difficult for foreign controlled companies to invest in this location
47	North Macedonia	No	N/A	Lack of membership makes it more difficult for foreign controlled companies to invest in this location
48	Russian Federation	No	N/A	Lack of membership makes it more difficult for foreign controlled companies to invest in this location
49	San Marino	No	N/A	Lack of membership makes it more difficult for foreign controlled companies to invest in this location
50	Serbia	No	N/A	Lack of membership makes it more difficult for foreign controlled companies to invest in this location

APPENDIX 6/3

Note:
1. */Lit manport. Cabdings reserved to national arises, *(CECD, 14)
1. */Lit manport. Cabdings reserved to national arises, *(CECD, 14)
1. */Lit manport. Cabdings reserved to national arises, *(CECD, 14)
1. */Lit manport. Cabdings reserved to researce on the territy of any of these states, which are owned in majority and controlled by nationals of these states, *(CECD, 16)
1. */Lit manport. Cabdings reserved to national arises, *(CECD, 14)
1. */Lit manport. Cabdings reserved to national arises, *(CECD, 15)
1. */Lit manport. Cabdings reserved to national arises, *(CECD, 16)
1. */Lit manport. Cabdings reserved to national arises, *(CECD, 15)
1. */Lit manport. Cabdings reserved to national arises, *(CECD, 16)
1. */Lit manport. Cabdings reserved to sease) on westers, being the national flag of the Results of States, *(CECD, 16)
1. */Lit manport. Cabdings reserved to sease) on westers, being the national flag of the Results of States, *(CECD, 16)
1. */Lit manport. Cabdings reserved to sease) on westers, being the national flag of the Results of States, *(CECD, 16)
1. */Lit manport. Cabdings reserved to sease) on westers, being manufacture, or a cabding and states, *(CECD, 16)
1. */Lit manport. Cabdings reserved to sease) on westers, being and states, which are owned by test reserved to sease) on westers, westers on the sease on westers, westers on westers, westers on the sease on westers, wester

http://discret/action/a